## Respecting the Autonomy of Elderly Patients in Switzerland:

# Hindering and Promoting Factors in Clinical Settings

## Inauguraldissertation

zur

Erlangung der Würde eines Dr. sc. med.

vorgelegt der

Medizinischen Fakultät

der Universität Basel

von

Milenko Rakic

aus Castaneda, Schweiz

Basel, 2020

Originaldokument gespeichert auf dem Dokumentenserver der Universität Basel
<a href="mailto:edoc.unibas.ch">edoc.unibas.ch</a>

Genehmigt von der Medizinischen Fakultät

auf Antrag von

Dissertationsleiterin: Dr. phil. Isabelle Wienand

Fakultätsverantwortliche: Prof. Dr. med. Bernice Elger

Koreferent: PD Dr. med. und Dr. med. dent. Christoph Leiggener

Externer Experte: Prof. Dr. phil. Walter Lesch

Basel, den 26.06.2018

Prof. Dr. Primo Schär

Dekan

# **Table of Contents**

Acknowledgements	1
Contributions	3
Summary	6
Zusammenfassung	9
Chapter 1: Background 1 Introduction	12
1.1 Demographic Shift and Increase in Chronic Diseases	13
1.2 Treatment Options in Light of Chronic Diseases	
1.3 Palliative Care	
1.4 Theoretical Background: Autonomy	17
1.5 Conclusion	
2 Research Objectives and Empirical Methods	21
2.1 Research Objectives	22
2.2 Empirical Methods	23
Chapter 2: Autonomy and Fear of Synthetic Biology: How Can Patients' Autonomy Be Enhanced in the Field of Synthetic Biology? A Qualitative Study with Stable Patients  Chapter 3: Enhancing Patients' Autonomy by Involving Them in Research Ethics  Committees	
<b>Chapter 4</b> : The Beneficence of Hope: Findings from a Qualitative Study with Gout and Diabetes Patients	73
Chapter 5: Feelings of Burden in Palliative Care: a Qualitative Analysis of Medical Reco	
Chapter 6: Advance Care Planning in Acute Hospitals: A Retrospective Analysis	
of Medical Records	.118
Chapter 7: General Discussion	.135
1 Hindering and Promoting Factors in a Research Context	.136
2 Hindering and Promoting Factors in a Palliative Care Context	.142
3 Overview	145
4 Research Limitations	146
5 Implications for Further Research	.147
6 Conclusion	147

Appendix	153
Questionnaire for Patients	154
Medical Records Data Extraction Sheet	161

## **Acknowledgements**

In the following, I would like to thank my colleagues, family, and friends for their supervision, assistance, and support throughout my PhD. I could not have completed this thesis without them.

First of all, I wish to thank the Swiss Academy of Medical Sciences for funding the project "Respect for patient self-determination as quality indicator in palliative care: current state, problems and solutions in acute care hospitals". I am thankful to Prof. Dr. Bernice Elger, who designed the project and gave me the opportunity to work on such important topics. I would like to acknowledge the support of all collaborators from the participating centers who facilitated data collection and provided valuable suggestions on the manuscripts. Finally, I would like to express my sincere gratitude to all the patients who participated in the studies and provided us with personal insights into their difficult situations.

Further, I would like to thank my PhD-Committee. I thank Prof. Dr. Bernice Elger as the faculty representative for making this project possible and for her supervision. Furthermore, I thank Dr. Dr. Christoph Leiggener and Prof. Dr. Walter Lesch for their time and guidance. Finally, I am indebted to Dr. Isabelle Wienand for her tireless support over the past three years and constant encouragement. Thank you!

I would like to thank Dr. Eva de Clercq and Dr. Tenzin Wangmo for their mentorship and patience regarding all kinds of questions related to my PhD, as well as Brian Cheng for editing my thesis. My colleagues at the Institute for Biomedical Ethics have been invaluable during the last three years. Moreover, I would like to thank the administration of the Institute for Biomedical Ethics for their support. Finally, Michael Rost: we started our PhD's on the exact same day, and without you, I would not be here today. Thank you so much for everything. It was a pleasure to work with you!

I am truly thankful to my friends who carried me through difficult times in my life and during my PhD. Tolga Dittrich, you always supported me through ups and downs during the past nine years. Thank you so much for being my friend and providing me with the necessary strength and comfort to pursue my passions.

I would also express my gratitude to my entire family and relatives for their constant support and trust in me.

#### **Contributions**

The study was designed and supervised by Prof. Dr. Bernice Elger, head of the Institute for Biomedical Ethics (IBMB) at the University of Basel. She wrote the proposal and received a Grant from the Swiss Academy of Medical Sciences. Dr. Isabelle Wienand and I drafted the study tools and prepared the submission to obtain ethical approval from the ethics committee (Ethikkommission der Zentral- und Nordwestschweiz, approval number 2015-197). Dr. Nadia Pacurari (medical records in the French-speaking part), Sophie Haesen (medical records in the German-speaking part) and I (medical records in the German-speaking part of Switzerland) carried out data collection. Further, Dr. Isabelle Wienand and I conducted relevant data management, additional data collection, and data analysis.

This thesis consists of seven chapters: background (Chapter 1), journal articles (Chapter 2-6, each chapter comprising one article), and general discussion (Chapter 7). In the following, my contributions to each journal article as well as the contributions of the co-authors are presented in detail.

**Rakic M**, Wienand I, Shaw D, Nast R, Elger BS. Autonomy and Fear of Synthetic Biology: How Can Patients' Autonomy Be Enhanced in the Field of Synthetic Biology? A Qualitative Study with Stable Patients. *Science and Engineering Ethics*. Apr 2017;23(2):375-388.

I carried out the data analysis and drafted the initial manuscript. Dr. David Shaw, Prof. Dr. Bernice Elger and Dr. Isabelle Wienand supervised the data analysis, and supported me with the manuscript preparation. Rebecca Nast carried out the interviews. All authors provided critical revisions to the manuscript and approved the final version.

**Rakic M**, Dittrich T, Elger BS, Shaw D. Enhancing Patients' Autonomy by Involving Them in Research Ethics Committees. *International Journal for Quality in Health Care: Journal of the International Society for Quality in Health Care*. Nov 2017;29(7):896-900.

I carried out the data analysis and drafted the initial manuscript with the support of Tolga Dittrich. Dr. David Shaw and Prof. Dr. Bernice Elger supervised the data analysis and supported me with the manuscript preparation. All authors provided critical revisions to the manuscript and approved the final version.

Wienand I, **Rakic M**, Elger B, Shaw D. The Beneficence of Hope: Findings from a Qualitative Study with Gout and Diabetes Patients. *Journal of Bioethical Inquiry*. Jun 2018;15(2):211-218.

Dr. Isabelle Wienand analysed the data and drafted the initial manuscript. Prof. Dr. Bernice Elger, Dr. David Shaw, and I supported with the data analysis and manuscript preparation. I read and provided critical revision for the whole manuscript, as well as permission for submission. All authors provided critical revisions to the manuscript and approved the final version.

**Rakic M**, Escher M, Elger BS, Eckstein S, Pacurari N, Zwahlen S, Wienand I. Feelings of Burden in Palliative Care: A Qualitative Analysis of Medical Records. *Journal of Palliative Care*. Jan 2018;33(1):32-38.

Dr. Isabelle Wienand, Dr. Nadia Pacurari, and I carried out data collection. Dr. Isabelle Wienand and I analyzed the data. I drafted the initial manuscript. Dr. Monica Escher, Dr. Sandra Eckstein, Prof. Dr. Bernice Elger, Dr. Nadia Pacurari, Dr. Isabelle Wienand, and Dr. Susanne Zwahlen supervised the data analysis and supported me with the manuscript preparation. All authors provided critical revisions to the manuscript and approved the final version.

Rakic M, Rost M, Wienand I, Elger BS, Escher M. Advance Care Planning in Swiss Acute Hospitals: A Retrospective Analysis of Medical Records.

Dr. Isabelle Wienand and I carried out data collection. Michael Rost and I analysed the data. I drafted the initial manuscript. Dr. Monica Escher, Prof. Dr. Bernice Elger, and Dr. Isabelle Wienand supervised the data analysis and supported me with the manuscript preparation. All authors provided critical revisions to the manuscript and approved the final version.

#### **Summary**

The steady increase in human life expectancy has been accompanied by a growing prevalence of patients with chronic diseases and multiple medical conditions, and the number of people in need of Palliative Care in Switzerland is expected to continue rising. Few people will die from a sudden death, whereas the majority will face an extended dying process, which will lead to a significant need for formal (e.g. hospital, retirement home) and informal care (e.g. family) throughout the course of a patient's illness. At the same time, autonomy has emerged as a crucial principle in our modern societies. It plays a significant role in the many domains of our lives in which autonomous decision-making is promoted, and autonomy is particularly relevant for elderly patients. Therefore, different approaches on how to improve patients' quality of life and promote their autonomy in clinical settings are increasingly important to discuss. The purpose of this PhD thesis is to integrate empirical and normative analyses using a qualitative and quantitative approach in order to promote respect of elderly patients' autonomy. This PhD thesis consists of seven chapters (background, journal articles, and discussion). The background chapter (Chapter 1) introduces chronic diseases, demographic shift, clinical trials, autonomy, and Palliative Care. Chapter 1 also discusses the research objectives and the methodological approach. Each of the chapters 2-6 consists of a deep dive into one journal article.

Chapter 2 analyses patients' views regarding potential participation in clinical trials of synthetic biology. Our results indicate that information should be communicated with great accuracy and transparency to allay irrational fears of patients and minimize the risk that researchers present facts too optimistically in an effort to persuade patients to participate in the trials. Patients must be adequately informed in order to be able to autonomously decide whether or not to participate.

Chapter 3 explores patients' thoughts about the idea of participating in research ethics committees. If the patients were allowed to participate in the decision-making process, their autonomy would be more respected. Analysis of the interviews revealed a patients' preference towards involvement in research ethics committees. The main motivation for the interviewed patients to participate was the improvement of therapeutic options in the future. Our study adds important knowledge about the idea of patients becoming research ethics committee members by exploring their perceptions of the prospective role.

Chapter 4 focuses on different aspects of hope in the context of human health and well-being and explores the varieties of hope expressed by patients. Three concepts of hope emerged from the interviews: hope as certainty, hope as reflective uncertainty, and hope as self-therapy. Health professionals ought to be more aware of the three concepts of hope and their great potential as a coping strategy for patients. Hope seems to be genuinely beneficent for patients' well-being in a therapeutic and in a research context.

Chapter 5 analyses patients' medical records to determine what is reported about burden and overburden and who seems to be mostly affected. Daily life situations reveal that the concept of autonomy is challenged when the two parties (e.g. patients and their families) have different interests, resources, and needs. Patients often felt burdened by their disease, financial problems, situation at home, and family members' reactions to their disease. Families felt burdened by issues related to patients' medical condition, providing home care, or financial and social aspects.

Chapter 6 discusses different factors determining the application of advance care planning and advance directives. By enabling patients to determine their medical treatments in advance and make their preferences known, advance care planning promotes respect for patients' autonomy. For most patients, a discussion about advance directives, values, and wishes was

documented in the medical records. Of those patients, almost two-thirds engaged in advance care planning. Knowing patients' preferences in advance is needed to base end-of-life decisions on the patients' values and wishes.

Chapter 7 provides a general discussion of the findings derived from the journal articles. More precisely, the discussion coalesces the empirical findings and discusses their ethical significance. Afterwards, limitations and implications for further research are discussed.

#### Zusammenfassung

Die zunehmende Lebenserwartung in der Schweiz wird von einer gestiegenen Prävalenz chronisch kranker Menschen und solcher, welche Palliative Care benötigen, begleitet. Nur wenige Menschen werden eines plötzlichen Todes sterben. Die Mehrheit wird nach langer Krankheit versterben. Dieser lange Krankheitsprozess des Patienten bedarf einerseits vermehrter formeller (bspw. Spital, Altersheim) und anderseits informeller (bspw. Familie) Pflege. Gleichzeitig hat der Begriff der Autonomie als wesentliches Prinzip der Medizinethik an Bedeutung gewonnen. Autonomie spielt eine wichtige Rolle in vielen Domänen unseres Lebens, in welchen wir Entscheidungen (bspw. medizinische Eingriffe) treffen müssen. Des Weiteren bleibt Autonomie für Patienten<sup>1</sup> auch im höheren Alter wichtig. Deshalb werden in den wissenschaftlichen Artikeln verschiedene Ansätze vorgestellt und diskutiert, wie man die Lebensqualität und die Autonomie von älteren Patienten in einem klinischen Setting verbessern kann. Das Ziel der vorliegenden Doktorarbeit ist es, mithilfe von qualitativen, sowie quantitativen Ansätzen, empirische und normative Analysen zu synthetisieren, um den Respekt der Autonomie älterer Patienten zu fördern. Diese Doktorarbeit besteht aus sieben Kapiteln: Hintergrund, fünf wissenschaftliche Artikel, Diskussion. Im 1. Kapitel (Hintergrund) wird näher auf chronische Krankheiten, demografischen Wandel, klinische Versuche und Palliative Care eingegangen. Ausserdem werden die Forschungsziele und Methoden diskutiert. Die Kapitel 2-6 bestehen je aus einem wissenschaftlichen Artikel.

\_

<sup>&</sup>lt;sup>1</sup> Aus Gründen der Lesbarkeit wird nur die männliche Form benutzt. Es sind jedoch alle Geschlechter gleichermassen gemeint.

Im 2. Kapitel werden die Resultate zur Partizipation von Patienten an klinischen Versuchen, welche synthetische Biologie beinhalten, diskutiert. Unsere Resultate deuten darauf hin, dass Informationen mit hoher Genauigkeit und Transparenz kommuniziert werden müssen, um einerseits irrationalen Ängsten seitens der Patienten vorzubeugen und um andererseits zu vermeiden, dass Forscher Fakten zu optimistisch präsentieren, um Patienten zu einer Teilnahme an einem klinischen Versuch zu überreden. Patienten müssen adäquat informiert werden, damit sie autonom entscheiden können, ob sie an einem klinischen Versuch mitmachen wollen oder nicht.

Kapitel 3 untersucht die Meinung von Patienten zu einer möglichen Teilnahme an Forschungsethikkommissionen. Eine aktive Teilnahme der Patienten an solchen Kommissionen würde ihre Autonomie angemessen respektieren. Die Ergebnisse zeigen, dass Patienten sich prinzipiell vorstellen können, Mitglieder einer solchen Kommission zu sein. Die Verbesserung der therapeutischen Optionen wurde als ein Hauptgrund für eine mögliche Teilnahme aufgeführt. Indem auf die Sichtweisen der Patienten eingegangen wird, liefert unsere Studie wichtiges Wissen bezüglich der Idee. Patienten in Forschungsethikkommissionen zu inkludieren.

Kapitel 4 beschäftigt sich mit der Hoffnung im Zusammenhang mit Gesundheit und Wohlbefinden von Patienten. Zudem wird die Vielfalt verschiedener Formen von Hoffnung diskutiert. Aus der Analyse der Patienteninterviews wurden drei Konzepte von Hoffnung herausgearbeitet: Hoffnung als Sicherheit, Hoffnung als reflektierte Unsicherheit und Hoffnung als Selbsttherapie. Gesundheitsfachleute sollten sich dieser drei Arten von Hoffnung bewusst werden und dem damit verbundenen Potential als Bewältigungsstrategie für Patienten. Hoffnung scheint wohltuend für das Wohlbefinden der Patienten zu sein.

In der in Kapitel 5 vorgestellten Studie werden Patientenakten analysiert, um zu eruieren, was über Belastungen und Überbelastungen geschrieben wird und wer am meisten davon betroffen ist. Patienten fühlen sich oft durch ihre Krankheit, finanziellen Probleme, die Situation zuhause und die Reaktion der Angehörigen auf ihre Krankheit belastet. Angehörige fühlen sich oftmals durch den Gesundheitszustand des Patienten, Pflege zuhause, sowie finanzielle und soziale Aspekte belastet. Alltagssituationen zeigen auf, dass das Konzept der Autonomie herausgefordert wird, wenn zwei Parteien (bspw. Patienten und deren Familien) unterschiedliche Interessen, Ressourcen oder Bedürfnisse haben.

Kapitel 6 zeigt verschiedene Faktoren auf, welche das Advance Care Planning, sowie das Ausfüllen einer Patientenverfügung beeinflussen. Durch das vorzeitige Ermöglichen, sich mit ihren medizinischen Behandlungen auseinanderzusetzen, fördert das Advance Care Planning die Achtung der Patientenautonomie. Bei den meisten Patienten fand eine Diskussion zu ihren Wünschen, Werten oder Patientenverfügungen statt. Jene wurden auch in den Patientenakten dokumentiert. Patientenwünsche zu kennen ist Voraussetzung, um Pflege und Behandlung am Ende des Lebens nach den Wünschen und Vorstellungen der Patienten zu gestalten.

Im Kapitel 7 werden die Resultate der wissenschaftlichen Artikel insofern zusammengeführt und diskutiert, als die empirischen Resultate synthetisiert und ihre ethische Signifikanz aufgezeigt wird. Abschliessend werden die Limitationen und Implikationen für weitere Forschung diskutiert.

# Chapter 1 – Background

#### 1 Introduction

#### 1.1 Demographic shift and increase in chronic diseases

In 2050, almost 22% of the global population will be older than 60 years as compared to 12% in 2013. In developed countries, the number of people older than 60 years will increase from 22% in 2013 to 32% in 2050<sup>1</sup>. The growing number of elderly people can be traced back to a decrease in infant mortality, increase in life-expectancy, and a declining birth rate<sup>1</sup>. Consequently, this age group (60 years and older) is growing faster (2% per year) than other age groups<sup>2</sup>. According to the Swiss Federal Statistical Office, Switzerland is not excluded from this population trend, and it is expected that fewer than 20% of people will be younger than 20 years by the year 2060<sup>3</sup>. This demographic shift raises concerns about the increased national burden of chronic disease because such conditions are strongly correlated with older age<sup>2,3</sup>.

The World Health Organization (WHO) defines chronic disease as disease that is not transmitted from one person to another, is of extended duration, and has a relatively slow progression. Cardiovascular diseases, cancers, respiratory diseases, and diabetes comprise four primary types of chronic diseases<sup>4</sup>. Less than 5% of the Swiss population will die from a sudden death; the vast majority of the population will die after a chronic disease. More precisely, 30-40% of patients will die after a disease duration of 8-10 years<sup>3</sup>. Cardiovascular diseases (32%) and cancers (26%) are the most frequent causes of death in Switzerland<sup>5</sup>. Furthermore, the number of elderly people suffering from more than one disease (multimorbidity) will increase<sup>3</sup>. According to statistical prognoses, the demographic shift will lead to an increase in number of people in need of care from 125,000 in 2010 to 182,000 in 2030 in Switzerland<sup>3</sup>.

On the one hand, human life expectancy is steadily increasing, chronic diseases are on the rise, and the expected number of people in need of care in Switzerland will continue to increase<sup>1-3</sup>. On the other hand, only a minority will die from a sudden death<sup>5</sup>, whereas the majority will face a lingering dying process, which will lead to an increase in both formal (e.g. retirement home) and informal care (e.g. family) throughout the course of the patients' disease<sup>3</sup>. Therefore, different approaches to improve patients' quality of life and promote their autonomy in clinical settings (throughout the course of their disease) will be discussed in the following sections as autonomy will become increasingly important and relevant to clinical care in the future<sup>3</sup>.

#### 1.2 Treatment options in light of chronic diseases

Patients who suffer from one or multiple chronic diseases do not die immediately<sup>10</sup>. Nevertheless, studies report a shorter life expectancy and decreased quality of life as compared to healthy persons<sup>6,7</sup>. A possible solution to address the decreased quality of life is the application of novel technologies, such as synthetic biology (SB), to synthetically reprogram metabolic systems in the body as shown in a preclinical study with mice<sup>8</sup>. SB is currently used to enable production of specific chemicals (e.g. antibiotics)<sup>9</sup>. With these developments<sup>8</sup>, it may soon be possible to modify cells that are able to produce a certain substance that induces insulin production in the body and decreases the blood sugar concentration of a diabetic patient. With such a medical therapy, the patient does not need to regularly inject insulin and hence, would improve his quality of life. However, before a new treatment is approved and available for patients, it must be tested in first-in-human (FIH) trials to ensure safety and efficacy in humans<sup>10</sup>. To date, terminal cancer patients comprise the primary target population for FIH trials because of the absence of therapeutic alternatives and the hope that experimental treatments will help them to survive<sup>10</sup>. Given the limited remaining life spans of these terminal patients, clinical studies with severely ill patients (e.g.

refractory cancer) can never fully discern the long-term risks<sup>11</sup>. Additionally, patients with chronic diseases (e.g. gout, diabetes, and certain cancers) might benefit from the medications developed through clinical trials as compared to terminal cancer patients<sup>6,7,12,13</sup>. For this target group, it might be possible to treat some of the chronic diseases with synthetically reprogrammed metabolic systems<sup>8</sup>. This would lead to an increase in patients' life expectancy and quality of life.

Nonetheless, research ethics committees, which evaluate clinical trials, tend to reject studies involving stable patients because of high risk (e.g. risks of potential damage from clinical trial overweigh possible benefits) to these patients to participate<sup>14</sup>. This attitude has been criticized as unjustified paternalism<sup>15</sup>. Lack of consideration for patients' willingness to accept risks and the paternalistic attitude of research ethics committees<sup>14,16</sup> appear to create an anachronistic situation regarding autonomy. This fact could lead to a restriction of the patient's right to make autonomous decisions about participation in clinical trials (e.g. FIH trial). Therefore, shifting from paternalistic decision-making to a patient-centred, shared decision-making model that includes patients in research ethics committees would increase patients' ability to make fully autonomous health decisions<sup>17</sup>. Nevertheless, the possibility of unpredictable side-effects (e.g. long-term effects) of biotechnological interventions (e.g. SB) in humans needs to be articulated clearly to study participants, for example through informed consent (IC), to ensure autonomous decision-making.

#### 1.3 Palliative Care

Another important aspect of the demographic shift is that chronic diseases often lead to complex physical and psychological end-of-life situations. Palliative Care (PC) addresses the whole spectrum of patients' needs and suffering<sup>3</sup>. Therefore, chronic diseases (e.g. cancers) will also increase the number of people in need of PC and thus, PC will become more important in the future<sup>3</sup>. According to the WHO, PC is an interdisciplinary and holistic approach that encompasses several domains of care (physical, social, psychological, and spiritual)<sup>18</sup>. Moreover, PC not only focuses on the requirements of patients but also on the needs of their families, friends, and significant others. Therefore, the aim of PC is to offer services to help patients and their families cope not only with the patient's illness, but also with the families' own needs<sup>18</sup>. Thus, patients suffering from a life-threatening disease should have the possibility to receive PC, which prioritizes quality of life and prevention of suffering<sup>3</sup>.

The Swiss Federal Office for Public Health estimates that 40,000 people are currently in need of PC and that this number will increase to 53,000 people by 2032<sup>3</sup>. A representative poll showed that half of the surveyed people knew what PC entails. Moreover – when being asked about their attitude regarding PC – around 80% of the surveyed people could envision receiving PC treatment if they suffer from an incurable disease<sup>19</sup>. The wish for PC meets an important need, namely, the need to live as autonomous and self-determined a life as possible<sup>3,20</sup>. This increased request for self-determination has led to changes in the Adult Protection Law to promote PC and patients' autonomy<sup>3,21</sup>. However, a survey from 2011 indicated that the possibility to ensure patients' autonomy was seen as the main reason to justify assisted suicide<sup>22</sup>, whereas PC or Advance Directives (ADs) were rather unknown as tools to promote one's autonomy<sup>3,22</sup>. To meet society's wish to live as autonomously as possible, but also to react to the demographic shift, Switzerland launched a national strategy

on PC with the goal to implement PC in the Swiss health care sector to make PC available for patients in need of it. Furthermore, with the launch of the national strategy on PC, patients should be given an alternative means to assisted suicide through which they can exercise their autonomy<sup>3</sup>.

To support decision-making in PC situations, one common and recommended tool is the writing of legally binding ADs along with advance care planning (ACP)<sup>23,24</sup>. ACP ensures that the patient's wishes and preferences are known before the time at which the patient becomes unable to choose medical treatments (e.g. due to lack of mental capacity)<sup>23,24</sup>. By enabling patients to proactively determine their medical treatments in advance, ACP promotes respect for patients' autonomy<sup>25</sup>, which is an important defining principle of PC<sup>20</sup>. Moreover, the new Adult Protection Law in Switzerland enables and encourages competent adults to write ADs and to appoint a surrogate decision-maker in case of their lack of mental capacity<sup>21</sup>.

#### 1.4 Theoretical Background: Autonomy

Autonomy means self-rule (Greek autos = self and nomos = rule)<sup>26</sup>. Since the 1960's, there has been a greater emphasis on the notion of 'autonomy' both in society and medical ethics<sup>27,28</sup>. As it is common among other researchers in medical ethics, I use autonomy/self-determination interchangeably<sup>26,29</sup>.

The evolution of autonomy can be tracked back to mostly two developments. First, the increased rejection of paternalism in medicine due to public changes (e.g. the responsible citizen), and second, the plurality in common understandings as to what a good life entitles<sup>28</sup>. Therefore, every individual may decide upon his or her values on what a good life might be<sup>28</sup>. In practical philosophy, several concepts of autonomy exist based on different philosophical schools<sup>29</sup>. These concepts vary insofar as they have divergent criteria that must be fulfilled in

order to view a person as an autonomous being<sup>29</sup>. In what follows, two of these concepts are discussed, namely the Kantian and the Utilitarian philosophical views on autonomy, because these two models in modern philosophy have significantly shaped the current notion of autonomy<sup>28</sup>.

Kant's understanding of autonomy (see *Grundlegung zur Metaphysik der Sitten*<sup>30,31</sup>, 1785) posits that autonomous persons are fundamentally free, that is they are able to refrain their individual wishes and preferences and act according to the universal principles of rationality<sup>28,30</sup>. That is, the ability to orient their own will towards the Categorical Imperative, according to which free choices ought to be made based on universalizable maxims<sup>29</sup>. The first formula of the Categorical Imperative reads so: "Act only in accordance with that maxim through which you at the same time can will that it become a universal law." <sup>30, p. 66</sup>. Furthermore, the second formula of the Categorical Imperative underlines the priceless value of human persons: "So act that you use humanity, as much in your own person as in the person of every other, always at the same time as an end and never merely as means" <sup>30, p. 66</sup>. Taken together, no person should be merely instrumentalized and every person has intrinsic value<sup>28</sup>. These values still seem relevant in modern medical ethics, as shown by the indispensable utilisation of IC that aims to ensure patients autonomy<sup>32</sup>.

The Utilitarian understanding of autonomy (see John Stuart Mill, *On liberty*<sup>28,33</sup>, 1859) emphasizes the individual freedom of everyone to live in accordance to their own world views. Nevertheless, this freedom is restricted when a third party is being put at risk. According to Mill's definition of autonomy, putting oneself at risk is not a reason to neglect someone's autonomy and act paternalistically<sup>28</sup>.

Depending on the philosophical approach (e.g. Kantian or Utilitarian), the definition of autonomy might differ. However, both agree that two prerequisites must be fulfilled for a

person to be autonomous: liberty and agency. The first condition means that a person can decide in an independent manner, that is, free from others' influence. Agency implies that a person has the capacity to act in an rational manner<sup>26</sup>. To find a consensus of these different ethical theories of autonomy, Beauchamp and Childress introduced four prima facie principles as guiding norms for the new discipline of bioethics: autonomy, non-maleficence, beneficence, and justice<sup>26,34</sup>. They have acquired an utmost importance in the field of medical ethics<sup>28</sup>. Beauchamp and Childress defined autonomy as follows: "At a minimum, personal autonomy encompasses self-rule that is free from both controlling interference by others and from certain limitations that prevent meaningful choice, such as inadequate understanding, 26, p. 100. Nevertheless, it needs to be said that the individualistic approach (e.g. the patient is the sole decision-maker) of Beauchamp and Childress regarding autonomy has been widely criticized, mostly because it fails to consider the social and cultural determinants of one's autonomy, that is their importance on one's lives and decision-making processes<sup>35,36</sup>. This model is known as relational autonomy, for an in-depth discussion on relational autonomy, see also chapter 7. A possible problem of the individualistic approach might be when elderly patients are faced with stereotypical assumptions regarding their autonomy (e.g. the older the patient, the less competent and less autonomous)<sup>37</sup>. Therefore, social, cultural, and existential aspects (e.g. patients' lifeworld) should also be taken into account, especially when patients face special situations (e. g. disease, ageing) that might affect their autonomy<sup>37,38</sup> (for an indepth discussion on lifeworld, see also chapter 7).

In order to ensure the respect for patients' autonomy regarding medical treatments, IC has emerged as a prominent consideration in medical ethics since the late 1940s<sup>26</sup>. Moreover, IC empowers patients to decide for or against a certain medical treatment after being informed<sup>26,29</sup>. In addition, IC requires more than a sole provision of information regarding medical treatment<sup>26</sup>. In order to be able to make an autonomous decision regarding a medical

treatment, the following prerequisites of an IC must be fulfilled: "disclosure" (e.g. adequate delivery of information so that patients can make an informed decision), "understanding" (e.g. nature and purpose of treatment), "voluntariness" (e.g. patient agrees to certain treatment without undue influenced from another person), "competence" (e.g. capable of understanding consequences of the consent and able to make a free choice), and "consent"<sup>26</sup>, p. 79

#### Conclusion

The expected demographic shift toward older age will be accompanied by an increase in the prevalence of chronic diseases, multimorbid patients, and deaths in clinical settings (e.g. hospital, retirement home)<sup>3</sup>. At the same time, autonomy has become more crucial in our society<sup>3</sup> and plays a significant role in many domains (e.g. legal) that aim to promote autonomous decision-making<sup>37</sup>. For elderly patients, autonomy remains relevant, and they will likely not accept limitations placed on their autonomy <sup>39</sup>. Because autonomy has achieved great significance in society<sup>3</sup> and in medical ethics (e.g. right to make own decisions<sup>26</sup> or not being used merely as a means to an end<sup>30</sup>), the autonomous decision-making should be respected and promoted<sup>35</sup>. Thus, the purpose of this PhD thesis is to integrate empirical and normative analyses in order to support respecting elderly patients' autonomy in clinical settings. More precisely, the thesis' aim is to identify and discuss possible hindering and promoting determinants of autonomous decision-making to better ensure patients' autonomy. One promoting factor of patients' autonomy is active participation in decisionmaking processes (e.g. through research ethics committees). Research ethics committees facilitate patient access to and comprehension of medical information. Moreover, through precise and tailored information (e.g. specific IC), patients have the opportunity to competently decide on their own and give their consent whether to participate in clinical trials or to receive specific treatment. What is more, this process ensures voluntary participation of patients. In case of lack of mental capacity, their future autonomy can be restored through the active promotion of ACPs<sup>23,24</sup> and thus, ensures that patients' autonomy will be respected.

Hindering factors include – amongst others – situation of dependence (e.g. fear of putting too much burden on family members), old age (e.g. lack of mental capacity), unrealistic hope (e.g. possible barrier to understand the medical situation and with that to give an IC) – in which it becomes difficult for the patients to express their preferences autonomously. Besides, disclosure, voluntariness or consent might be diminished. In addition, severe illness and fear of unknown treatments (e.g. within SB) could lead to deprivation of patients' autonomy as patients might not be able to decide in an independent manner.

## 2 Research Objectives and Empirical Methods

#### 2.1 Research Objectives

This thesis is part of two larger research projects entitled "Ethical issues of cutting edge biotechnology: Embedded interdisciplinary risk benefit evaluation of first-in-human trials in synthetic biology and nanomedicine" and "Respect for patient self-determination as quality indicator in palliative care: current state, problems and solutions in acute care hospitals." The nature of each research projects has a clear ethical orientation and aims to contribute to a better recognition and respect for patients' preferences and values.

The first research project aimed to obtain information on how chronically ill patients perceive SB in general. Moreover, it sought to evaluate the willingness of patients to participate in FIH trials, including the arguments provided for and against such participation. As already discussed in the introduction, there is a research gap regarding patients' participation in clinical trials of new technologies and concerning attitudes of stable patients suffering from chronic disease with only slightly impaired life expectancy as compared to near-death patients<sup>16</sup>. Evidence of fears towards SB in the general population<sup>40</sup> should motivate research to study patients' attitudes towards clinical trial participation. These attitudes can be seen as an indicator for their preferences and tendencies to participate in research involving SB, as well as means to promote and ensure the respect of their autonomy.

The objective of the second research project was to evaluate the quality of PC in three acute care hospitals in Switzerland based on the respect for patients' autonomy (their values and wishes). The project aimed to describe and analyze autonomy-enhancing activities – provided by the hospitals (e.g. ACP) and desired by the patients (e.g. ADs) –, patients' characteristics (e.g. age, disease), and satisfaction with PC consultations. To our knowledge, there has been no systematic research to evaluate elderly patients' autonomy in PC in Switzerland. The few

existing studies focus on the lack of health care personnel in the evaluation of patients' values<sup>41,42</sup> or decision-making in intensive care units<sup>43,44</sup>.

To fulfill these objectives, this thesis aims to answer the following research questions:

- 1. What are patients' opinions about synthetic biology and first-in-human trials using medical applications of SB?
- 2. What are patients' attitudes towards participation in RECs as established members?
- 3. What kinds of hope are being expressed by patients regarding first-in-human trials using medical applications of SB?
- 4. What kinds of burden are being expressed by PC patients in acute hospitals?
- 5. Which factors determine the application of ACP and/or the completion of an AD?
- 6. How does ACP occur in clinical settings?

#### 2.2 Empirical Methods

The PhD project consists of 7 chapters that stem from two larger research projects, which were carried out between 2012 and 2018. The project employed a qualitative and quantitative methods design: a qualitative part from the first research project in which 36 patients suffering from diabetes or gout were interviewed to explore their views and preferences regarding participation in clinical trials (e.g. SB). For the second project, we performed a quantitative and qualitative analysis of retrospectively collected data from 300 PC patients' medical records. The exact methods are described in greater detail in the methods sections of each article included in this thesis (see Chapters 2-6).

#### References

- 1. United Nations. World Population Ageing. World Population Ageing 2013; <a href="http://www.un.org/en/development/desa/population/publications/pdf/ageing/WorldPo">http://www.un.org/en/development/desa/population/publications/pdf/ageing/WorldPo</a> pulationAgeingReport2013 Accessed 05 December 2017.
- 2. Divo MJ, Martinez CH, Mannino DM. Ageing and the epidemiology of multimorbidity. *The European respiratory journal*. 08/19 2014;44(4):1055-1068.
- 3. Bundesamt für Gesundheit (BAG) und Schweizerische Konferenz der kantonalen Gesundheitsdirektorinnen und -direktoren (GDK). Nationale Strategie Palliative Care 2013–2015. 2012; <a href="https://www.bag.admin.ch/dam/bag/de/dokumente/nat-gesundheitsstrategie-palliative-care/nationale-strategie-palliative-care/nationale-strategie-palliative-care/nationale-strategie-palliative-care/nationale-strategie-palliative-care/nationale-2015.pdf">https://www.bag.admin.ch/dam/bag/de/dokumente/nat-gesundheitsstrategie-palliative-care/nationale-strategie-palliative-care/nationale-strategie-palliative-care/nationale-2015.pdf</a> Accessed 05 December 2017.
- 4. WHO. Noncommunicable diseases. 2017;

  <a href="http://www.who.int/topics/noncommunicable\_diseases/en/">http://www.who.int/topics/noncommunicable\_diseases/en/</a> Accessed 05 December 2017.
- 5. Bundesamt für Statistik. Todesursachenstatistik 2015. 2017; <a href="https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm">https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm</a>
  <a href="https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm">https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm</a>
  <a href="https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm">https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm</a>
  <a href="https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm">https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm</a>
  <a href="https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm">https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm</a>
  <a href="https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.3742835.htm">https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit.assetdetail.assetde
- 6. Franco OH, Steyerberg EW, Hu FB, Mackenbach J, Nusselder W. Associations of diabetes mellitus with total life expectancy and life expectancy with and without cardiovascular disease. *Archives of Internal Medicine*. 2007;167(11):1145-1151.
- 7. Wandell PE. Quality of life of patients with diabetes mellitus. An overview of research in primary health care in the Nordic countries. *Scandinavian Journal of Primary Health Care*. Jun 2005;23(2):68-74.

- 8. Ye H, Baba MD-E, Peng R-W, Fussenegger M. A Synthetic Optogenetic Transcription Device Enhances Blood-Glucose Homeostasis in Mice. *Science*. 2011;332(6037):1565-1568.
- 9. Benner SA, Sismour AM. Synthetic biology. *Nature reviews. Genetics.* Jul 2005;6(7):533-543.
- 10. Dresser R. First-in-Human Trial Participants: Not a Vulnerable Population, but Vulnerable Nonetheless. *The Journal of Law, Medicine & Ethics: a Journal of the American Society of Law, Medicine & Ethics.* 2009;37(1):38-50.
- 11. Murren JR, DiStasio SA, Lorico A, et al. Phase I and pharmacokinetic study of novobiocin in combination with VP-16 in patients with refractory malignancies.

  \*Cancer J. 2000 Jul-Aug 2000;6(4):256-265.
- 12. Chen JH, Lan JL, Cheng CF, et al. Effect of Urate-lowering Therapy on the Risk of Cardiovascular Disease and All-cause Mortality in Patients with Gout: A Casematched Cohort Study. *The Journal of Rheumatology*. Sep 2015;42(9):1694-1701.
- 13. Bauer UE, Briss PA, Goodman RA, Bowman BA. Prevention of chronic disease in the 21st century: elimination of the leading preventable causes of premature death and disability in the USA. *The Lancet*. 2014/07/05/ 2014;384(9937):45-52.
- 14. Shaw D. The right to participate in high-risk research. *The Lancet.* 2014/03/15/2014;383(9921):1009-1011.
- 15. Edwards SJ, Kirchin S, Huxtable R. Research ethics committees and paternalism.

  \*Journal of Medical Ethics. Feb 2004;30(1):88-91.
- 16. Shaw D, Elger B. Putting patients on research ethics committees. *Journal of the Royal Society of Medicine*. Aug 2014;107(8):304-307.

- 17. Charles C, Gafni A, Whelan T. Decision-making in the physician-patient encounter: revisiting the shared treatment decision-making model. *Social Science & Medicine* (1982). Sep 1999;49(5):651-661.
- 18. World Health Organization. WHO Definition of Palliative Care. 2002; http://www.who.int/cancer/palliative/definition/en/ Accessed 05 November 2017.
- 19. Bundesamt für Gesundheit (BAG). Zusammenfassung der Ergebnisse zur Studie "Palliative Care". 2009; <a href="https://www.bag.admin.ch/bag/de/home/service/publikationen/forschungsberichte/forschungsberichte-palliative-care/bevoelkerungsbefragung-palliative-care.html">https://www.bag.admin.ch/bag/de/home/service/publikationen/forschungsberichte/forschungsberichte-palliative-care/bevoelkerungsbefragung-palliative-care.html</a>
  Accessed 15 December 2017.
- 20. World Health Organization Regional Office for Europe. Better Palliative Care for Older People. 2004;
  <a href="http://www.euro.who.int/">http://www.euro.who.int/</a> data/assets/pdf\_file/0009/98235/E82933.pdf Accessed 15
  May 2017.
- 21. Bundesamt für Justiz. Neues Erwachsenenschutzrecht tritt am 1. Januar 2013 in Kraft.

  2011; <a href="https://www.bj.admin.ch/bj/de/home/aktuell/news/2011/ref\_2011-01-12.html">https://www.bj.admin.ch/bj/de/home/aktuell/news/2011/ref\_2011-01-12.html</a>
  Accessed 07 December 2017.
- 22. Der Bundesrat. Palliative Care, Suizidprävention und organisierte Suizidhilfe. 2011; <a href="https://www.ejpd.admin.ch/dam/data/bj/gesellschaft/gesetzgebung/archiv/sterbehilfe/ber-br-d.pdf">https://www.ejpd.admin.ch/dam/data/bj/gesellschaft/gesetzgebung/archiv/sterbehilfe/ber-br-d.pdf</a> Accessed 20 February 2018.
- 23. SAMW. Patientenverfügungen. 2013; <a href="https://www.rwi.uzh.ch/dam/jcr:f7c5b696-6fab-4e8a-9004-18800a87c0e7/SAMW-RL%20Patientenverf%C3%BCgung.pdf">https://www.rwi.uzh.ch/dam/jcr:f7c5b696-6fab-4e8a-9004-18800a87c0e7/SAMW-RL%20Patientenverf%C3%BCgung.pdf</a>
  Accessed 08 May 2017.
- 24. SAMW. Palliative Care. 2017; <a href="www.samw.ch/dam/jcr:d7ae1138-0213-481b.../richtlinien-samw-palliative-care.pdf">www.samw.ch/dam/jcr:d7ae1138-0213-481b.../richtlinien-samw-palliative-care.pdf</a> Accessed 08 May 2017.

- Singer PA, Robertson G, Roy DJ. Bioethics for clinicians: 6. Advance care planning.
   Canadian Medical Association Journal. December 15, 1996 1996;155(12):1689-1692.
- 26. Beauchamp TL, Childress, J.F. *Principles of Biomedical Ethics*. 7 ed. New York: Oxford University Press; 2009.
- 27. O'Neill O. *Autonomy and Trust in Bioethics*. Cambridge: Cambridge University Press; 2002.
- 28. Maio G. *Mittelpunkt Mensch: Ethik in der Medizin*. korrigierter Nachdruck der 1. Auflage. Stuttgart: Schattauer; 2012.
- 29. Bobbert M, Werner MH. Autonomie/Selbstbestimmung. In: Lenk C, Duttge G, Fangerau H, eds. *Handbuch Ethik und Recht der Forschung am Menschen*. Berlin, Heidelberg: Springer; 2014:105-114.
- 30. Allen W. Wood. Kantian Ethics. New York: Cambridge University Press; 2008.
- Kant I. Groundworks of the Metaphysics of Morals [Grundlegung zur Metaphysik der Sitten]. Second edition, edited by M. Gregor and J.Timmermann, introduction by C.
   M. Korsgaard: Cambridge: Cambridge University Press; 2012.
- Faden RR, Beauchamp TL, King NMP. A History and Theory of Informed Consent.
   Oxford: Oxford University Press; 1986.
- 33. John Stuart Mill. *On Liberty and Other Essays*. Edited and with an Introduction and Notes by J. Gray: Oxford: Oxford University Press; 1998.
- 34. Gillon R. Medical ethics: four principles plus attention to scope. *BMJ : British Medical Journal*. 1994;309(6948):184-188.
- 35. Fjordside S, Morville A. Factors influencing older people's experiences of participation in autonomous decisions concerning their daily care in their own homes:

- a review of the literature. *International Journal of Older People Nursing*. 2016;11(4):284-297.
- 36. Mackenzie C, Stoljar N. Relational Autonomy: Feminist Perspectives on Autonomy, Agency, and the Social Self. New York: Oxford University Press; 2000.
- 37. Sherwin S, Winsby M. A relational perspective on autonomy for older adults residing in nursing homes. *Health Expectations*. 2011;14(2):182-190.
- 38. Entwistle VA, Carter SM, Cribb A, McCaffery K. Supporting Patient Autonomy: The Importance of Clinician-patient Relationships. *Journal of General Internal Medicine*. 2010;25(7):741-745.
- 39. Rodgers V, Welford C, Murphy K, Frauenlob T. Enhancing autonomy for older people in residential care: what factors affect it? *Int J Older People Nurs*. Mar 2012;7(1):70-74.
- 40. Link H-J. Playing God and the Intrinsic Value of Life: Moral Problems for Synthetic Biology? *Science and Engineering Ethics*. June 01 2013;19(2):435-448.
- 41. Ankuda CK, Block SD, Cooper Z, et al. Measuring critical deficits in shared decision making before elective surgery. *Patient Education and Counseling*. Mar 2014;94(3):328-333.
- 42. Dev S, Abernethy AP, Rogers JG, O'Connor CM. Preferences of people with advanced heart failure a structured narrative literature review to inform decision making in the palliative care setting. *American Heart Journal*. Sep 2012;164(3):313-319.e315.
- 43. Lustbader DR, Nelson JE, Weissman DE, et al. Physician reimbursement for critical care services integrating palliative care for patients who are critically ill. *CHEST*. Mar 2012;141(3):787-792.

44. Scheunemann LP, McDevitt M, Carson SS, Hanson LC. Randomized, controlled trials of interventions to improve communication in intensive care: a systematic review. *CHEST*. Mar 2011;139(3):543-554.

Chapter 2

Autonomy and Fear of Synthetic Biology: How Can Patients' Autonomy Be

Enhanced in the Field of Synthetic Biology? A Qualitative Study with

**Stable Patients** 

Reprinted with the permission of © 2016 Springer Nature

Citation: Rakic M, Wienand I, Shaw D, Nast R, Elger BS. Autonomy and Fear of Synthetic

Biology: How Can Patients' Autonomy Be Enhanced in the Field of Synthetic Biology? A

Qualitative Study with Stable Patients. Science and Engineering Ethics. Apr 2017;23(2):375-

388.

(Impact Factor: 2.229)

#### **Abstract**

We analyzed stable patients' views regarding synthetic biology in general, the medical application of synthetic biology, and their potential participation in trials of synthetic biology in particular. The aim of the study was to find out whether patients' views and preferences change after receiving more detailed information about synthetic biology and its clinical applications. The qualitative study was carried out with a purposive sample of 36 stable patients, who suffered from diabetes or gout. Interviews were transcribed verbatim, translated and fully anonymized. Thematic analysis was applied in order to examine stable patients' attitude towards synthetic biology, its medical application, and their trial participation. When patients were asked about synthetic biology in general, most of them were anxious that something uncontrollable could be created. After a concrete example of possible future treatment options, patients started to see synthetic biology in a more positive way. Our study constitutes an important first empirical insight into stable patients' views on synthetic biology and into the kind of fears triggered by the term "synthetic biology." Our results show that clear and concrete information can change patients' initial negative feelings towards synthetic biology. Information should thus be transmitted with great accuracy and transparency in order to reduce irrational fears of patients and to minimize the risk that researchers present facts too positively for the purposes of persuading patients to participate in clinical trials. Potential participants need to be adequately informed in order to be able to autonomously decide whether to participate in human subject research involving synthetic biology.

**Key words**: synthetic biology, autonomy, stable patients, first-in-human trial

#### Introduction

There is not yet any standard definition of synthetic biology (SB) on which most scientists agree, but according to Balmer and Martin, the main aims of SB can be described as the following: "1) The production of minimal living genomes; 2) the design of interchangeable parts that can be assembled into pathways for the fabrication of novel components; 3) the construction of entirely artificial cells; and 4) the creation of synthetic molecules"<sup>1, p. 3</sup>.

To sum up, SB uses engineering principles to consciously design biological systems or living organisms<sup>1</sup>. While several studies in animals have been promising<sup>2,3</sup>, clinical trials of more advanced phases (especially clinical phases II, III) are not yet common<sup>4</sup>. A possible barrier to clinical trials could be diffuse fears of SB and other types of cutting edge biotechnology as we find by and large in society's reactions to these new technologies<sup>5</sup>. Although most of the new devices have been tested in preclinical phases only, some have already successfully completed clinical phase I trials: for example a drug-sensing hydrogel used to control the release of an antibiotic (novobiocin), which is thought to be important for cancer treatment<sup>6,7</sup>. What are the reasons for public fears of SB? In his analysis Hans-Jürgen Link explores the fears of SB. He shows that several types of fear are based on ethical considerations related partly to religious worldviews, and partly to fears of risks, as illustrated by the key words "playing God" and "creation of artificial life." These fears seem to arise from the following opinions: in popular publications related to SB, it is often mentioned that scientists are trying "to play God." The common understanding – based upon religious beliefs as well as ethical arguments – is that it is morally wrong, because humans are not supposed to create new forms of life. A common claim is that life, as a gift of God (God is the giver of life), is intrinsically valuable, and may not be altered. According to this theological point of view, humans are considered guardians of God's creation<sup>8</sup>. Therefore SB appears ominous, inasmuch as it disturbs the relation between creator and creature. What is more, SB blurs the distinction between natural i.e. created by God and artificial i.e. fabricated by man<sup>9</sup>. Thus, scientists are asked to stop doing research based on SB in order to stay away from this slippery slope<sup>9</sup>. Moreover, the importance of unpredictable side-effects in biotechnological interventions in humans ought to be seriously taken into account, and explained clearly to the study participants. Otherwise, as the German philosopher Hans Jonas argues present and future generations will be restricted in or even deprived of their autonomy<sup>10</sup>. Indeed, respect for the autonomy of patients and research participants is central for modern bioethics. The ethical principle of patients' autonomy or self-determination is violated when participants are not given full and transparent information about the clinical trials. Beauchamp and Childress define autonomy as follows: "At a minimum, personal autonomy encompasses self-rule that is free from both controlling interference by others and from certain limitations that prevent meaningful choice, such as inadequate understanding", 11, p. 100. Therefore, a patient's level of autonomy is intrinsically linked to the amount and quality of information she or he received about possible adverse events. That is, no "meaningful choice" can be made on the basis of partial, limited or inaccurate information. Lack of clear information could lead to ethically unjustified restriction or deprivation of autonomy<sup>10</sup>. Participants ought to be adequately informed in order to be able to decide autonomously whether they agree to participate in human subject research involving medication or devices developed by cutting edge biotechnology or SB<sup>10,11</sup>. Since the designation "SB" implies a specific type of modern biotechnology that is known to have solicited controversy<sup>1,5</sup>, not mentioning SB to the patients would mean withholding information from them that could be relevant for their decision-making. To sum up, the ethical principle of patients' autonomy in connection with SB is a challenge for researchers, especially with regard to first-in-human trials (FIH) as well as more advanced clinical phases (II, III). Indeed, how can they both respect the participants' autonomy and proceed with their scientific research in SB? In other words, to which extent

are autonomy and research in unknown fields compatible? FIH trials are performed to ensure the security of interventions in humans<sup>12</sup>. The group of near-death cancer patients is a target group for FIH trials, given the absence of therapeutic alternatives and the hope that new treatments will help them survive<sup>12</sup>. However, in studies, involving end-stage cancer patients<sup>6</sup>, long-term risks will never be fully known. There is a clear research gap concerning attitudes of stable patients suffering from chronic disease with only somewhat impaired life expectancy as compared to near-death patients<sup>13</sup>. More specifically, the evidence that fears exist in the general population<sup>9</sup> should motivate research to know more about stable patients' attitudes towards clinical trial participation. These attitudes can be seen as an indicator for their preferences and tendencies to participate in research involving SB. Indeed, false beliefs and non-justified fears are not only a barrier to patients' autonomy, but they may present serious obstacles to indispensable FIH studies on new SB devices.

In order to understand how patients' opinion about SB might affect their willingness to participate in FIH trials using medical applications of SB, we carried out interviews with 36 stable patients with the aim to a) analyze their opinion on research and technology in general, b) examine their opinion about SB that might affect their willingness to participate in FIH trials, and c) determine whether their views and preferences change after having received more detailed information about SB and its potential clinical applications.

### **Methods**

We employed an empirical research design to explore stable patients' attitudes towards SB and clinical trial participation using medical applications of SB. After a literature search for translational research in cutting edge biotechnology (including in particular SB), we developed an interview guide. Based on the previous literature research, we selected the topics which should be included in our interview guide (e.g. examples of clinical trials,

patients' current health status, etc.). We were then able to develop a realistic example of a hypothetical medical application in the field of SB.

### Study tool

The semi-structured interview guide consisted of ten open questions, each with several subquestions. This paper investigates the following questions and sub-questions as featured in Tables 1, 2 and 3.

### Table 1. Question and sub-questions about research and technology in general

- 1. To increase treatment options for diseases, much research is done in laboratories and hospitals. Let us go a step back. Research is done in almost every area of life. What do you think of technical development and research in general? Are you more neutral or how would you consider yourself?
  - 1.1 Would you consider yourself rather critical or supportive of technology?
  - 1.2 Are there areas of life, where you welcome technology development and others, where you reject such development?

Semi-structured interview about research and technology in general.

### Table 2. Question and sub-questions about knowledge of and opinion on SB.

- 2. A very new research branch is SB. Have you already heard the phrase "synthetic biology?"
  - 2.1 If yes: How do you imagine it? How did you hear about SB? Afterwards explain briefly.
  - 2.2 If not: I wrote an explanation that you can read through. Give patient "card 2" for illustration.
  - 2.3 Subsequently, give patient "graphic 1 "for illustration and explain shortly.
  - 2.4 Have you questions about this?
  - 2.5 Reading about it [SB], what do you think? What do you find good, and what not?
  - 2.6 If not mentioned: What do you think of SB as a treatment option for diseases? What do you find good, and what not?

Semi-structured interview about knowledge of and opinion on SB (Readers can request Fig. 1 [Fig. 1 displaying manipulation of human cell] from the corresponding author).

### Table 3. Question and sub-questions about possible future treatments in the field of SB.

### 3. I would like to do a short thought experiment with you.

- a. hypothetical example for gout patients (see Annex)
- b. hypothetical example for diabetes patients (see Annex)
- 3.1 When you hear it [hypothetical example of a medical treatment in the field of SB], what kind of thoughts go through your head? What do you find good about it, what not so good?

Semi-structured interview about future treatments in the field of SB.

The hypothetical medical application presented during the interview depicted the implantation of autologous cells that had been modified by using SB technologies. Patients were informed that the implanted cells might be able to cure diabetes or gout. In that case, no further medication would be needed.

Ethical approval was obtained from the cantonal research ethics committee (Ethikkommission Nordwest- und Zentralschweiz: Nr. EK 193/12). We carried out a pilot study in 2013 to ensure participants' comprehension. As a result, two slight modifications were made. First, the interview structure was put in a chronological order, and second, wording of the questions was adapted to improve accurate understanding.

### Sampling and data collection

The qualitative study involved a purposive sample of 36 patients. We recruited participants directly in the hospital during their routine check-ups. The recruitment duration lasted six months in 2014/2015. Written informed consent was obtained from each individual participant included in the study. Average interview duration was 44 minutes (min. 16 minutes; max. 1 hour and 42 minutes). All interviews were conducted in Swiss-German by the same researcher. Interviews were conducted at the patient's home, our Institute or in the hospital, where a separate room was available. Interviews were transcribed verbatim,

translated into High German and fully anonymized. Quotes were later translated into English. Finally, patients' names were anonymized.

### **Analysis**

Thematic analysis was applied to the interview transcripts. We did not use a pre-existing theoretical framework while reading the interviews, as we were aiming for a bottom-up method. A bottom-up approach means that the themes we identified in our interviews were explicitly linked to the data we analyzed. Possible themes were identified via a stepwise procedure. First, all authors read the interviews in order to obtain preliminary themes in an inductive way. Afterwards, these themes were discussed among the authors. Second, we discussed the previously identified themes. We discussed further, and agreed on the selected themes, thereby narrowing the overall research question. In a third step, we reread all interviews with a particular focus on the themes we had chosen (Tables 1, 2, 3). Next, themes (e.g. fear of SB, attitude towards SB, willingness to participate in clinical trials with SB devices) were refined into fully elaborated themes that are presented in the results section 14.

## **Results**

Most of the included patients tended to have a positive attitude to research and technology in general (a). The term "synthetic biology" was either unknown or only vaguely known or feared (b). After a concrete example of possible future treatment options, patients started to see SB in a more positive way (c).

### a) Patients' views on research and technology in general

Most interviewed patients held positive attitudes about research and technology in general. For example:

**Ulla:** Yes I think it [research] is good that there is maybe something, where you have sort of a box with insulin, which automatically injects [insulin] or such things. (...) I found it [research] good.

**Frieda:** We are not in the Middle Ages. No I find it [technology] good, (...) because mankind has always new diseases and new problems and technology causes these [problems] on one side, on the other it [technology] also helps. I have great opportunities, because if I cleaned [my] blood without technology (...), I would not be living anymore.

Otto: Yes, it [research] must be. Otherwise you cannot try new drugs. Without research there is no curing.

In these patients' opinion, research needs to be done in order to improve health conditions and to develop new medical therapies which could facilitate patients' everyday life. Most acknowledged that they themselves would need new drugs or therapies. Patients knew that a chronic disease often leads to a worsening of their health condition. Therefore, patients are aware of the fact that they would more likely need, and profit from these developments.

Nevertheless, some patients had a more nuanced picture of research and technology. According to them, research is only permissible if it is likely to lead to an improvement.

**Hanni:** If there is a good improvement for this disease, why not? But if there is no real improvement, you start to question the sense of the whole.

Few patients seemed to realise that it usually takes a long time until drugs are developed, approved and finally used to improve health or therapy. Only a few interviewees mentioned that they would not profit immediately from new findings.

**Erik:** In general, yes that [research] is very good. I am also for research (...). If you just think, how long it takes (e.g.) with cancer [research]; how long they have done research. In principle, they did not bring it [cancer research] to the end yet. That is tremendous the research nowadays.

## b) Patients' knowledge of and opinion on SB

Most patients expressed worries about and criticism of SB. In several interviews, it appeared that patients had no or only vague knowledge of SB or mixed it with other research branches.

**Frieda:** For me that [SB] is a closed book. I can imagine something. (But) at the end it [the human] will be a robot. I am not for these kinds of things.

**Erik:** Is there agriculture within? It is tremendous the research nowadays.

Time and again, the interviewees misunderstood SB as genetic engineering. The association seemed to generate a negative picture among patients (e.g. "to play God").

**Noah:** Well, if you can cure any disease with genetic engineering, I am for it. But if you manipulate the genes that – how can I say – you only have blonde people [I would say] rather not. If somehow a superhuman gets created [the interviewee is against genetic engineering]. For the improvement of diseases, sure [positive adjusted towards genetic engineering]. Make that they [diseases] disappear. As long as you do not want to play God I find everything good."

**Petra:** With genetic engineering we should slowdown, in my opinion. Otherwise we will have like in a science fiction movie all the same babies in the world: artificially

generated. That [artificially generated] is [something] we do not want. We have still limits and if you cannot continue, we have reached our limit. Rather say: okay I am sorry, but we cannot help you anymore, we have reached our limit.

The idea of genetic modification seemed to evoke the fear of creating something uncontrollable, similar to Frankenstein's monster. Hence, some of the patients expressed a sceptical opinion about what is going to happen, if researchers want to create a so-called "perfect human" with the help of SB.

**Anna:** I see disadvantages in breeding a perfect human some day or that something is bred, which cannot be controlled anymore (...).

**Noah:** Theoretically it [SB] is good, but since they [scientists] are not (...) far with it [SB], it is arguable that they [scientists] already want to start with it [SB] in humans. One could make a gene (another gene), which produces something weird. You can see it with plants. They are very good (...), but they [plants] produce something that they should not.

### c) Patients' attitudes towards possible future treatments based on SB

Using a concrete hypothetical example we explained to patients how a possible medical treatment based on SB could look like (see Methods). After this, most of the patients started to see SB in a more positive way.

**Ines:** (...) it [example of a possible SB treatment] would be good, if it were possible. Hopefully, they [SB modified cells] will get along with the old cells in the body (...); [Thank to SB] you would have to pay less attention to your diet and that would be a progress.

**Yves:** Yes, instead of measuring [blood sugar] and injecting [insulin] (if they were already that far?) (...) the body would get back its actual function, which is not working well at the moment.

The concrete example we used seemed to help most of the interviewees get a clearer picture of what SB actually is. With this increased knowledge, most of the patients were able to better distinguish between SB and other research branches (e.g. genetic engineering or other types of biotechnology) that they had previously mixed up. They also expressed better understanding of their own possible benefits regarding new therapies or improvement of their own health. For example Erik, a patient who did not know in the beginning what SB exactly was, felt very positive about SB after he received a concrete example of a possible future treatment based on SB.

**Erik:** Yes, is that it? If one can do that [example of possible SB treatment], I am in favor, sure. (...) Because human beings will be better off afterwards.

### **Discussion**

To our knowledge this study is unique in that we are the first to conduct an empirical study with stable patients in order to understand how their opinion about SB might affect their willingness to participate in FIH trials. (A Pubmed search for the combined terms "synthetic biology" AND "patient autonomy" or "first-in-human trial" or "stable patients" obtained zero results.) Patients' opinions about research and technology in general, the subject of SB and treatment options in this field could be useful in the future, when the FIH trials pass into more advanced clinical phases (II and III)<sup>2,3</sup>. While studies exist on the theoretical background of the general public's fear of and opinion about SB<sup>9</sup> and media coverage on SB<sup>15,16</sup>, to date none has been carried out on an empirical level with stable patients. Therefore, a further novelty of our study is the fact that we explore the attitudes of those who could actually be directly involved in this type of research, namely stable patients for clinical trials in the field of SB.

### Stable Patients: Possible target group for clinical trials with SB

These first findings are of substantial importance, because previously only near-death cancer patients have been targeted for this type of study<sup>13</sup>. Given the absence of therapeutic alternatives, it is debateable whether these patients can autonomously choose to participate in clinical trials<sup>12</sup>. In contrast, stable patients suffering from a chronic illness such as diabetes are not near death. However, many of them report a decreased quality of life<sup>17</sup>. Even if they have not yet presented symptoms of cardiovascular disease, they have an overall decreased life expectancy compared to non-diabetic populations<sup>17</sup>. Similar observations have been made for gout patients<sup>18</sup>. Stable patients could profit from SB treatment, which is currently under development. For example, Kemmer et al. constructed a synthetic mammalian gene circuit to regulate uric acid homeostasis *in vivo*<sup>19</sup>.

### Enhancing patients' autonomy by providing sufficient information

Therefore, it is important to obtain a first impression of stable patients' opinion about SB in order to better understand why they would participate or refuse to participate in FIH trials. By improving, completing and correcting the patients' knowledge about SB researchers will thus be able to act at an early stage to ensure patients' autonomy while also facilitating FIH trials. According to Ong et al., patients who received more information from the physician were more satisfied overall<sup>20</sup>. From this finding one might conclude that providing information and thereby influencing patients' perspective is nothing new. However, there is a distinction between providing information on something which is relatively common for patients (e.g. cancer and cancer treatment) and providing information on something new and unknown (e.g. SB) where patients are not familiar with, or suffer from biases regarding, the presented subject as reported in the results. To a large part, participants' fears of and negative attitudes towards participation in clinical trials involving SB were due to misconceptions of what SB entails (e.g. creation of a "perfect human" or "something that cannot be controlled anymore").

### **Changing Attitudes towards SB**

Even though most of the patients did not know exactly what SB was in the beginning, their attitudes towards SB changed to a more positive direction after they were provided with accurate information. It is important to note that people who are scared of SB can nonetheless envisage having a medical SB intervention. Therefore, our results also indicate a change in patients' opinion after they received information regarding medical treatments in SB. This is an important finding which is likely to be generalizable to patients in other countries, where sceptical attitudes about biotechnology and genetically modified organisms have also been identified<sup>21</sup>. We have good reason to believe that the changes of Swiss patients' opinions about SB would also occur in patients from other countries. The interviewees' answers

showed that negative attitudes were most often related to lack of accurate knowledge and confusion over various technologies. The media often convey a negative view on these new technologies, by invoking religious and environmental arguments<sup>9,15</sup>. It is important to note that most of the participants in our study expressed rather positive attitudes to research and technology in general. They stated, for example, that research must be done in order to improve their own health and everyday lives.

### Reasons for patients' ambivalent attitudes towards SB

Participants' ambivalent attitude – both negative and positive – towards SB reflects the theoretical background, which has already been described in the research literature regarding fears related to SB but also the possibilities of medical improvements thanks to SB<sup>1,5,6,9</sup>. In addition, SB is often compared with biotechnology, since it deals with similar issues (e.g. the construction of life). Moreover, some scientists see SB as the next step of biotechnology<sup>15,22</sup> and that comparison also leads to a rather negative picture of SB. There is a general tendency in society to distrust new technologies, especially biotechnology<sup>5,15</sup>. Schmidt et al. argue that a new debate about genetically modified organisms (GMOs) should be avoided. Since SB is often associated with GMOs in public opinion, overstating both the risks and benefits of SB, even outside any GMO context, might generate a new debate. Moreover, the authors outline that transnational companies should not dominate the conversation. In previous debates about GMOs the transnational companies did not consider social and moral concerns expressed by a wider community (e.g. consumers). They conclude that with promotion of an open dialog, mistakes from the past could be avoided<sup>23</sup>. These above discussed findings give our study additional importance, since we have included patients in the discussion about SB.

### Public debates about SB

Nevertheless, it seems that SB is still in a phase where the public has particularly poor and sometimes mistaken knowledge about the topic 15. There seems to be considerable uncertainty among the public regarding what to think about SB. The comparison of SB with biotechnology (which appears to be associated with negative connotations) could lead to a continuation of old debates<sup>5,15</sup>. Another possibility is that the public pays less attention to SB, because it is not seen as something distinct or new. The media, in spite of generally mentioning SB, do not in general discuss the very concrete healing potential of SB (technology) in patient care in an elaborate way<sup>5</sup>. Therefore, it is important to encourage more specific discussions about SB to avoid possible spill-over from the abundantly negative media coverage of biotechnology<sup>5,15,22</sup>. These specific discussions could be achieved with an accurate set of information about SB. As already mentioned, after the interviewer informed patients in more detail about SB and its clinical applications with hypothetical examples for gout or diabetes treatment, most patients developed more positive opinions about SB. This observation clearly shows that accurate information can correct misconceptions and positively influence patients' opinion. Such information is a prerequisite for respecting and encouraging patients' autonomous decision-making<sup>24,25</sup> whether to participate in human subject research involving medication or devices involving SB.

In Switzerland, Schmid-Petri et al. showed that the public has little or no knowledge about SB. It is a complex field, and extensive explanation is needed in order to clarify its definition and main aims. As already noted, discussions in newspapers tend to be superficial and reports about SB are often neutral or even positive. Reports about potential risks and dangers have so far been rare. In contrast to the French-speaking part of Switzerland, in which possible regulations for SB have already been discussed in newspapers, SB is not reviewed in the German-speaking newspapers<sup>16</sup>. It can be said that SB is misrepresented, because of its

heterogeneous definition and association with biotechnology<sup>1,5</sup>. This makes it to explain to a broader public<sup>16</sup>.

### Limitations

Our study has some limitations. We included patients only from the German-speaking part of Switzerland and therefore the question arises as to whether our results are generalizable. Some regions in Europe are known to be relatively critical towards biotechnologies and genetically modified organisms<sup>21</sup>. In 2015 a popular petition for pre-implantation diagnostics was accepted by a great majority in the French-speaking part (e.g. Geneva; 82.2% votes of "yes" vs. 17.8% of "no"), whereas the acceptance was clearly lower in the German-speaking parts of Switzerland (e.g. Lucerne; 54.7% vs. 45.3%)<sup>26</sup>. These facts could partly explain the negative views of our Swiss German patients about SB as well as their diffuse fear of SB. Moreover, cultural differences between the German- and French-speaking parts of Switzerland have been discussed in other studies and are related among others to the influence of German-speaking technology opponents<sup>27,28</sup>.

Furthermore, we cannot exclude the possibility of a social desirability bias. Social desirability means that people tend to behave in a way they believe to be culturally acceptable<sup>29</sup>. Overall we have reasons to believe that the results of our study are relevant and representative for important trends in Europe. Indeed, it has been shown that fears of cutting edge biotechnologies (e.g. nanotechnology) exist throughout all countries<sup>30,31</sup>.

Our study has also the typical limitations of qualitative studies, i.e. that people who already have a strong interest in the particular field tend to participate<sup>32</sup>. The answers between patients varied and were sometimes even conflicting with one other. This could indicate that we were able to capture a sufficiently broad and heterogeneous sample.

However, our study also points to the risk that patients can be influenced by unrealistic hope or incomplete information where potential risks are underestimated. This could be due to inaccurate presentation of the information, but also due to a misconception regarding future treatment options (e.g. FIH trials)<sup>12</sup>. According to Appelbaum et al. the "therapeutic misconception" means that patients do not understand the setting of equipoise that defines research as such. Especially, if research takes place in a clinical setting such as hospitals, patients tend to deny the possibility of major disadvantages and are more likely to mistakenly assume a benefit, if they enroll in a clinical research study<sup>33</sup>. Clearly, a lack of information and accurate understanding undermines patients' decisional autonomy<sup>11,24,25</sup>. As Cialdini has shown, under uncertain circumstances, people tend to trust authorities (religious, political, media, etc.) even more than usual .These powerful figures greatly influence people's opinion and decision-making<sup>34</sup>. Therefore, a detailed knowledge of SB is crucial to guarantee patients' sufficient autonomy to make informed decisions (e.g. for possible future FIH trials)<sup>11,12</sup>. Research ethics committees have to examine very carefully the content of consent forms to make sure that information is detailed and accurate. Patients need to be able to appreciate the advantages as well as the disadvantages of clinical trials<sup>12</sup>.

### Conclusion

Our study contributes an important first insight into what stable patients think of SB and what kind of fears arise (e.g. "playing God") when SB is mentioned, and the effect on these fears once patients have received more detailed information. We have showed that some patients - when hearing about SB - might associate SB to those transhumanist dangerous experiments with humans ("playing God"). Therefore it is important to mention to the patient that the clinical use of SB is not related to these Promethean dreams. The patient's informed consent has little ethical value if it is based on loose associations and emotions. Similar recommendations were made in the field of nanotechnology. Satalkar et al. noticed too that

these terms trigger exaggerated fears in patients, but that these fears should be discussed openly with the physician as part of an adequate information process for research participants. They conclude that the word "nanotechnology" should be mentioned in the informed consent sheet in order to gain patients' trust but also to provide them a transparent discussion on clinical trials in this field<sup>25</sup>. Bromwich et al. discuss that informed consent should be adapted to the risks which the participants are taking. The authors argue that the concerns about the validity of informed consents increase with the risk of research in which the patients are involved<sup>25</sup>. Fagerlin et al. show that using plain language or pictographs can enhance patients' understanding the risk-benefit ratio<sup>24</sup>. The above discussed literature and our results reveal that providing accurate information is crucial for enhancing patients' autonomy. Researchers must present facts in a fair and balanced way. Therefore, detailed information about SB and its clinical applications are important in terms of future FIH trials<sup>11,12</sup>.

### Acknowledgments

The authors would like to thank Anna Genske for developing and adapting the interview guide. Furthermore, we would like to thank Marianne Weber for transcribing and translating the interviews with the 36 gout and diabetes patients.

## **Funding**

The research was funded by the Swiss National Science Foundation (SNF) [grant no PDFMP3 137194/1].

## **Appendix**

Hypothetical Example for Gout Patients:

"I will guide you into Mr. Müller's world in the year 2025. Mr. Müller suffers, as you, from gout, but he is not taking any drugs. He needs to go rarely to his GP to get his uric acid level controlled. Mr. Müller has a capsule implanted under his skin. This capsule contains human cells. These cells are from Mr. Müller himself. They were removed and modified in the laboratory.

These modified cells from Mr. Müller are able to produce a certain protein that removes the surplus uric acid in the blood. Additionally, a sensor was integrated that constantly measures the uric acid levels in the blood. If it [uric acid level] increases too strongly, the protein production is stimulated. She [cell] produces more of the uric acid degrading protein. Mr. Müller notices nothing of this. Everything happens automatically and alone.

Hypothetical Example for Diabetes Patients:

"I will guide you into Mr. Müller's world in the year 2025. Mr. Müller suffers, as you, from diabetes. Mr. Müller has a capsule implanted under his skin. This capsule contains human cells. These cells are from Mr. Müller himself. They were removed and modified in the laboratory.

These modified cells from Mr. Müller are able to produce a certain substance that stimulates the insulin production in the body and decreases the blood sugar concentration. Mr. Müller is wearing a plaster with incorporated LED-lamps on his skin. After food intake, Mr. Müller switches the LED-lamps on via touch of a button. The light ensures that the substance gets produced in the cells and distributed in the blood.

## References

- 1. Balmer A, Martin, P., . Syntetic biology: Social and ethical challenges. Institute of Science and Society, University of Nottingham, Biotechnology and Biological Sciences Research Council (BBSRC). 2008; http://www.synbiosafe.eu/uploads/pdf/synthetic\_biology\_social\_ethical\_challenges.p df Accessed 03 April 2015.
- 2. Xie M, Fussenegger M. Mammalian designer cells: Engineering principles and biomedical applications. *Biotechnology Journal*. 2015;10(7):10051018-n/a.
- 3. Ausländer S, Fussenegger M. Toehold gene switches make big footprints. *Nature*. 12/17/online 2014;516:333.
- 4. Ruder WC, Lu T, Collins JJ. Synthetic Biology Moving into the Clinic. *Science*. 2011;333(6047):1248-1252.
- 5. Torgersen H. Synthetic biology in society: learning from past experience? *Systems and Synthetic Biology*. October 10 2009;3(1):9.
- 6. Murren JR, DiStasio SA, Lorico A, et al. Phase I and pharmacokinetic study of novobiocin in combination with VP-16 in patients with refractory malignancies.

  \*Cancer J. 2000 Jul-Aug 2000;6(4):256-265.
- 7. Ehrbar M, Schoenmakers R, Christen EH, Fussenegger M, Weber W. Drug-sensing hydrogels for the inducible release of biopharmaceuticals. *Nature Materials*. 08/10/online 2008;7:800.
- 8. Spaemann R. Grenzen: zur ethischen Dimension des Handelns. Klett-Cotta; 2001.
- 9. Link H-J. Playing God and the Intrinsic Value of Life: Moral Problems for Synthetic Biology? *Science and Engineering Ethics*. June 01 2013;19(2):435-448.
- 10. Wilhelm K, Beck L, Mikat P. Lexikon der Bioethik 1. *Bd. I, Gütersloh.* 1998.

- 11. Beauchamp TL, Childress JF. *Principles of biomedical ethics*. 7th ed. New York: Oxford University Press, USA; 2009.
- 12. Dresser R. First-in-Human Trial Participants: Not a Vulnerable Population, but Vulnerable Nonetheless. *The Journal of Law, Medicine & Ethics*. 2009;37(1):38-50.
- 13. Shaw D, Elger B. Putting patients on research ethics committees. *Journal of the Royal Society of Medicine*. 2014;107(8):304-307.
- 14. Braun V, Clarke V. Using thematic analysis in psychology. *Qualitative Research in Psychology*. 2006/01/01 2006;3(2):77-101.
- 15. Kronberger N, Holtz P, Kerbe W, Strasser E, Wagner W. Communicating Synthetic Biology: from the lab via the media to the broader public. *Systems and Synthetic Biology*. October 10 2009;3(1):19.
- 16. Schmid-Petri H, Knocks, S., Sager, P., Adam, S., Synthetische Biologie in der Gesellschaft: Eine neue Technologie in der öffentlichen Diskussion. 2014; https://www.taswiss.ch/?redirect=getfile.php&cmd[getfile][uid]=2778 Accessed 04 May 2015.
- 17. Franco OH, Steyerberg EW, Hu FB, Mackenbach J, Nusselder W. Associations of diabetes mellitus with total life expectancy and life expectancy with and without cardiovascular disease. *Archives of Internal Medicine*. 2007;167(11):1145-1151.
- 18. Chen J-H, Lan J-L, Cheng C-F, et al. Effect of Urate-lowering Therapy on the Risk of Cardiovascular Disease and All-cause Mortality in Patients with Gout: A Casematched Cohort Study. *The Journal of Rheumatology*. 2015;42(9):1694-1701.
- 19. Kemmer C, Gitzinger M, Daoud-El Baba M, Djonov V, Stelling J, Fussenegger M. Self-sufficient control of urate homeostasis in mice by a synthetic circuit. *Nature Biotechnology*. 03/28/online 2010;28:355.

- 20. Ong LML, Visser MRM, Lammes FB, de Haes JCJM. Doctor-Patient communication and cancer patients' quality of life and satisfaction. *Patient Education and Counseling*. 2000/09/01/2000;41(2):145-156.
- 21. Ammann D. Schweizerische Arbeitsgruppe Gentechnologie SAG, Fact Sheet: Moratorien und Verbote weltweit. 2003; http://gentechfrei.ch/images/stories/pdfs/papiere/fs\_moratorien.pdf Accessed 06 May 2015.
- 22. Newson AJ. Current Ethical Issues in Synthetic Biology: Where Should We Go from Here? *Accountability in Research*. 2011/05/01 2011;18(3):181-193.
- 23. Schmidt M, Torgersen H, Ganguli-Mitra A, Kelle A, Deplazes A, Biller-Andorno N. SYNBIOSAFE e-conference: online community discussion on the societal aspects of synthetic biology. *Systems and Synthetic Biology*. June 01 2008;2(1):7-17.
- 24. Fagerlin A, Zikmund-Fisher BJ, Ubel PA. Helping Patients Decide: Ten Steps to Better Risk Communication. *JNCI: Journal of the National Cancer Institute*. 2011;103(19):1436-1443.
- 25. Bromwich D, Rid A. Can informed consent to research be adapted to risk? *Journal of Medical Ethics*. 2015;41(7):521-528.
- Pruss S. Klares Ja für PID: Referendum angekündigt. 14 June 2015. *Tagesanzeiger* 2015.
- 27. Elger BS. Attitudes of Future Lawyers and Psychologists to the Use of Genetic Testing for Criminal Behavior. *Cambridge Quarterly of Healthcare Ethics*. 2005;14(3):329-345.
- 28. Elger BS, Harding TW. Teaching Changes Attitudes to Genetic Testing for Aggressive Behaviour. *Medical Law International*. 2004;6(4):277-295.

- 29. Crowne DP, Marlowe D. A new scale of social desirability independent of psychopathology. *Journal of Consulting Psychology*. 1960;24(4):349-354.
- 30. Brumfiel G. A little knowledge. *Nature*. 07/17/online 2003;424:246.
- 31. Beachy RN. Facing Fear of Biotechnology. Science. 1999;285(5426):335-335.
- 32. Gott M, Ingleton C, Bennett MI, Gardiner C. Transitions to palliative care in acute hospitals in England: qualitative study. *BMJ*. 2011;342.
- 33. Appelbaum PS, Roth LH, Lidz CW, Benson P, Winslade W. False Hopes and Best Data: Consent to Research and the Therapeutic Misconception. *The Hastings Center Report*. 1987;17(2):20-24.
- 34. Cialdini RB. Influence: The psychology of persuasion. 2009; http://ir.nmu.org.ua/bitstream/handle/123456789/116954/06b89c8343b30b05a99d572 3277c39f8.pdf?sequence=1 Accessed 06 July 2015.
- 35. Satalkar P, Elger BS, Shaw D. Naming it 'nano': Expert views on 'nano' terminology in informed consent forms of first-in-human nanomedicine trials. *Nanomedicine*. 2016;11(8):933-940.

Chapter 3

**Enhancing Patients' Autonomy by Involving Them in Research Ethics** 

**Committees** 

Reprinted with the permission of © 2017 Oxford University Press

Citation: Rakic M, Dittrich T, Elger BS, Shaw D. Enhancing patients' autonomy by

involving them in research ethics committees. International Journal for Quality in Health

Care: Journal of the International Society for Quality in Health Care. Nov 2017;29(7):896-

900.

(Impact Factor: 2.342)

**Abstract** 

Objective Although clinical trial participants are the most affected by research ethics

committee's decisions, they are not formally represented on Swiss committees. We aimed to

find out what patients think about the idea of being members of such committees.

**Design** Latent thematic analysis was used to analyse the interviews.

**Setting** Patients were recruited in a Swiss university hospital.

**Participants** The study involved 26 patients suffering from diabetes or gout.

**Interventions** We conducted semi-structured interviews.

Main Outcome Measures We explored what patients think of being established members of

research ethics committees.

**Results** We identified three different attitudes among our participants regarding participation

in research ethics committees: (i) positive attitude regarding the idea of being members of

such committees (ii) ambivalent attitude and (iii) negative attitude. Patients belonging to the

first group (i) often mentioned that they wanted their health condition to be more visible.

Patients from the second group (ii) mentioned positive as well as negative aspects. Patients

from the third group (iii) said that patients in general did not have enough background

knowledge to be able to gain an overview of a whole clinical trial.

Conclusions Our study adds important knowledge about the idea of patients becoming

research ethics committee members by exploring their perceptions of being members. Stable

patients tended to be interested in the idea of participation and some specific

recommendations could be derived (patients could have an advisory instead of a decision-

making role on committees). However, further studies with more patients and further

quantitative research are needed.

**Key words:** research ethics committee, autonomy, stable patients

55

## Introduction

According to the Declaration of Helsinki, every trial involving human beings must be approved by a research ethics committee (REC) in advance. Each committee member must be independent of the researcher and sponsor and must operate free of any other inappropriate influence<sup>1</sup>. RECs are constituted by members of different specialities (e.g. physicians, statisticians, ethicists). In the UK, RECs need to include so-called "lay members" who are independent of health services. Half of them are required to never have worked in the healthcare sector. Some RECs also include patient representatives<sup>2</sup>. Swiss RECs are similar as they include experts from different scientific disciplines, but they are not required to include lay members<sup>2,3</sup>. Nevertheless, each state (canton) of the federalist Swiss Confederation can decide whether its REC needs to include lay persons, defined as any person who is not a physician. However, the lay person is not necessarily a representative of patients<sup>4</sup>. While three cantonal RECs (Zurich, St. Gallen, Vaud) employ patient representatives, only 1.6% of Swiss REC members are patient representatives<sup>4</sup>. In contrast, 41.5% are physicians<sup>4</sup>. This means that although clinical trial (CT) participants are the group most affected by the REC's decisions, they are not formally represented on Swiss RECs<sup>3</sup>. Including patient representatives on RECs, however, is not necessarily enough to support patient interests. The question is whether patient representatives are able to speak for patients who suffer from other diseases. Therefore, it might be helpful to include patients rather than patient representatives or lay members in RECs and decision-making regarding CTs to ensure that patients' voices are heard<sup>2</sup>.

A possible disadvantage of having patients as REC members could be the whole process of recruiting appropriate patients (e.g. stable patients), which could be very costly and time consuming<sup>2</sup>. CTs involving novel technologies (e.g. synthetic biology) and stable patients (i.e. not terminally ill), are more likely to be rejected because RECs consider it highly risky to

enrol stable patients in (e.g.) first-in-human (FIH) trials. In these cases, the potential damage to health outweighs the potential benefit for volunteering participants<sup>2</sup>. Terminally ill cancer patients are a common target group for CTs, given the absence of alternative treatment options. However, decisions of patients who have a life-threatening illness may be strongly affected by the lack of treatment alternatives<sup>5</sup>. Indeed, patients with a high degree of psychological strain might be more likely to accept possibly high risks and be victims of the therapeutic misconception (i.e. lacking awareness of the difference between clinical research and therapeutic treatment)<sup>6</sup>. Moreover, the informed consent process itself can be exhausting for severely ill patients and be considered as too harmful<sup>7</sup>. Enrolling stable patients in FIH trials (e.g. diabetes patients) could be a way to avoid these ethical issues. Stable patients have more treatment options than near-death cancer patients. Hence, they are more likely to make free choices<sup>5</sup> and less likely to fall victim to the therapeutic misconception<sup>6</sup>. Additionally, they might benefit from the medications developed through CTs since despite limited quality of life, life expectancy is only slightly decreased<sup>8,9</sup>. Therefore, RECs' tendency to reject highrisk studies involving stable patients could be criticized as unjustified paternalism<sup>10,11</sup>. It has been argued that stable patients are capable of giving reflective "independent" answers<sup>5</sup> and should be allowed to decide whether to take trial related risks. McKinstry concludes that paternalism, especially with respect to competent patients, is rarely justified<sup>12</sup>. Respect for patient autonomy is one of the four principles of modern biomedical ethics<sup>13</sup>. Two prerequisites for a person to be autonomous are identified: liberty and agency. This means that a person can decide independently, free from others' influence and has the capacity to act in an intentional manner<sup>13</sup>. Autonomy has acquired increased importance in medicine<sup>7</sup>. Failing to consider patients' willingness to accept risks, i.e. the paternalistic attitude of RECs<sup>2,11</sup>, seems to create an anachronistic situation regarding autonomy. This situation could lead to the patient's right to make autonomous decisions regarding participation in clinical

research being restricted. Therefore, shifting from paternalistic decision-making to a patient-centred and shared decision-making by putting patients on RECs increases patients' ability to engage in autonomous decision-making<sup>14</sup>. Thus, patients should have the opportunity to participate in decision-making regarding CTs.

## Aim of the Study

Given the identified advantages and disadvantages of a system where RECs include patients as members<sup>2</sup>, we aimed to find out what stable patients think about being members of RECs. Therefore, we explored this issue using qualitative interviews from a broader study on FIH trials with stable patients.

### Methods

We employed an empirical research design with the overall goal of exploring stable patients' attitudes towards participation in RECs as established members. The data were gathered as a part of a larger interview study where patients were asked about their attitudes towards synthetic biology and related CTs<sup>15</sup>. A pilot study was carried out in order to ensure patients' understanding. We used the 32-item checklist from consolidated criteria for reporting qualitative research (COREQ)<sup>16</sup>. We obtained ethical approval from the local REC.

### **Sampling and Data Collection**

The interview study involved a purposive sample of patients suffering from diabetes or gout. Inclusion criteria for this study were that patients were (i) > 18 years, (ii) stable, and (iii) suffered either from diabetes or gout. The recruitment was carried out in the hospital during a time period of six months in 2014/2015 while patients were there for a routine check-up. Our recruiting physicians approached suitable patients and asked if they would be interested in participating in an interview study. Subsequently, a research assistant (trained in qualitative research) went to the hospital and asked patients if they agreed to participate. In case of agreement, patients received a participant information sheet, and an informed consent form

and were given time to consider whether to consent. Afterwards, the research assistant approached the patients via phone or email and made an appointment. The interview took place at the patient's home, our research institute, or in the hospital. During the interview only the interviewer and the interviewee were present in the room. All interviews were conducted by the same research assistant. Written informed consent was obtained from all participants. On average, the interviews lasted for 44 min (range: 16 to 102 min). All interviews were conducted in Swiss German, transcribed verbatim and fully anonymized.

#### **Interview Guide**

After a literature search for research on cutting edge biotechnology, we developed an interview guide. Based on the previous literature research, we selected topics which should be included in our interview guide (e.g. examples of CTs, patients' current health status). The semi-structured interview guide provided a short explanation of our study and of RECs, followed by two questions regarding patients' opinion on RECs (Table 1). These two questions were specifically analysed in this paper.

### **Table 1: Questions about research ethics committees**

Before a new type of treatment (e.g. gout or diabetes capsules) can be tested on patients within clinical trials, a committee of experts decides whether this trial may to take place at all. This committee assesses the possible risks and benefits related to this trial, and weigh them up against each other. In Switzerland there are many different ideas about who should be represented in such a committee.

- 1. What do you think about this? Who should be represented in such a committee and why?
- 2. How do you see the participation of patients in such committees?

## **Analysis**

Of the 36 patients, 26 answered the question about the topic of patient representation in RECs. Ten patients responded that due to not knowing the concept of RECs they could not give an appropriate answer. Here, we present and analyze the responses of the remaining 26 patients. We used latent thematic analysis as it is independent of a particular theory or epistemology. Therefore, it can be applied across different approaches<sup>17</sup>. The project did not intent to develop a theory or understand the experiences of the patients, but to study their opinions on participation on RECs. First, all authors read the interview-transcripts in order to inductively obtain preliminary themes that are not a priori defined based on a theory<sup>17</sup>. The authors discussed the preliminary themes. Afterwards, we agreed on themes and coded the transcripts according to the agreed themes. Member check was done by the co-authors. During the analysis for the questions analysed for this paper, data saturation was reached according to Guest et al.<sup>18</sup>. The final themes are presented in Results.

## **Study Population**

Eighty-eight percent (n=23) of the patients were male; age ranged from 26 to 91 years. Other demographic information is presented in Table 2.

Table 2: Descriptive statistics of our study population (n = 26)

•	•
Gender (male)	88.5%
<b>Age</b> (M*; <i>SD</i> *) (1 missing)	64.9 (16.599)
Education (1 missing)	
University	19.2%
Apprenticeship	76.9%
Household (1 missing)	
• ≤ two person	88.5%
• > three person	7.7%
Disease	
Diabetes Mellitus I	23.1%
Diabetes Mellitus II	38.5%
• Gout	38.5%

<sup>\*</sup>M = mean; SD = standard deviation

### **Results**

We identified three different attitudes regarding participation on RECs: (i) exclusively mentioning positive aspects (ii) ambivalence about the idea, and (iii) exclusively mentioning negative aspects.

## Positive aspects of being members of RECs

The majority were very positive about patient participation in RECs. Participants often mentioned their own benefit as a reason to participate as members. More precisely, patients saw a chance of a long-term benefit on their own health condition from participating in RECs and perhaps allowing more CTs. Another frequent reason was that patients wanted their situation to be visible. Patients also said that they have experience with the disease and therefore they can tell what it is like to live with it. For this reason, they should be able to integrate their opinions on whether a CT should be performed within an REC. Another reason for patient participation in RECs was that they possibly add different perspectives on a certain aspect of the disease (Table 3).

### Table 3: Positive aspects of being members of RECs

**Patient 8:** This would be good. [Good] in the sense that they [patients] talk from experience, they know what happens in the body [e.g.] of a diabetes patient. In my opinion there should be both types of diabetes patients in there [committees]. This would mirror the reality very well (...), because they [diabetes patients] know what they are talking about and you can rely on their opinion. This would be the main goal that these kinds of people [patients] are represented in such committees (...), people who know the problem (diabetes diagnosis 10 years ago).

**Patient 1:** To be able to see the patients' situation. He/she can, only he/she can, say how hard the disability is, caused by the disease. He/she can estimate the agony caused by the disease. A physician is not able nor is a psychologist, right. That is something only the patient can (...) and I think that [patients] should be heard and incorporated [in RECS] (gout diagnosis one year ago).

**Patient 7:** Because they [patients] are interested that science gets on (...), because there are a lot of things that do not work 100% yet. This could be a reason that they [patients] are interested in being part of such committees and maybe they can positively influence the whole thing (gout diagnosis less than one year ago).

**Patient 16:** Because he/she [patient] can maybe add some things, which might get forgotten by more educated people (diabetes diagnosis 3 years ago).

### Ambivalent aspects of being members of RECs

A smaller second group of patients was ambivalent. Participants mentioned both positive (e.g. to see what is happening; patients experience) and negative aspects (e.g. patients may not have enough knowledge; they only see their own benefit) of being on RECs. Some patients said that it does not necessarily have to be patients but independent people who are on the committee (Table 4).

## Table 4: Ambivalent aspects of being members of RECs

**Patient 25:** There are pros and cons (...) which speak for and against it, right. Patients have some experiences as well. (...) if there are more [patients], there is an exchange of ideas and they could talk together and accordingly influence something in a positive or negative way (diabetes diagnosis 15 years ago).

**Patient 36:** I think that is delicate, because the "ego-thinking" comes up again, yes that could bring something or it [drug, treatment] could help a little bit. I do not think so (gout diagnosis 12 years ago).

**Patient 23:** For this [patient participation in RECs] that they can see what is being done. Against it [patient participation in RECs] that there will be maybe a pushing [due to own profit] in order to get it [drug] developed.

### Negative aspects of being members of RECs

Finally, a few patients did not see any positive aspects of making patients members of RECs. The main reason cited regarding why patients should not be part of RECs was that they do not have enough expertise to be able to gain an overview of the whole study. In addition, participants mentioned that they did not understand the purpose of having ordinary people on a REC (Table 5).

### Table 5: Negative aspects of being members of RECs

**Patient 13:** He/She [patient] does not have the overview of the whole thing [study], of what was and what might come (...). The patient cannot determine this (diabetes diagnosis 40 years ago).

**Patient 24:** They [patients] cannot decide, because they [patients] have no knowledge. They [patients] only can say if this [treatment, drug] would be something for me, but if it [treatment, drug] is really good, that is something they cannot decide (...) (diabetes diagnosis 41 years ago).

**Patient 28:** Because I think that we as ordinary mortals do not really have an idea what really goes on [Diabetes diagnosis 12 years ago].

Pat 29: (...) I need to leave this [decision] to the experts (Gout diagnosis one year ago).

In summary, our analysis showed that most participants were positive about the idea of being members of RECs. Interviewees illustrated their experience and their comprehension of the disease, but also their interest in propelling research. Many patients think that present RECs do not pay enough attention to their health condition, and that only patients are able to assess it.

## **Discussion**

To our knowledge, this is the first empirical study to specifically explore stable patients' perceptions of being REC members. The most commonly mentioned advantageous aspects were: (i) potential benefit for affected patients and (ii) the disease burden of the patients, which enables them to add new perspectives on certain aspects of the disease.

Surprisingly, although the interviewees were informed that participation in CTs does not lead to any personal benefit in terms of changes in therapy, they still anticipated personal benefit as one of the main incentives for the involvement of patients in RECs. This might be seen as a biasing factor affecting proponents of this idea, because the hope of long-term benefit for oneself might make one overlook the risks of participating. Considering this as an indicator for a persistent therapeutic misconception disqualifies this as an argument for involving patients in RECs.

However, this biasing factor does not seem to operate on a broader scale. This presumption is supported by the data of a large cross-sectional survey exploring community understanding of medical RECs<sup>19</sup>. According to this study the majority of the respondents were aware of the role of RECs.

The second most-mentioned motivation was disease-related knowledge, particularly regarding the accompanying psychological and physical strains, which could allow RECs to obtain additional insight into a specific medical condition. Generally, patients believed themselves to be most familiar with their specific medical issue and this specific aspect has not yet been addressed by RECs.

Interestingly, our cohort, with a relatively high average age of almost 65 years, had similar views to participants in a study investigating young participants' views on composition of RECs (mean: 16 years). Overall, participants expressed the need for a broader view provided by lay people<sup>20</sup>.

Amongst the negative aspects mentioned in our cohort were patients' lack of knowledge compared to experts, and potential difficulties in assessing the consequences of decisions made in the context of complex studies. Similarly, professional expertise was considered important among young people<sup>20</sup>.

Nevertheless, the argument that lack of medical knowledge could count against patients' inclusion in decision-making is not tenable. As stated by Edwards et al.<sup>10</sup>, it is not the task of RECs to judge the competence of patients. As in the shared decision-making model, participating patients would have to be properly provided with all relevant information (e.g. risks, potential benefits) before engaging in REC decisions.

Overall, stable patients seem to be particularly appropriate decision-makers as they suffer from a chronic disease with a significant burden 21-24. Due to the relatively slow progression and associated mortality of e.g. diabetes compared to acute diseases, stable patients are more likely to benefit from future research<sup>23,25</sup>. Therefore, REC decisions on CTs might affect them indirectly over the medium-term, and the problem of 'therapeutic misconception' could have a less prominent role<sup>6</sup>. The most important ethical consideration favouring the inclusion of stable patients is the need to respect patient autonomy. In order to meet this requirement, affected patients should have the opportunity to participate in decision-making in RECs through their representatives. It has been claimed before that qualified individuals are in the best position to decide which risks are acceptable for them<sup>10</sup>. From this point of view, the exclusion of competent patients from RECs can be seen as a form of paternalism, contradicting the concept of shared decision-making. Shared decision-making – in a clinical setting – means that neither the physician ("paternalism") nor the patient ("informed choice") on his own decide what is going to be done (e.g. what kind of treatment etc.). Instead, the decision is made together<sup>26</sup>. This requires the ability of an individual to make his or her own, independent choices and respecting patients' perspectives as being just as valuable as experts'

perspectives<sup>26,27</sup>. The impression of RECs as a paternalistic characterized decision-making body<sup>10,28</sup> could be countered by enhancing patients' autonomy through involving them in decision-making.

### Limitations

The limitations of this current study relate to its qualitative design and relatively low number of participants: the subjectivity of data analysis and the limited generalizability of the results that we obtained from German-speaking patients. Furthermore, 10 out of 36 persons could not say anything about the idea of patients as members of RECs because they were lacking respective knowledge. Since our investigation only focused on stable patients, generalizable statements about other patient groups are limited. Another possible limitation could be the overrepresentation of male patients (88.5%). However, an epidemiological study showed that 80% of gout patients were male<sup>29</sup>. Also, a review showed a male excess in diabetes mellitus I, whereas diabetes mellitus II seemed to be fairly distributed among both sexes<sup>30</sup>. Generally, it cannot be ruled out that supposedly healthy lay members suffer from a chronic disease themselves, in which case their decision-making may also be affected.

### Conclusion

From the previous considerations we conclude that stable patients appear to be suitable as REC members in the field of clinical research. This is particularly true for CTs in which novel treatment technologies will be evaluated in affected but stable patients. The reasons for this are various: First, it can be argued that respect for patients' autonomy as one of the four fundamental principles of medical ethics requires the involvement of patients instead of healthy third parties (e.g. physicians, ethicists, lay members). Furthermore, the interviews revealed a tendency towards the idea of involvement of stable patients. The primary stated motivation for this was the improvement of therapeutic options in the future. This seems to be supported by the fact that chronic diseases such as diabetes commonly have a slowly progressive clinical course over several years and do not pose an acute threat to life compared

to, for example, advanced stages of cancer. In addition to this idea of being able to influence new therapeutic options in the future, there are presumably further reasons why it could be appealing for stable patients to be part of RECs. Taking part in decision-making regarding CTs could create a feeling of inclusion and respect for patient perspectives. In this way, early integration of persons affected could influence the general acceptance of research in a positive manner. In contrast to terminally ill patients, stable patients are more likely to be able to fulfil the demanding work of a REC member because of their better health condition. Relatively slow disease progression with possible stable phases under therapy, comparatively good life expectancy, and diverse treatment options especially in early stages may prevent stable patients from experimental treatment approaches.

The improvement in quality of life in the long-term instead of a short-term prolongation of life expectancy appears to be the primary objective for many of the stable patients. Therefore, the possible objection that stable patients could generally have a bias towards approving research on novel treatment technologies is not applicable. Patients could play an important role when it comes to the benefit-risk assessment of future therapeutic interventions. The main additional benefit of this patient group, compared to healthy REC members, is the disease-burden that these patients are aware of and the direct insight into patients' perspectives that they could provide.

For practical purposes it could generally make sense to reach out to stable patients that match the inclusion criteria of the planned study. The patients included in RECs should not take part in the study themselves in order to avoid biased opinions. Another important point is that patients would have to be properly informed about their role in the REC as affected patient representatives. By excluding patients from taking part in the trial and by providing detailed information the influence of therapeutic misconception should be reduced. A point of concern in this context could be that patients who take part in decision-making but do not participate

in the study themselves, may overlook potential risks to participants of the trial by attaching greater weight to potential long-term benefits. To counter this, patients could have an advisory instead of a decision-making role on RECs.

# **Funding**

This work was funded by the Swiss National Science Foundation [grant number 137194]

# References

- WORLD MEDICAL ASSOCIATION (WMA). WMA, Declaration of Helsinki 2008;
   http://www.wma.net/en/30publications/10policies/b3/ Accessed 07 September 2016.
- 2. Shaw D, Elger B. Putting patients on research ethics committees. *Journal of the Royal Society of Medicine*. Aug 1 2014;107(8):304-307.
- The Swiss Federal Council. Organisationsverordnung zum Humanforschungsgesetz.
   https://www.admin.ch/opc/de/classified-compilation/20121179/index.html
   Accessed 31 August 2016.
- 4. The Swiss Federal Office of Public Health. Tätigkeiten der Ethikkommissionen für die Forschung. 2014; http://kofam.ch/filemanager/downloads/gutachten\_und\_berichte/BAG\_Zusammenfass ung Ethik A4 D low.pdf Accessed 05 September 2016.
- 5. Dresser R. First-in-human trial participants: not a vulnerable population, but vulnerable nonetheless. *The Journal of law, medicine & ethics : a journal of the American Society of Law, Medicine & Ethics.* Spring 2009;37(1):38-50.
- 6. Henderson GE, Churchill LR, Davis AM, et al. Clinical Trials and Medical Care:

  Defining the Therapeutic Misconception. *PLoS Medicine*. Nov 2007;4(11).
- 7. O'Neill O. *Autonomy and Trust in Bioethics*. Cambridge: Cambridge University Press; 2002.
- 8. Chen JH, Lan JL, Cheng CF, et al. Effect of Urate-lowering Therapy on the Risk of Cardiovascular Disease and All-cause Mortality in Patients with Gout: A Casematched Cohort Study. *The Journal of rheumatology*. Sep 2015;42(9):1694-1701.
- 9. Franco OH, Steyerberg EW, Hu FB, Mackenbach J, Nusselder W. Associations of diabetes mellitus with total life expectancy and life expectancy with and without

- cardiovascular disease. *Archives of internal medicine*. Jun 11 2007;167(11):1145-1151.
- 10. Edwards SJL. Research ethics committees and paternalism. *Journal of Medical Ethics*. 2004;30(1):88-91.
- 11. Shaw D. The right to participate in high-risk research. *The Lancet*. 2014;383(9921):1009-1011.
- 12. McKinstry B. Paternalism and the doctor-patient relationship in general practice. *The British Journal of General Practice*. 1992;42(361):340-342.
- 13. Beauchamp TL, Childress, J.F. *Principles of Biomedical Ethics.* 7 ed. Oxford University Press 2009.
- 14. Charles C, Gafni A, Whelan T. Decision-making in the physician-patient encounter: revisiting the shared treatment decision-making model. *Social Science & Medicine*. 9// 1999;49(5):651-661.
- 15. Rakic M, Wienand I, Shaw D, Nast R, Elger BS. Autonomy and Fear of Synthetic Biology: How Can Patients' Autonomy Be Enhanced in the Field of Synthetic Biology? A Qualitative Study with Stable Patients. *Science and Engineering Ethics*. May 2 2016.
- 16. Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. *International Journal for Quality in Health Care*. 2007;19(6):349-357.
- 17. Braun V, Clarke V. Using thematic analysis in psychology. *Qualitative Research in Psychology*. 2006;3(2):77-101.
- 18. Guest G, Bunce A, Johnson L. How Many Interviews Are Enough? *Field Methods*. 2006/02/01 2006;18(1):59-82.

- 19. Fritschi L, Kelsall HL, Loff B, Slegers C, Zion D, Glass DC. A cross-sectional survey to investigate community understanding of medical research ethics committees. *J Med Ethics*. Jul 2015;41(7):545-548.
- 20. Audrey S, Brown L, Campbell R, Boyd A, Macleod J. Young people's views about the purpose and composition of research ethics committees: findings from the PEARL qualitative study. *BMC medical ethics*. Sep 02 2016;17(1):53.
- 21. Stolar M. Glycemic control and complications in type 2 diabetes mellitus. *The American journal of medicine*. Mar 2010;123(3 Suppl):S3-11.
- 22. Fowler MJ. Microvascular and macrovascular complications of diabetes. *Clinical diabetes*. 2008;26(2):77-82.
- 23. Morgan CL, Currie CJ, Peters JR. Relationship between diabetes and mortality: a population study using record linkage. *Diabetes care*. Aug 2000;23(8):1103-1107.
- 24. Nathan DM. Long-term complications of diabetes mellitus. *The New England journal of medicine*. Jun 10 1993;328(23):1676-1685.
- 25. Collins TC, Petersen NJ, Menke TJ, Souchek J, Foster W, Ashton CM. Short-term, intermediate-term, and long-term mortality in patients hospitalized for stroke. *Journal of clinical epidemiology*. Jan 2003;56(1):81-87.
- 26. Makoul G, Clayman ML. An integrative model of shared decision making in medical encounters. *Patient education and counseling*. Mar 2006;60(3):301-312.
- 27. Elwyn G, Frosch D, Thomson R, et al. Shared Decision Making: A Model for Clinical Practice. *Journal of General Internal Medicine*. 05/23 08/22/received 01/03/revised 04/03/accepted 2012;27(10):1361-1367.
- 28. Garrard E, Dawson A. What is the role of the research ethics committee? Paternalism, inducements, and harm in research ethics. *Journal of Medical Ethics*. July 1, 2005 2005;31(7):419-423.

- 29. Mikuls TR, Farrar JT, Bilker WB, Fernandes S, Schumacher HR, Jr., Saag KG. Gout epidemiology: results from the UK General Practice Research Database, 1990-1999.

  \*\*Annals of the rheumatic diseases. Feb 2005;64(2):267-272.
- 30. Gale EA, Gillespie KM. Diabetes and gender. *Diabetologia*. Jan 2001;44(1):3-15.

# **Chapter 4**

\_

# The Beneficence of Hope: Findings from a Qualitative Study with Gout and Diabetes Patients

Reprinted with the permission of © 2018 Springer Nature

Citation: Wienand I, Rakic M, Elger B, Shaw D. The Beneficence of Hope: Findings from a

Qualitative Study with Gout and Diabetes Patients. Journal of Bioethical Inquiry.

Jun 2018;15(2):211-218.

(Impact Factor: 0.817)

**Abstract** 

This paper explores the importance of hope as a determining factor for patients to participate

in first-in-human trials for synthetic biology therapies.

This paper focuses on different aspects of hope in the context of human health and well-being

and explores the varieties of hope expressed by patients. The research findings are based on

interview data collected from stable gout and diabetes patients. Three concepts of hope have

emerged from the interviews: hope as certainty (H1); hope as reflective uncertainty (H2);

hope as self-therapy (H3).

The purpose of the paper is twofold. First, it aims to underline the significance of hope in

patients' medical decision-making, as well as the beneficence of hope for patients' well-

being, and for progress in research. Second, it shows how philosophical investigations – in

particular Descartes' - explore the phenomenon of hope, and provide medical empirical

research with profitable insights and tools.

Keywords: Hope, Certainty, Optimism, Self-therapy, Virtue, Well-being

74

# Introduction

Hope is crucial for human lives and particularly for those affected by ailments<sup>1,2</sup>. And yet, it is not easy to determine the specific nature of hope, as it is a diffuse phenomenon. Indeed, researchers have identified forty-nine different definitions of hope<sup>3</sup>. As Eliott has noted, the perception of hope differs markedly across time and culture. For instance, Ancient Greece defined hope as a divine evil "that encourages foolish optimism",4, p. 2. In the Latin Christian tradition (e.g. 1 Corinthians 13:13) hope (spes) is a virtue among the three theological virtues (faith, hope, and charity). It is viewed as a God-given disposition – or infused virtue – to direct us to God and thus guides us through our journey on earth. Whereas the Medieval thinker Aquinas insisted that without this infused virtue, we would not feel the emotion of hope<sup>5</sup>, modern philosophers such as Camus<sup>6</sup> (1942) and Bloch<sup>7</sup> (1959) conceive hope in immanent terms. Psychologists and medical scholars do not generally explain hope as an effect of divine grace. However, they do not completely give up the idea that hope is a virtue<sup>8</sup>. Miller reminds us that, for the 19<sup>th</sup> century English physician Thomas Percival, the physician is a "minister of hope and comfort to the sick". Miller endorses this view as "still apt"8. In general, researchers focus their attention upon the psychological mechanisms that generate emotions of hope<sup>9</sup>. Indeed, in the last four decades an increasing number of empirical studies – conducted mainly in psychology and psychiatry – have developed tools for measuring hope, for example "The Hope Scale" (see 10). This measures the effect of hope on bodily ailments and illnesses (see<sup>11</sup>). Andresen et al. demonstrate that hope can play a role in shortening recovery processes<sup>12</sup>. Huguelet adds that hope may be both a motivating factor for healing more promptly and a *consequence* of healing<sup>13</sup>. Maier and Shibles (2012) define hope as an emotion which produces positive bodily feelings<sup>14</sup>. Maier and Shibles also establish a direct link between hope and humor:

Both hope and humor take and run around life's problems. A negative situation may not be within our control, but humor is. [...] The most hopeless situation is death, which is why humor is one of the few ways in which it can be coped with or explored 14, p. 150.

Rivka Jacoby<sup>15</sup> shows that hope is also a resourceful aid for coping in stressful situations. Particularly for terminally ill patients, the significance of hope has been abundantly documented (see<sup>16,17</sup>). As Cotter and Foxell note, health professionals in palliative care widely agree that,

...although hope tends to change in people with terminal illness, maintaining a delicate balance of death and hope for a cure often remains an important task up until the time of death, even when people acknowledge that cure is virtually impossible. The dying person needs to envision future moments of happiness, fulfilment, and connection 18, p. 7.

In a similar vein, recent empirical research in psychiatry has showed hope to be important for the recovery process of schizophrenia patients, even when their chances of reintegration into society are limited (see<sup>19</sup>).

The significance of hope can also be identified in the context of first-in-human (FIH) trials, as we observed in thirty-six interviews conducted with stable gout and diabetes patients. The aim of the interviews was to have a better understanding of patients' attitude towards hypothetical participation in cutting edge biotechnology research with a special focus on synthetic biology. Hope emerged as a distinctively rich theme – explicitly or tacitly present in many interviews with stable, but chronically ill patients with decreased life expectancies. The present paper attempts to clarify the importance of hope for patients who participate in clinical research, and particularly in FIH trials. This article thus explores a novel research

question that has hardly been discussed in the literature<sup>20,21</sup>. Based on the patients' narratives we identified three kinds of hope: hope as certainty (H1); hope as reflective uncertainty (H2); hope as self-therapy (H3). As the interviews illustrate hope can express a strong confidence in the goodness or beneficence of the scientific research's outcome, whether for the patient, or for future patients (H1). Yet, hope can also imply scepticism, fear, and concerns about the outcome of scientific research (H2). Finally, hope can also function as self-therapy allowing patients to remain positive despite uncertain health outcomes (H3). The purpose of the paper is twofold. First, it aims to underline the significance of hope in patients' decision-making and to encourage the promotion of hope, as beneficent not only for patients' well-being, but also as a component of research. Second, it shows how philosophical investigations – in particular Descartes'– explore the complex phenomenon of hope and provide medical empirical research with useful theoretical insights and methodological tools. Descartes emphasizes the plasticity of hope: hope is a psychological coping strategy; hope can coexist with uncertainty; and hope is a great incentive to recover health.

### 1. Various Understandings of Hope

As Albert Camus writes in his philosophical essay *The Absurd*, hope arises when meaningfulness is at stake, that is, when we have to deal with purposeless events, or more fundamentally when we try to make sense of the vulnerability and finitude of our lives: "Absurdity, hope and death carry on their dialogue", (our translation; see also<sup>22</sup>). In a similar vein to the existentialist way of evaluating hope as a way to comprehend the apparent absurdity of human existence, the atheist Marxist-inspired conceptions of hope such as Ernst Bloch's utopic model shares with Camus that hope is an immanent principle of moral agency. In his influential book *Das Prinzip Hoffnung*<sup>7</sup> (1959) Bloch shows that hope is a strong

motivating force of political actions. Without hope for a more just society, no dialectic change in history would be thinkable, no political reform would ever take place.

Despite their evident contrasts, the transcendent religious and the immanent atheistic models of hope agree on hope as confidence in the good that humans can potentially achieve. Being hopeful seems to be prima facie a beneficial trait of character, for it encourages committing oneself to noble causes such as childcare, education and healthcare in the world. In this regard, hope has a critical normative dimension as it challenges unfair conditions: for example insufficient childcare, and societal exclusion of people with mental conditions<sup>19</sup>. As well as being a strong motivational drive to positively change aspects of one's own life or to contribute to modify the course of history for the better, hope also generates a generally positive outlook on the potential of humans. An interesting consideration of hope raised by Aguinas concerns whether one can teach and learn hope. Aguinas thinks it is the effect of divine grace, whereas Bloch defends the views that it can be acquired by human efforts. Recent literature reports that a variety of hope-therapy programmes have been designed<sup>23</sup>. In contrast to despair, hope turns us into optimistic people. However, the idea of "hope" is not fully interchangeable with "optimism". Averill, Caitlin, and Chon suggest we should distinguish hope from optimism. Hope is a positive attitude toward desirable but unlikely events, whereas optimism expresses confidence in events which are likely to take place. So for instance we do not need hope for tomorrow to come, but we need hope in order to think that a fairer distribution of medical care is possible. More importantly, one needs hope, and not optimism, to defend a lost cause, for hope is not about evaluating the world as it is in a

-

positive way, but about believing that the world could be better than what it is. Optimism is a

<sup>&</sup>lt;sup>1</sup> We are not speaking here about the classical form of metaphysical optimism, as it is defended by Leibniz in his *Theodicy* (published in 1710 [24. Leibniz G.W. *Theodicy*. *Edited by A. Farrer and translated by E.M. Huggard*. New Haven: Yale University Press1952.]). Optimism is understood here in a more psychological sense, that is as a tendency to interpret positively what happens and will happen to others and oneself.

Weltanschauung which considers the world as predominantly good. In contrast, hope is a belief that the world as it is, is not predominantly good and that it could (or should) be different. Another difference highlighted by Averill and colleagues is that the events we are hoping for *matter* a lot to us. This personal involvement need not be the case for optimism. This latter distinction is harder to substantiate and we tend to disagree with Averill on the point, as both hope and optimism seem to imply a certain amount of personal engagement with the world. Indeed, a favourable course of events matters as much to the optimist as to the hopeful person. Hope differs from optimism in that hope implies that certain states of affairs should be avoided, even if it seems *prima facie* unlikely, if not impossible, to change them. In contrast, an optimist would typically look for the positive in any state of affairs, on the premise that the world as it is, is fundamentally good. In other words, hope has a stronger utopian dimension that is more critical of cultural, social, and political affairs than mere optimism. Hope evaluates the present in view of a possible and better future, while the optimist is content with the existing condition, the status quo. Our suggestion is that these distinctions between hope and optimism can also be relevant within a medical context. Consider this sentence from a physician to her/his incurable patient: "although there is little reason for optimism, don't lose hope", p. 96. Smith et al. (2011) confirm the common occurrence of this statement in their study. The authors explain that the more honest and precise information patients receive, the more hopeful they can be (see<sup>25</sup>). A recent study shows also that "disclosure of prognosis by the physician can support hope, even when the prognosis is poor". 26, p. 5636. For certain persons, to be hopeful means to believe in miracles. Cooper et al. also remind us that "many people, including healthcare providers, believe that miracles can and do happen, even in the most traumatic experiences" 16, p. 2. The authors add that the belief in miracles can refer to at least two different conceptions:

Some writers note that the belief in miracles is based on irrationality, meaning that something will occur despite the laws of sciences. Others frame the belief in a miracle as a statement of faith or piety. (p. 2)

However, there is also good reason to doubt whether hope can be given any epistemic value, and has any quantifiable use at all, since even the optimist recognizes the medical fact that the patient is incurable. So what is the point of hope? In certain ways hope seems to be the last evil in Pandora's Box, an illusion which we should fight against, as recommended by Hesiod and Aesop<sup>4</sup>. Should we not be more suspicious about the psychological twist of hope, as Ludwig Feuerbach<sup>27</sup> showed in his critical analysis of religion? For the German philosopher, hope, particularly Christian hope is an infra-conscious projection of our desires for divine immortality (see<sup>28</sup>). Seen in this light, hope is a kind of defiance and denial of our finitude.

Indeed, there are some cases in which hope seems to be an inadequate or even bad habit, a self-deceptive behaviour<sup>9</sup>. Psychologists speak of "false hope"<sup>4,29</sup>. The English language has a word to depict this out-of-place confidence: "pollyannaism". The *Oxford English Dictionary* defines pollyannaish as "naively cheerful and optimistic; unrealistically happy" (OED Online 2018<sup>30</sup>). Ruddick<sup>31</sup> emphasizes that physicians should avoid supporting patients' false hopes, as this attitude amounts to a "paternalistic violation of patient autonomy"(p. 343). In the field of research ethics, unrealistic optimism can lead to what is called "therapeutic misconception", when the patient conflates research with therapy; when s/he believes that "every aspect of the research project [...] is designed to benefit him [or her] directly"<sup>32, p.12</sup>, see also<sup>21</sup>.

To sum up, hope plays a key role in human lives as it helps us face uncertain or fatal outcomes and interpret them in a pro-active sense. We have seen that philosophers and ethicists are wary of the potential harm of hope, as it rests on belief about improbable events and can lead to self-deception. However, our empirical data illustrate that the epistemic poverty of hope does not seem to be a problem for the interviewed patients. First, the medical knowledge that there will be little or no improvement in the patients' condition can coexist with the hope for a better outcome for future generations. Second, patients use hope as a measurable good ("little hope"), i.e. to indicate a low level of knowledge. Third, they seem to conceive of hope as a kind of therapeutic tool helping them to get through the day and envisage their near future cheerfully. While trying to identify the different combinations of hope in our collected material, we realized that Descartes made similar distinctions regarding hope in his private correspondence with Princess Elisabeth of Bohemia. The next section analyses our empirical data on hope in light of the philosophical insights of Descartes. We suggest that (1) the phenomenon of hope, as expressed in the interviews with gout and diabetes patients, is complex, even paradoxical; (2) the finesse of Descartes' descriptions of hope seems to capture the complexity of hope and makes him in that regard a valuable resource for today's debate in medical ethics; and (3) health professionals should be perhaps more aware of the plasticity of hope: hope can be an expression of self-deception but also of moral certainty, reflective uncertainty, and mental health.

# 2. Three Kinds of Hope

### Methodological Remarks

In the thirty-six interviews conducted with stable patients suffering from gout and diabetes, we analysed how patients reacted to hypothetical participation in FIH trials using synthetic biology devices. The interview guide consisted of ten open questions; each with several sub-

questions prompting patients on particular details. During the interview, the interviewer provided information about potential medical applications of synthetic biology and FIH research. Each interviewed patient was provided with the hypothetical example of the implantation of synthetically modified cells. Patients were informed that the hypothetical implanted cells would cure diabetes and gout and make other medication redundant. The average interview duration was forty-four minutes. Interviews were transcribed verbatim and quotes were translated into English. We applied a bottom-up, rather than a top-down approach: we did not use an existing theoretical framework of hope while reading the interviews. Hope emerged as one of the themes from the interview material (see<sup>33</sup>).

This main theme was divided into three subcategories as patients referred to different kinds of hope in the medical setting<sup>2</sup>: hope as certainty (H1); hope as reflective uncertainty (H2); hope as self-therapy (H3). In the final phase the chosen theme, hope, including the empirically identified three sub-forms of hope – H1, H2, and H3 – were analysed in the context of Descartes' moral philosophy. Finally, a recommendation was made for clinical settings: hope is not only a positive attitude towards a particular event but can even entail fear and doubt. However, hope implies an essential confidence in the ultimate outcome of events which matter to us. As such, hope should be promoted for the sake of both patients' well-being and clinical research. Details of the methods are described elsewhere<sup>33</sup>.

## 3.1 Hope as Certainty

For a number of patients, hope expresses an unshakable confidence in the effectiveness of medical treatment for themselves, and/or for future generations of patients. Hope expresses the patients' trust that clinical research will contribute to human well-being in the long run.

<sup>&</sup>lt;sup>2</sup> The paper does not claim that the kinds of hope occurring in the interviews are the only existing kinds of hope. For an overview of hope theories, see Rand and Cheavens 2009.

Consider for instance patient Danielle's answer to the question of whether she has hope that the trial can be helpful: "Yes, absolutely. Yes, I have the hope that it will be of help in the future." See too the general statement from patient Yves about confidence in future therapies: "One hopes that there will be better therapies in the future to help others."

The certainty which these patients envisage is not based upon scientific research, or upon complicated statistics. Neither is it the result of a long and metaphysical meditation or a blind faith in medical progress<sup>34</sup>. Rather, it is a kind of moral certainty, that is, a positive evaluation of present and future human potentialities as well as a strong faith in the overall goodness of things. In other words, hope is a conviction that things will turn out for the best. This seems to be a *spiritual resource*<sup>3</sup> that is available to us in adverse situations, particularly in the case of serious illnesses<sup>4</sup>. The view that hope is a coping strategy, a form of medicine for the soul, is not a new idea. It was a widespread view in early modern moral philosophy, which was strongly advocated by Descartes<sup>35,36</sup>. For Descartes, spiritual fortitude does not only relate to the mental part of the human: being strong affects both mind and body. Thus, according to the Cartesian doctrine of the mind-body union, thoughts can produce improvement or deterioration of bodily health, and vice-versa. Conceived as a certainty about an ultimately good outcome, hope is a medicine for the soul and for the body, as the Latin adage says mens sana in corpore sano (a healthy mind in a healthy body)<sup>2</sup>. Ultimately, the conviction that all will be for the best, despite present pains and worries, helps protect us from interpreting our own fate and death (as well as the fate and death of other human beings) as something inherently negative.

<sup>&</sup>lt;sup>3</sup> "Spiritual" is used here in its etymological meaning: of the, related to the spirit (mens, anima).

### 3.2 Hope as Reflective Uncertainty

Hope can also imply the awareness of uncertainty or doubt regarding the chance of recovery. As patient Chris replies to the interviewer, he would participate in the trial for the following reason: "Yes there is a chance that the treatment is efficacious. Efficacious and supportable. That there is meaning in it."

Chris states further that the reason why a paralyzed patient participates in a clinical trial is that he hopes and does not know whether he will feel better. Chris then concludes that it is an instinctive decision:

...also if it is a paralyzed or damaged [patient], who then says: "Yes", this and that I hope, then I do not know. Then, it is a gut feeling based decision (*Bauchentscheidung*)."

# Patient Danielle speaks of there being "little hope":

And therefore there can always be something, when something foreign enters the body, which causes a reaction. Of course you hope that there will be always positive reactions, but the contrary can happen as well. [...] There is little hope that it [treatment] is already useful, because it is an experiment/trial.

The fact that hope can formulate uncertainty about future outcomes is not idiosyncratic to gout and diabetes patients. On the contrary, the patients interviewed tend to illustrate a recognizable pattern in human psychology. Folkman<sup>37</sup> notices a similar kind of hope among cancer patients. They learn how to live well despite the fact that the outcomes are only "plausible". The French philosopher too compares hope as a speculative form of knowledge in *The Passions of the Soul* (published in 1649) and his moral correspondence. In his letter to Princess Elisabeth of Bohemia from 3 November 1645 Descartes points out this reflective dimension of hope: hope includes uncertainty, or as he writes "conjectures" and "no assurance":

As for the state of the soul after this life, I have much less knowledge of it than M. Digby. For leaving aside what faith teaches us, I confess that, by natural reason alone, we can make many conjectures to our benefit and have some high hopes, but no assurance <sup>38, p. 126</sup>.

Hope is not a blind denial of what human reason cannot control. The particularity of hope as reflective uncertainty is that it does not make us fall into despair. Hope occupies this very subtle position in our self-awareness which *compensates* for our cognitive weaknesses and *reminds* us of them.

We can now better understand the plasticity of the concept referred to earlier: hope implies both *certainty* in the sense of overall faith in the ultimate actuality of goodness and *uncertainty* in the sense of recognition of human limits. In our view, there is no necessary contradiction in stating that hope can have both meanings, since *certainty* is not on the same level as *uncertainty*. *Certainty* here relates to the moral conviction that values which matter to us (such as health and well-being) will materialize one day. *Uncertainty* relates to the awareness of lacking factual knowledge (e.g. medical knowledge). This distinction between moral certitude and reflective uncertainty, which Descartes develops in his *Principles of Philosophy*, is particularly useful here<sup>39</sup>. Not only does it remind us that human life belongs to a great extent to the realm of uncertainty; it also emphasizes that metaphysic certainty – which is a divine attribute – is not a prerequisite for human lives to fare well. Hope is the awareness of lacking this absolute knowledge; it is reflective positive knowledge of one's limits. Thus hope is not an unclear concept with self-contradictory elements. On the contrary, hope has a sound self-therapeutic function when it combines both planes, that is, certainty and uncertainty.

## 3.3 Hope as Self-therapy

It would be a harmful kind of hope if a terminally ill patient hoped that s/he could be cured. An example of a realistic hope would be the wish to die at home. A further justifiable form of hope is that the same terminally ill patient hoped that some treatment could be found for future patients suffering from the same incurable illness. In this hypothetical example, hope produces *consolation*. In other words, the positive emotion of hope provides comfort and therefore compensates for the grief that this patient feels when facing imminent death 40,41. A similar psychological mechanism can be observed with diabetes and gout patients, albeit with the difference that these patients are not in life-threatening conditions. A way to cope with their illness is to hope that research will cure them or future generations. The statement of patient Ines indicates how hoping can be self-therapeutic: she does not consider her illness as an immutable fate but something which can potentially create change for the better: "I always have the hope that they [scientists] will develop something that could help me someday too." Patient Yves shares the view too that research will contribute to better health for himself or future generations. "They [patients] hope to get a better therapy with time."

The significance of hope functioning as self-therapy seems to apply for life-threatening illnesses, such as cancer<sup>42</sup>. In his letter to Elisabeth of Bohemia from May or June 1645 Descartes emphasizes too the importance of hopeful thoughts for the sake of recovering health:

In this regard [the curing of sadness], I judge the waters of Spa very appropriate, especially if your Highness in taking them observes what the doctors usually recommend, and clears her mind entirely of all sorts of unhappy thoughts, and even also of all sorts of serious meditations concerning the sciences. She should occupy herself by imitating those who convince themselves they think of nothing in looking at the greenery of a wood, the colors of a flower, the flight of a bird, and such things that

require no attention. This is not to waste time but to employ it well. For one can, in doing this, satisfy oneself by the hope that by this means one will recover perfect health, which is the foundation of all the other goods that one can have in this life. (<sup>38, p.</sup> our italies)

Descartes' suggestions that Princess Elisabeth treat her depressive states by diverting her mind to pleasant activities are still valuable insights for today's range of therapies offered to cancer patients, such as music therapy<sup>43</sup>.

## 3. Conclusion

Based upon thirty-six interviews conducted with diabetes and gout patients, this paper shows the significance of hope in the context of clinical research. The findings presented and discussed are original in as much as research on hope has been focusing, until now mainly on palliative care and terminally ill patients. Our study aimed at finding out the reactions of diabetes and gout patients to hypothetical participation in FIH trials. In their answers, the interviewees used hope in a cognitive sense: hope can imply certainty (H1) and reflective uncertainty (H2). In our view, the affirmation of H1 and H2 does not amount to a self-contradiction. On the contrary, the patients' answers showed the useful nature and plasticity of hope: as a coping strategy against uncertainty and anxiety (H1 and H2) and as self-therapy (H3). These results are positive in the sense that they confirm the therapeutic potency of hope. They also corroborate the surprising functioning of our mental resources: being ill does not make one necessarily despaired, but prompts one to be hopeful for a treatment to be found. Moreover, these findings show how philosophical investigations – in particular Descartes' – explore the phenomenon of hope and provide medical research with useful resources to interpret the seemingly paradoxical nature of the empirical data on hope.

Finally it is important to stress that the three aspects of hope (H1, H2, and H3) are compatible with modern standards of bioethics in therapy and in research, in particular related to respect for patient autonomy. A traditional myth stemming from the era of medical paternalism has been that patients should not be fully informed by doctors or researchers in order to preserve hope. However, empirical studies from the past fifty years prove the contrary (see<sup>44</sup>): these studies have shown that full and clear information is beneficial. Patients suffer psychologically the most from uncertainty and deception. Using the three aspects of hope identified both in patients' answers and reflected in the works of Descartes helps to understand why the traditional paternalistic understanding of hope is a misconception. Indeed, the combination of H1 and H2 coexists with honest information as hope is an innate certainty that is not the same as optimism following information about a good prognosis. Physicians can and should reinforce hope as self-therapy (H3) in the same way that it exists in dying patients: hope is able to cope with the worst case, the end of life. Hope in these three senses does not mean exaggerating benefits to future patients or reinforcing therapeutic misconceptions in patients participating in research and it does not mean hiding information from a dying patient.

In short, health professionals should be more aware of the great potential of hope as a coping strategy for patients. Despite its apparent contradiction, hope seems to be genuinely beneficent for patients' well-being in a therapeutic and in a research context.

# Acknowledgments

We thank Anna Genske and Rebecca Nast for conceiving and conducting the interviews with thirty-six gout and diabetes patients, as well as to Marianne Weber for transcribing the interviews.

# References

- 1. Hammelstein P, Roth M. Hoffnung Grundzüge und Perspektiven eines vernachlässigten Konzeptes [Hope: principles and perspectives of a neglected concept]. Zeitschrift für Differentielle und Diagnostische Psychologie. 2002;23(2):191-203.
- 2. Scioli A, Biller HB. *Hope in the Age of Anxiety*. Oxford: Oxford University Press; 2009.
- 3. Schrank B, Stanghellini G, Slade M. Hope in psychiatry: a review of the literature.

  \*\*Acta psychiatrica Scandinavica. Dec 2008;118(6):421-433.
- 4. Eliott J. What have we done with hope? A brief history. In *Interdisciplinary* perspectives on hope, edited by JA Eliott. New York: Nova Science Publishers; 2005.
- 5. Aquinas T. Summa Theologica I-II.62.1. Second and Revised Edition, 1920.

  Translated by Fathers of the English Dominican Province. . 1920;

  <a href="http://www.newadvent.org/summa/2062.htm">http://www.newadvent.org/summa/2062.htm</a> Accessed 05 February 2018. .
- 6. Camus A. Le mythe de Sisyphe: essai sur l'absurde: Nouvelle édition augmentée d'une étude sur Franz Kafka. Paris: Gallimard; 1942.
- 7. Bloch E. Das Prinzip Hoffnung. Frankfurt/Main. Suhrkamp; 1959.
- 8. Miller G. Hope is a virtue. *Journal of Child Neurology*. Dec 2012;27(12):1616-1617.
- 9. Averill JR, Catlin G, Chon KK. *Rules of Hope*: Springer: New York; 1990:9-35.
- 10. Roth M, Hammelstein P. Hope as an emotion of expectancy: first assessment results.

  \*Psycho-Social Medicine. 04/12 2007;4:Doc05.
- 11. Snyder CR. *Handbook of hope: Theory, measures, and applications*. San Diego: Academic press; 2000.

- 12. Andresen R, Oades L, Caputi P. The experience of recovery from schizophrenia: towards an empirically validated stage model. *The Australian and New Zealand journal of psychiatry*. Oct 2003;37(5):586-594.
- 13. Huguelet P. The Contribution of Existential Phenomenology in the Recovery-Oriented Care of Patients with Severe Mental Disorders. *Journal of Medicine and Philosophy.* 2014;39(4):346-367.
- 14. Maier B, Shibles WA. *The philosophy and practice of medicine and bioethics: a naturalistic-humanistic approach*. Vol 47: Dordrecht: Springer; 2011.
- 15. Jacoby R. 'The miserable hath no other medicine, but only hope': Some conceptual considerations on hope and stress. *Stress and Health.* 1993;9(1):61-69.
- Cooper RS, Knight L, Ferguson A. Trust, Hope, and Miracles. Edited by E.
   Wittemberg. Oxford: Oxford University Press; 2015.
- 17. Rocker GK et al. End of life care in the ICU: from advanced disease to bereavement.

  Oxford University Press; 2010.
- 18. Cotter VT, Foxwell AM. *The meaning of hope in the dying*. Oxford: Oxfrod University Press; 2015.
- Schrank B, Wally J, Schmidt B. Recovery and hope in relation to schizophrenia.Oxford: Oxford University Press; 2012.
- 20. Eaves ER, Ritenbaugh C, Nichter M, Hopkins AL, Sherman KJ. Modes of hoping: understanding hope and expectation in the context of a clinical trial of complementary and alternative medicine for chronic pain. *Explore (New York, N.Y.)*. Jul-Aug 2014;10(4):225-232.
- 21. Jansen LA. Two concepts of therapeutic optimism. *Journal of medical ethics*. 05/07 2011;37(9):563-566.

- 22. Marcel G. *Homo viator prolégomènes à une métaphysique de l'espérance* [Homo viator. Prolegomena to a metaphysics of hope]. Paris: Montaigne; 1947.
- 23. Lopez SJ, Ciarlelli R, Coffman L, Stone M, Wyatt L. *Hope therapy: Helping clients build a house of hope. In Handbook of hope: Theory, measures, and applications,* edited by CR Snyden. Sand Diego, CA: Academic; 2000.
- 24. Leibniz G.W. *Theodicy*. Edited by A. Farrer and translated by E.M. Huggard. New Haven: Yale University Press1952.
- 25. Smith TJ, Dow LA, Virago EA, Khatcheressian J, Matsuyama R, Lyckholm LJ. A pilot trial of decision aids to give truthful prognostic and treatment information to chemotherapy patients with advanced cancer. *The journal of supportive oncology*. Mar-Apr 2011;9(2):79-86.
- 26. Mack JW, Wolfe J, Cook EF, Grier HE, Cleary PD, Weeks JC. Hope and prognostic disclosure. *Journal of clinical oncology: official journal of the American Society of Clinical Oncology*. Dec 10 2007;25(35):5636-5642.
- 27. Feuerbach L. *The Essence of Christianity*, translated by G. Eliot. New York: Dover publications; 2008.
- 28. Godfrey JJ. Hope-In. *A Philosophy of Human Hope*: Dordrecht: Martinus Nijhoff; 1987:37-46.
- 29. Snyder CR, Rand KL. The case against false hope. *American Psychologist*. 2003;58(10):820-822.
- 30. Oxford English Dictionary Online. Pollyannaish, adj. 2018; <a href="http://www.oed.com/view/Entry/263983?redirectedFrom=pollyannaish#eid">http://www.oed.com/view/Entry/263983?redirectedFrom=pollyannaish#eid</a>. Accessed February 22, 2018.
- 31. Ruddick W. Hope and deception. *Bioethics*. 1999;13(3-4):343-357.

- 32. Horng S, Grady C. Misunderstanding in clinical research: distinguishing therapeutic misconception, therapeutic misestimation, and therapeutic optimism. *IRB: Ethics and Human Research*. 2003;25(1):11-16.
- 33. Rakic M, Wienand I, Shaw D, Nast R, Elger BS. Autonomy and Fear of Synthetic Biology: How Can Patients' Autonomy Be Enhanced in the Field of Synthetic Biology? A Qualitative Study with Stable Patients. *Science and engineering ethics*. Apr 2017;23(2):375-388.
- 34. Stempsey WE. A new stoic: the wise patient. *The Journal of medicine and philosophy*. Aug 2004;29(4):451-472.
- 35. Levi A. French Moralists the Theory of the Passions, 1585 to 1649. Oxford: Clarendon Press.; 1964.
- 36. Descartes R. *Der Briefwechsel mit Elisabeth von der Pfalz*. Edited by I. Wienand andO. Ribordy. Vol 659: Hamburg: Meiner; 2015.
- 37. Folkman S. Stress, coping, and hope. *Psychological aspects of cancer*: Springer: New York; 2013:119-127.
- 38. Shapiro L. The correspondence between princess elisabeth of bohemia and rene descartes. Chicago: Chicago University Press; 2007.
- 39. Descartes René. *The Philosophical Writings of Descartes, vol 1.* Translated by J. Cottingham, R. Stoothoff, and D. Murdoch. Cambridge: Cambridge University Press; 2006.
- 40. Chochinov HM. Thinking outside the box: depression, hope, and meaning at the end of life. *Journal of palliative medicine*. Dec 2003;6(6):973-977.
- 41. Herth K. Fostering hope in terminally-ill people. *Journal of advanced nursing*. Nov 1990;15(11):1250-1259.

- 42. Shekarabi-Ahari G, Younesi J, Borjali A, Ansari-Damavandi S. The effectiveness of group hope therapy on hope and depression of mothers with children suffering from cancer in tehran. *Iranian journal of cancer prevention*. Fall 2012;5(4):183-188.
- 43. Boyde C, Linden U, Boehm K, Ostermann T. The Use of Music Therapy During the Treatment of Cancer Patients: A Collection of Evidence. *Global Advances in Health and Medicine*. 11/01 2012;1(5):24-29.
- 44. Elger BS. Le paternalisme médical: mythe ou réalité? Aspects philosophiques et empiriques d'un phénomène persistant. Genève: Editions Médecine & Hygiène; 2010.

# **Chapter 5**

\_

# Feelings of Burden in Palliative Care: a Qualitative Analysis of Medical Records

Reprinted with the permission of © 2018 SAGE Publications

Citation: **Rakic M**, Escher M, Elger BS, Eckstein S, Pacurari N, Zwahlen S, Wienand I. Feelings of Burden in Palliative Care: A Qualitative Analysis of Medical Records. *Journal of Palliative Care*. Jan 2018;33(1):32-38.

(Impact Factor: 0.618)

### **Abstract**

**Background** Care for palliative care patients is often provided by unpaid caregivers (e.g. family members) who take care of the patient's daily needs (e.g. bathing, dressing). Family members of palliative care patients are involved in numerous ways. These tasks and responsibilities can make them feel burdened and even overburdened.

Aim We specifically looked at patients' medical records to determine what is being reported about burden and overburden and who seems to be mostly affected. Burden was understood as a weight or task that is difficult to accept or carry, whereas overburden indicates that this weight or task cannot be carried anymore.

**Methods** We looked at 300 medical records of palliative care patients written by health-care professionals. Written notes were analysed using latent content analysis as it helps to analyse large amounts of textual data qualitatively and to understand the underlying concepts of what was said.

Results Most patients (73.5%) had a cancer diagnosis. Mean age was 67.6 years (range, 22-98 years). Burden and overburden were identified as main categories and further divided into the following sub-categories: for patients and families. According to the written notes, patients often felt burdened by their disease, financial problems, situation at home and families' reactions to their disease. By and large, patients felt overburdened by their own disease. Families often felt burdened because of issues related to patients' medical condition, providing home care or financial and social aspects. Families mentioned homecare and the decision-making process as being overburdening.

**Conclusion** Findings in the palliative care patients' medical records are inasmuch important, as they point at the health-care staff's awareness of possible weights and tasks that might be burdensome for patients and their families. Attention should be drawn to the documentation of medical records in order to identify recurrent difficulties and to help discuss these.

**Keywords:** palliative care; burden; qualitative research; caregivers

# Introduction

The Swiss Federal Office for Public Health (FOPH) estimates that 40,000 people are currently in need of palliative care (PC), and this number will increase to 53,000 people by 2032<sup>1</sup>. According to the World Health Organization, PC encompasses an interdisciplinary and holistic approach focusing on several domains of care, such as physical, social, psychological and spiritual care of patients and their families<sup>2</sup>. Another important defining principle of PC is the promotion of autonomy<sup>3</sup>. Palliative care offers a range of support services to help families cope with the patients' illness, but also with the families' own problems<sup>2</sup>. Care for PC patients is often provided by unpaid caregivers<sup>4</sup>, or someone (e.g. family member, friend, neighbor) who takes care of the patient's daily needs (e.g. bathing, dressing)<sup>5</sup>. Family members of PC patients are involved in numerous ways: They care for the patients' daily needs, are in close contact with health professionals, and sometimes act as the patients' surrogates. These multiple roles can affect their own health, as Kristjanson and Aoun suggest<sup>6</sup>. Older caregivers most often provide care to partners, friends, and neighbors, whereas younger caregivers typically provide care to their parents<sup>4</sup>. Emanuel et al. suggest that even for care that requires qualification (e.g. nursing) patients generally receive care from family members or friends rather than from paid caregivers<sup>7</sup>. However, family members often indicate a lack of self-confidence in providing care for their relatives<sup>8</sup>.

The FOPH states in its report on the PC situation in Switzerland that the inclusion of informal caregivers in the provision of care is a central need of PC patients. The report also emphasizes that patients fear that they could be a burden for their family caregivers<sup>9</sup>. According to a generally accepted definition, a burden is a weight or task – in a physical and psychological sense – that is difficult to accept or carry, whereas overburden indicates that this weight or task – in our case caring for PC patients, their families and friends – cannot be carried anymore<sup>10,11</sup>.

Compared to younger PC patients, older patients are more frequently isolated (e.g. loss of partner) and have a limited informal care network. Family caregivers who provide care for older patients may feel burdened by such a challenging task<sup>12</sup>. Furthermore, the ethical principle of autonomy, which is highlighted by the FOPH, might be difficult to respect and implement fully because of possible barriers regarding the provision of home care<sup>9</sup>. For example, patients want to spend their last days at home, but their relatives cannot provide the desired homecare as it is too burdensome<sup>13</sup>. The family's needs should also be recognized and taken into account, so they can support their loved one. Hence, it is important to address possible hindering factors – such as burden and overburden of patients and of their families Since the number of patients in need of PC will increase in the future<sup>1</sup> and caregivers often feel burdened by caring for them<sup>12</sup>, our study focused on the different notions of burden and overburden in a PC context. More precisely, we looked at patients' notes to determine what is being reported about burden and overburden in the medical records and who seems to be mostly affected according to these notes. To do so, we focused on written notes of the healthcare staff working in a PC setting. Using latent content analysis, we analysed 300 medical records of patients who received PC at 3 Swiss university hospitals.

# **Methods**

## Study design

We employed a qualitative research design since it helps to investigate patients' attitudes and preferences in more depth than quantitative research<sup>14</sup>. We used content analysis to analyse the large amounts of textual data (notes or comments in medical records)<sup>15</sup>. Moreover, in order to ensure high quality in reporting qualitative research, we have applied – where possible – the 32-item consolidated criteria for reporting qualitative research (COREQ)

checklist<sup>16</sup>. In total, we examined 300 medical records from 3 university hospitals in Switzerland using a self-developed data extraction sheet.

#### **Data extraction sheet**

The data extraction sheet focused on patients' conditions, and specifically on their attitudes and preferences regarding PC. The extraction sheet covered the following aspects: (1) demographics, (2) diagnosis, (3) information about advance directives, and (4) all notes written by the medical team about patients' attitudes and preferences regarding PC. The extraction sheet consisted of items with categorical responses (e.g. gender), continuous variables (e.g. age), and open-ended items (e.g. notes about patient wishes and preferences). Items were developed from the research team's knowledge in the field (M.R., I.W., B.S.E.) and based on discussions with collaborating physicians (M.E., S.E., S.Z.).

## **Study population**

Inclusion criteria were (1) patients received PC at one of the three Swiss university hospitals, and (2) patients were older than 18 years of age. The necessary information (lists of patients) was provided by the research partners at the 3 university hospitals (A, B, C). Exclusion criterion was if a patients' medical record had already been collected and the same patient appeared again later on in the list, because of multiple visits in the hospital during the period of data collection. In such a case, this patient was excluded in order to avoid a duplicate. All information that could lead to an identification of patients was deleted after data collection.

#### **Data collection**

Data collection took place between April and September 2016 and was carried out by 4 research team members. Ethical approval was obtained from the local research ethics committee (EKNZ; Nr. EK 2015-197). Because only routine non-genetic data were collected, informed consent from each patient was not needed (providing an opt-out), based on the Swiss federal law of human subject research<sup>17</sup>. Patients are routinely informed that health-

related data may be collected for research purposes during their hospital stay, and those who refuse such data collection must actively request exclusion. Before starting with data collection, the responsible physicians and data managers from each university hospital provided access to patients' medical records. All medical records were available in digital format. We collected data throughout the patients' medical records (e.g. cover page with mostly demographic information, such as age, sex, and diagnosis), PC reports (special focus on advance care planning, patient's wishes, discussions which took place with the patient), and discharge reports (which provided a good summary of the whole hospitalization of a patient). We searched the notes written by all the health-care professionals (e.g. physicians, nurses, psychologists). We included in the analysis summarized discussions, which took place between patients and health-care professionals (e.g. case manager, social worker, physician, nurse) and between physicians and families (and patients). We used the information written between the first PC consultation and the day the patient either left the hospital or died. Each patient received a special code, which ensured anonymity regarding the gathered data. Researchers who extracted the data discussed the first 5 extractions to achieve standardization of extraction, and continuously discussed their extractions, when needed. Patients were included sequentially. More specifically, we started with the patient who received PC on January 1 2016, and continuously went back until the year 2015, thereby extracting 100 medical records per hospital. The same procedure was followed in all three university hospitals resulting in a final number of 300 extractions. Patients' demographics and characteristics are presented in Table 2.

## **Data analysis**

Data were analysed using software for qualitative research (MAXQDA 12). Content analysis was chosen as it helps develop categories and analyse large amounts of textual data qualitatively<sup>15</sup>. We were particularly interested in the underlying concepts of what was written. This approach is known as latent content analysis 18. The analysis process started with several readings of all extracted data by 1 researcher (M.R.), aiming for an overview of the data. In the next step, 2 members of the research team (M.R., I.W.) agreed on the main themes that emerged from the data. In this paper, we report on one of them: health-care staff's perception of burden and overburden. This topic was considered important as it was found recurrently across the medical records. M.R. started with an initial open coding and then organized the codes into main categories as well as sub-categories. I.W. crosschecked the coded passages. Then, based upon the text material, they made a distinction between "burden" and "overburden". We used the generally accepted definition of burden and overburden, that is, a burden is a weight or task – in a physical and psychological sense – that is difficult to accept or carry<sup>11</sup>. Overburden, indicates that the weight or task – in our case caring for PC patients, their families and friends – cannot be carried any more<sup>10</sup>. As found in the literature, the term 'family' was understood in a broader way<sup>19</sup> and we therefore also included the category 'friends' in our analysis. The research team defined "burden" and "overburden" as the 2 main categories, after checking that both terms were used in the research literature<sup>20,21</sup>. M.R. started with the coding regarding "burden" and "overburden". To guarantee accuracy and consistency of the coded segments, I.W. re-read all the previously coded text passages and cross-checked them. All transcripts were analysed in the original languages (French, German). Quotes were translated into English, and this translation was edited by a native English-speaking researcher. All authors agreed on the conceptual map of main categories and sub-categories presented in the results section (see Table 1).

# Results

Our analysis identified burden and overburden as main categories which were further divided into 2 sub-categories: patients and families (see table 1).

Table 1. Main categories and sub-categories from content analysis

Main categories	Sub-categories
	(a) for patients
1. Burden	(b) for families
	(c) for patients
2. Overburden	(d) for families

# Study sample

Forty-seven percent (143/299) of the patients were female. Mean age was 67.6 years, ranging from 22 to 98 years. Cancer was diagnosed in seventy-three percent (211/287) of the patients (see Table 2).

Table 2. Demographics of palliative care patients for each hospital $(n = 300)$			
University hospital	A	В	С
Age (M*; <i>SD</i> *)	70.55 (14.77)	63.02 (14.87)	69.11(14.54
Min., max. age	23-98	22-91	27-94
Gender (female)	51%	39%	53%
Diagnosis (cancer)	73%	78%	70%

<sup>\*</sup>M = mean; *SD* = standard deviation;

## Cause of burden for patients

Patients' perceived burden was mostly related to obligations they felt towards their own family. Diseases were also often mentioned as a cause of burden for patients. Other less mentioned burdens were related to financial problems or to current domestic situation.

## **Family**

Examining possible causes for why patients felt burdened, we found that patients experienced a feeling of burden, as they had insufficient time and space to be alone without family being constantly around them. Moreover, patients were said to have problems talking about diagnosis with the family, because they were worried about the family's reaction. Other patients were described as being afraid to express their emotions because they were embarrassed and because they wanted to protect their family and friends: Sometimes, patients were reported to be more worried about other family members than themselves. At times, the health-care staff noted that this could cause burden for the patients. In these difficult moments, the notes indicated that patients felt the need to clarify their relationship with their children (see Table 3).

## Table 3. Family

"He also means that he is glad that his family is not always with him, as he also needs time for himself" [Patient record (PatRec) 1]

"(...) [Patient] reports difficulties in informing her family about the disease and its progression. She [patient] fears that everyone will cry, actually she wishes for strong people" [PatRec 2]

"She [patient] said that she was ashamed to cry and she didn't do it in order not to worry her close family and friends" [PatRec 3]

"The patient was very sad and scared, he wanted to meet the whole family in the presence of the psychologist in order to discuss about his father role. He had had his first son with his ex/first wife and he was feeling guilty towards this first son" [PatRec 4]

### Disease

The notes indicated that patients seemed aware that their disease was continuous and lifeprolonging measures were counterproductive and, as such, felt burdened by the illness. We found that patients were described as feeling helpless regarding their own illness. Moreover, pain and lack of physical autonomy were sometimes noted as burdensome (see Table 4).

#### Table 4. Disease

"Patient loves her family, but she knows that the disease has progressed and that life-prolonging measures are just a prolongation of suffering" [PatRec 5]

"Psychologist noticed that this anger was related to a feeling of helplessness regarding his disease"

[PatRec 6]

"Patient feels burdened by pain and the associated immobility" [PatRec 7]

# Financial problems

According to the notes of health-care personnel, an additional reason that patients felt burdened was monetary concerns. It was also evident that worrying about money issues sometimes caused additional anger for patients (see Table 5).

## Table 5. Financial problems

"The patient was worried about financial problems" [PatRec 8]

"[...] this anger was related to a feeling of helplessness regarding his disease, his financial and family situation" [PatRec 9]

## Fear of putting too much burden on the partner

Some patients were reported as scared of asking too much from their partners. Patients seemed to realize that their partners also needed some help at home and consequently wished for supervision at home. Sometimes, patients did not want their care to be continued at home

because they knew that their partner could not provide the needed support for them (see Table 6).

Table 6. Fear of putting too much burden on the partner

"Patient wishes for support at home, also to relieve his wife" [PatRec 16]

"Patient knows that she cannot go home, because her partner relies on her "guidance". Partner cannot take care of her" [PatRec 17]

## Situation at home

An important finding was that the current household situation could lead to further burden not only for partners but also for patients. For instance, the illness of their partner is noted to create an additional burden for the patient (see Table 7).

Table 7. Situation at home

"Patient reports about difficult domestic situation, because his partner is suffering from Parkinson's disease" [PatRec 10]

# Cause of burden for families

# Issues related to patients' medical condition

The notes in the medical records mentioned patients' condition and issues related to it (e.g. being in the final stage of the illness) as the most frequent reason why partners felt burdened. Family members mentioned that they needed some space for themselves. Sometimes, they seemed to feel burdened because of the patients' imminent transfer to a hospice. Moreover, relatives were described as having expressed the difficulty of accepting the patient's current medical condition and treatment. Nevertheless, it must be underlined that an initial burden could turn into relief for the family (see Table 8).

## Table 8. Issues related to patients' medical condition

"Discussion with spouse and daughter of patient; they are told why patient is in a terminal stage. Spouse finds difficult to accept the situation (...)" [PatRec 11]

"Wife expresses the wish to sleep in a separate room from her husband in order to be able to have some space for herself" [PatRec 12]

"Wife reports about difficult past weeks. In the beginning, she had difficulties with the transfer to the hospice, because of the fear of an "end station". "However she does not feel that way anymore and is looking forward to the transfer" [PatRec 13]

"Husband sees situation critically: He has big hope that patient will stabilize, to be mobile. On the other hand it was always clear for the patient: "Not at all cost" [PatRec 14]

### Home care

Another recurring aspect was the burden related to caring for patients at home. More specifically, the most frequent feeling of burden was the inability to take care of the patient at home because the patient was too disabled (see Table 9).

### Table 9. Home care

"Patient and wife are discussing the transfer to the hospital [name of hospital]. Wife seems to be relieved not to have to take patient home in this condition" [PatRec 1]

"Conversation with wife, physicians, and patient: Wife cannot presently imagine a return home, because patient is too much in need of care" [PatRec 15]

"Conversation with long-standing friend of patient: patient has always been stubborn; she lives in a shabby flat, if the patient returns home, her friend will hand over her keys, she only sees the patient in a hospital etc." [PatRec 20]

#### Patients' wishes to die

Additional reasons for a partner's feeling of burden were when the patient wished to die with an organization that supports assisted suicide (e.g. EXIT) and the difficulty to accept or deal with this wish (see Table 10).

Table 10. Patients' wishes to die

"Partner wants to help her satisfy this wish, but seems to struggle with this" [PatRec 18]

## Financial and social aspects

Another aspect of burden of care was the burden that partners felt when they had to make non-medical decisions on behalf of the patient, such as bureaucratic or monetary questions. The wife of one patient was worried that she and her son would be left alone after her husband's death (see Table 11).

# Table 11. Financial and social aspects

"She was also dealing with many administrative and financial issues related to the patient's inheritance" [PatRec 19]

"She was aware of the severity of the situation and very affected by the speed of the disease and that she was going to be alone with her son in [name of the city], knowing that her social network was little developed in [name of the city]" [PatRec 19]

### Cause of overburden for patients

### Disease

Overburdening was most often caused by patients' disease. Often, the fast disease progression made them feel overburdened. Disease aggravation could also lead to the inability to cope with the disease. Moreover, a poor symptom management could represent a source of overburden (see Table 12).

### Table 12. Disease

"Patient is heavily burdened by the course of the disease (neurological deterioration)" [PatRec 21]

"Because patient cannot continue to live like this, he was in contact with EXIT, he wants to go this way" [PatRec 18]

"Discussion with patient and nursing assistants from institution: Patient's symptoms are not well-managed, [patient] cries a lot: "I cannot anymore; I cannot stand it any longer" [PatRec 22]

### Cause of overburden for families

## Issues related to patients' medical condition

The most frequent cause for feeling overburdened was patients' medical condition. Moreover, relatives had great difficulties dealing with patients' present circumstances and their constantly deteriorating health status. Partners sometimes mentioned that patients did not receive extensive medical treatment (see table 13).

## Table 13. Issues related to patients' medical condition

"Discussion with wife, cousin and cousins' husband at patient's bedside: Wife reports that she is overburdened with the current situation" [PatRec 23]

"Discussion with husband without patient; [husband] reports about overstrain; the constant aggravation of the patient's situation was a shock for him" [PatRec 24]

"Discussion with wife (alone); she is heavily burdened, she has the impression that patient is not receiving comprehensive care and that he is now 'abandoned'" [PatRec 25]

#### Home care

Partners often mentioned that they could not take care of patients anymore because they needed continual support. Some patients realized that their partners could not take care of them because of the amount of care they needed. Sometimes partners who were already taking care of children felt overloaded by additionally taking care of patient (see Table 14).

### Table 14. Home care

- "Partner considers taking care of the patient and his dog as overburdening, because she [patient] needs care around the clock" [PatRec 26]
- "Patient tells that she hopes to get back home, but she also knows that her partner would be overburdened" [PatRec 17]
- "Situation at home for wife and two little children no longer tenable" [PatRec 27]
- "Patient cannot take care of herself at home anymore, an acquaintance who is supporting her feels overburdened" [PatRec 20]

# **Decision-making**

Partners who needed to make a decision regarding the patient's medical treatment sometimes felt overburdened by this role. Another report shows that not only medical decisions but all kinds of decisions were overburdening for the patient's partner (see Table 15).

## Table 15. Decision-making

- "Husband reports that he was overburdened, when the assistant doctors wanted to know from him if an infection should be treated with antibiotics" [PatRec 24]
- "Wife reports that she was overburdened with the current situation and also with all the decisions that were brought to her" [PatRec 23]

## **Discussion**

Our objective was to highlight documented information about burden and overburden in the medical records of PC patients. More precisely, we wanted to analyse the possible challenges faced by patients and informal caregivers as documented by health-care staff. Furthermore, we aimed to understand the possible underlying reasons which made these stakeholders feel burdened and sometimes even overburdened. So, we analysed how burden and overburden of patients and their families were documented in the patients' medical records. The distinction between burden and overburden was evaluated to analyse when burden becomes overburden. Burden can be further divided and refined into caregiver burden (CB) (e.g. burden which arises of taking care for patients) and self-perceived burden (SPB) (e.g. patient's perception of being a burden to their family members). Moreover, Lee et al. showed a connection between both kinds of burdens and PC treatment. They conclude that patients could choose certain treatments to avoid putting burden on their informal caregivers or if they have the feeling to be a burden (SPB) to their family members. On the other hand, burdened informal caregivers could influence patients' decision regarding certain treatments. Finally, the authors demonstrate that higher CB or SPB could lead to a deprivation of patient autonomy<sup>22</sup>. These findings align with our results. Our analysis showed that it was recurrently written in the medical records that patients expressed the fear of putting too much burden on the partner. A study suggest that CB and SPB can be decreased through several interventions, such as support programs for informal caregivers, advance care planning for patients, promotion of communication<sup>22</sup>, and the organisation of "family conferences" to discuss matters important to patients and their relatives<sup>23</sup>. Our results also suggest that particular attention should be given to families because of their role as informal caregivers. More specifically, we found that home care was often mentioned as being a cause of burden and overburden for patients' families. Our findings agree with current literature. Interview studies with informal

caregivers pointed out that they faced a high risk of being burdened as a consequence of caring for their loved one<sup>24,25</sup>. Proot et al. concluded that caring for a patient at home could be both, a mental and physical burden<sup>24</sup>. Given et al. summarized that the burden on informal caregivers was different from anxiety or depression. As a possible reason for this burden, the authors listed disease progression and the subsequent greater need of care<sup>26</sup>. This topic was also found in the medical records, as it was noted that issues related to patients' medical condition (e.g. disease progression) act as a possible factor for patients' families to feel burdened and overburdened.

Our results show that disease, home care, patients' wishes, financial concerns, and social issues were objective things that patients and families had to deal with and which might let them to feel burdened and even overburdened. Findings regarding what was documented in the medical records are inasmuch important, as they point at the health-care staff's awareness of possible weights and tasks that might be burdensome for PC patients and their families. Moreover, these issues documented by health-care staff appeared to confirm findings from other empirical studies regarding causes of burden and overburden in PC<sup>22,24-26</sup>.

### Limitations

Medical notes were written by health-care professionals, but these notes might not be exhaustive. What is more, if health-care staff are not sensitive enough to the burden issue, the feelings expressed by patients and their families and friends might not be fully recorded in the notes. Furthermore, the medical team might superficially misinterpret signs of burden (e.g. tiredness or physical exhaustion). Finally, these notes might not always give the details or verbatim wording of the dialogue. Sometimes, documentation in medical records of communication leaves out important verbal and non-verbal information.

## Conclusion

Our results indicate that, according to the written notes, patients felt burdened or overburdened because of their own disease, the situation at home or their own family. Moreover, the analysis of medical records suggests that families often seemed to mention issues related to the patient's medical condition, the decision-making process, as well as social and financial issues as being burdensome. To conclude, our analysis showed that – according to the FOPH<sup>9</sup> – the central aim of including informal caregivers in the provision of care for PC patients might be difficult to implement. Attention should be drawn to the documentation of medical records in order to identify recurrent difficulties - such as feelings of burden and overburden documented in the medical records - and to help discuss these difficulties (e.g. "family conferences<sup>23</sup>).

# References

2017.

- 1. Bundesamt für Gesundheit (BAG) and Schweizerische Konferenz der kantonalen Gesundheitsdirektorinnen und -direktoren (GDK). Nationale Strategie Palliative Care 2013–2015. 2012; <a href="http://www.gdkcds.ch/fileadmin/docs/public/gdk/themen/strategie\_bundkantone/palliativecare/bt\_nationale\_strategie\_palliative\_care\_2013-2015\_d.pdf">http://www.gdkcds.ch/fileadmin/docs/public/gdk/themen/strategie\_bundkantone/palliativecare/bt\_nationale\_strategie\_palliative\_care\_2013-2015\_d.pdf</a> Accessed 23 May
- World Health Organization. WHO Definition of Palliative Care. 2002;
   <a href="http://www.who.int/cancer/palliative/definition/en/">http://www.who.int/cancer/palliative/definition/en/</a> Accessed 15 May 2017.
- World Health Organization Regional Office for Europe. Better Palliative Care for Older People.
   http://www.euro.who.int/\_\_data/assets/pdf\_file/0009/98235/E82933.pdf Accessed 15 May 2017.
- 4. Marcus J. Hollander GL, Neena LC. Who Cares and How Much? The Imputed Economic Contribution to the Canadian Healthcare System of Middle-Aged and Older Unpaid Caregivers Providing Care to The Elderly. *Healthcare Quarterly*. 2009;12(2):42-49.
- 5. Reinhard SC GB, Petlick NH, et al. Supporting Family Caregivers in Providing Care. 2008; https://www.ncbi.nlm.nih.gov/books/NBK2665/ Accessed 12 May 2017.
- 6. Kristjanson LJ, Aoun S. Palliative Care for Families: Remembering the Hidden Patients. *The Canadian Journal of Psychiatry*. 2004;49(6):359-365.
- 7. Emanuel EJ, Fairclough DL, Slutsman J, Alpert H, Baldwin D, Emanuel LL. Assistance from Family Members, Friends, Paid Care Givers, and Volunteers in the Care of Terminally Ill Patients. *New England Journal of Medicine*. 1999;341(13):956-963.

- 8. Bucher JA, Loscalzo M, Zabora J, Houts PS, Hooker C, BrintzenhofeSzoc K. Problem-Solving Cancer Care Education for Patients and Caregivers. *Cancer Practice*. 2001;9(2):66-70.
- 9. Bundesamt für Gesundheit SKdkGu-dGupc. Allgemeine Palliative Care Empfehlungen und Instrumente für die Umsetzung. 2015; https://www.bag.admin.ch/bag/de/home/themen/strategien-politik/nationalegesundheitsstrategie-palliative-care/grundlagen-zur-strategie-palliativecare/allgemeine-palliative-care.html Accessed 23 June 2017.
- 10. Merriam-Webster. Definition of overburden. <a href="https://www.merriam-webster.com/dictionary/overburden">https://www.merriam-webster.com/dictionary/overburden</a> Accessed 2 June 2017.
- 11. Merriam-Webster. Definition of burden. <a href="https://www.merriam-webster.com/dictionary/burden">https://www.merriam-webster.com/dictionary/burden</a> Accessed 2 June 2017.
- 12. Garlo K, O'Leary JR, Van Ness PH, Fried TR. Caregiver Burden in Caregivers of Older Adults with Advanced Illness. *Journal of the American Geriatrics Society*. 11/18 2010;58(12):2315-2322.
- 13. Wilson F, Ingleton C, Gott M, Gardiner C. Autonomy and choice in palliative care: time for a new model? *Journal of Advanced Nursing*. 2014;70(5):1020-1029.
- 14. Green J, Britten N. Qualitative research and evidence based medicine. *BMJ*. 1998;316(7139):1230-1232.
- 15. Vaismoradi M, Turunen H, Bondas T. Content analysis and thematic analysis: Implications for conducting a qualitative descriptive study. *Nursing & Health Sciences*. 2013;15(3):398-405.
- 16. Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. *International Journal*

- for Quality in Health Care: Journal of the International Society for Quality in Health Care. Dec 2007;19(6):349-357.
- 17. Die Bundesversammlung der Schweizerischen Eidgenossenschaft. Humanforschungsgesetz 2014; <a href="https://www.admin.ch/opc/de/classified-compilation/20061313/index.html">https://www.admin.ch/opc/de/classified-compilation/20061313/index.html</a> Accessed 15 May 2017.
- 18. Graneheim UH, Lundman B. Qualitative content analysis in nursing research: concepts, procedures and measures to achieve trustworthiness. *Nurse Education Today*. Feb 2004;24(2):105-112.
- 19. Hardwig J. What about the family? *The Hastings Center report.* Mar-Apr 1990;20(2):5-10.
- 20. Pivodic L, Van den Block L, Pardon K, et al. Burden on family carers and carerelated financial strain at the end of life: a cross-national population-based study. *European Journal of Public Health.* Oct 2014;24(5):819-826.
- 21. Krug K, Miksch A, Peters-Klimm F, Engeser P, Szecsenyi J. Correlation between patient quality of life in palliative care and burden of their family caregivers: a prospective observational cohort study. *BMC palliative care*. Jan 15 2016;15:4.
- 22. Lee JE, Shin DW, Cho J, et al. Caregiver burden, patients' self-perceived burden, and preference for palliative care among cancer patients and caregivers. *Psycho-Oncology*. 2015;24(11):1545-1551.
- 23. Moneymaker K. The Family Conference. *Journal of Palliative Medicine*. 2005/02/01 2005;8(1):157-157.
- 24. Proot IM, Abu-Saad HH, Crebolder HFJM, Goldsteen M, Luker KA, Widdershoven GAM. Vulnerability of family caregivers in terminal palliative care at home; balancing between burden and capacity. *Scandinavian Journal of Caring Sciences*. 2003;17(2):113-121.

- 25. Payne S, Smith P, Dean S. Identifying the concerns of informal carers in palliative care. *Palliative Medicine*. 1999;13(1):37-44.
- 26. Given BA, Given CW, Kozachik S. Family Support in Advanced Cancer. *CA: A Cancer Journal for Clinicians*. 2001;51(4):213-231.

# **Chapter 6**

\_

# **Advance Care Planning in Swiss Acute Hospitals:**

# A Retrospective Analysis of Medical Records

Rakic M, Rost M, Wienand I, Elger BS, Escher M. Advance Care Planning in Acute Hospitals: A Retrospective Analysis of Medical Records.

**Abstract** 

Objective We wanted to describe how advance care planning occurs in major Swiss acute

hospitals that have established mobile palliative care teams. Moreover, we wanted to identify

different factors determining the application of advance care planning and advance directives.

**Methods** We collected data retrospectively from medical records of palliative care patients.

We analysed the data descriptively and employed inferential statistics (Chi-square test of

independence, generalized linear mixed model).

Results For most patients (98.3%) a discussion about advance directives, values and wishes

was documented in the medical records. Almost two thirds of the patients (62.6%) engaged in

advance care planning and around one quarter (26.3%) laid down their preferences in an

advance directive. Moreover, we found that engaging patients in advance care planning

resulted in higher odds of having an advance directive by a factor of 17.5.

**Conclusion** Knowing patients' preferences in advance is needed to base end-of-life decisions

on the patients' values and wishes. Our analysis of patients' medical records showed that

engaging patients in advance care planning increases the number of advance directives.

Patients should be supported to state their goals and preferences by involving them in

advance care planning.

Keywords: advance care planning, advance directives, palliative care, end of life,

quantitative research

119

# Introduction

In 2014 around 64,000 persons died in Switzerland and 26,900 of them (42%) were older than 85 years<sup>1</sup>. More than 41% of total deaths occurred in a hospital and the average age of patients who received Palliative Care (PC) was 68 years<sup>2</sup>. In a clinical setting, older patients often lack decisional capacity and half of the decisions are made by patients' surrogates<sup>3</sup>. Empirical data show that one third of patients with an acute illness aged  $\geq 75$  years suffered from a delirium. Delirium is an acute and confusional state, linked to an increased mortality rate<sup>4</sup>. Furthermore, the incidence of dementia increases with age<sup>5</sup>. These statistics reflect the importance to provide advance care planning (ACP) as the elderly population increases<sup>6</sup> and many may not be able to decide for themselves<sup>3</sup>. By promoting ACP early in the course of disease, patients' autonomy can be respected, even if patients become unable to decide for themselves what kind of treatments they want or not<sup>7</sup>.

PC is a holistic approach that centres on quality of life of patients and their families<sup>8</sup>. End-of-life decisions can be burdensome, especially for incapacitated patients when family members do not know their wishes<sup>9</sup>. ACP is recommended to support decision-making in these situations, along with the writing of legally binding advance directives (ADs)<sup>10,11</sup>. By enabling patients to determine their medical treatments in advance and make their preferences known, ACP implements the ethical principle of respect for patients' autonomy<sup>7</sup>. Previous data show that ACP can help respect patients' wishes at the end of life<sup>12</sup> and that physicians comply with patients' advance directives<sup>13</sup>. Furthermore, ACP not only increased patients' but also families' satisfaction with PC (e.g. satisfaction with dying process, decreased level of stress and anxiety) during the last hospital stays<sup>12</sup>.

ADs are considered an important component of ACP<sup>14</sup>. Empirical evidence showed that the most important reason for patients to write ADs was promotion of their autonomy<sup>15</sup>.

Similarly, the Mental Capacity Act in the UK and the Adult Protection Law in Switzerland enable and encourage to write ADs. Moreover, adults can appoint a surrogate decision-maker<sup>16,17</sup>.

However, in Switzerland only 11% of PC patients had written an AD prior to hospital admission<sup>18</sup>. Rao et al. reported that the main reason for not having an AD was a lack of awareness among patients<sup>19</sup>. Studies focusing on determinants for completing an AD indicated that patients who died at home or in a retirement home were more likely to have an AD than patients who died at hospital<sup>20</sup>.

Switzerland is facing an increase in deaths occurring in hospitals with complex end-of-life situations<sup>2,3</sup>. Knowing patients' preferences is needed to align end-of-life-care with their values<sup>15</sup>, and to alleviate the family's burden in decision-making<sup>12</sup>. Therefore, the main aim of this study is to find out how clinical practice takes place in a PC setting of an acute care hospital. More precisely, we want to a) describe how ACP occurs in major Swiss acute hospitals that have established mobile PC teams and, b) identify different factors determining the application of ACP and/or the completion of an AD in Switzerland. By employing a retrospective approach focusing on PC inpatients cared for by mobile PC teams, we captured variables likely to influence patients' self-determination, such as socio-demographics and illness.

## Methods

## Study design

We collected data from three Swiss university hospitals which have hospital-wide palliative care consultations ("mobile specialised team"). We reviewed the medical records of the patients seen by the PC teams. Ethical approval was obtained from the responsible ethics committees in Switzerland (EKNZ; Nr. EK 2015-197).

## **Study population**

At the time when a patient's medical situation was seen as a palliative one, the treating physicians contacted the hospital's mobile PC team in order to discuss the possible PC approaches (e.g. pain management, spiritual or psychosocial support) with the patient. PC patients who met the following criteria were included: (a)  $\geq$  18 years, (b) having received PC treatment in one of the three participating hospitals. Eligible patients were identified from the consultation lists established by the collaborating physicians. We exclusively searched the electronic files of those patients. Patients who had more than one hospital stay during the inclusion period were included only once.

### **Data extraction sheet**

The data extraction sheet included patients' characteristics (e.g. gender, nationality, age), palliative care related variables (e.g. diagnosis, treatments), and variables related to patients' autonomy (e.g. ACP, AD). Information on burden in PC are discussed in another paper that used the qualitative data of the medical records [blinded for peer review]. The notes in the patients' files were written by all the health-care team. The digital extraction sheet consisted of items with categorical responses (e.g. gender), continuous variables (e.g. age), and openended items (e.g. notes about patients' wishes and preferences). All items were developed from the interdisciplinary research team's knowledge in the field and based on discussions with the collaborating physicians.

### **Data collection**

Data collection was carried out between April and September 2016. We started collecting the data of the first patient who was referred to a palliative care team on 1<sup>st</sup> January 2016 and we then continuously went back to the year 2015 until we had included 100 patients at each hospital. The medical records were available in digitalized form. We collected data throughout the medical records. This included the cover page, PC reports, and discharge

reports. We searched the notes written by all the health-care professionals (e.g. physicians, nurses, psychologists). We included all the information written between the first PC consultation and the day the patient either left the hospital or died.

Data were entered in the previously developed extraction sheet. The research team discussed the first five extractions aiming for standardization of extracting and continuously discussed the extractions, when needed. Since not every variable of the extraction form was contained in all medical records or could be found, sample sizes for most of the variables were less than the number of included patients (N = 300). Patients' anonymity was ensured by using a predefined algorithm employing numbers unrelated to birth date or other personal identifiers. Furthermore, all identifying information was deleted after data collection.

### Statistical analysis

A research assistant entered all extracted records into SPSS.22 (SPSS Inc, Chicago, IL) and a second researcher checked for correctness of data entry. Statistical analyses were performed using SPSS.22. In accordance with the aim of this study, variables related to PC (e.g. diagnosis, place of care, several PC treatments) and patients' autonomy (e.g. ACP, AD, discussion about wishes and values) were analysed. Regarding ADs and ACP: the research team searched in the medical records for documented discussions on ACP between the mobile PC team and the patient (PC treatment, in particular resuscitation, artificial nutrition; patient's wishes; values; preferences of family members and of the patient's general practitioner, nurses and other relevant care givers). Moreover, the research team examined the medical records for any documented information on patients' ADs (long or short form, if the patient received advice during hospitalisation, and if the patient had preferences for or against specific treatment). In a first step, variables were analysed descriptively. In a second step, we employed inferential statistics, namely Chi-square test of independence and Generalized Linear Mixed Model (GLMM). Reported P values were 2-sided and statistical

significance level was set at P < .05. In cases of multiple testing Bonferroni-Holm correction was applied to control for the increased likelihood of a type I error. The GLMM adjusted for centre clustering.

# **Results**

# Characteristics of the sample

Fifty-two percent of the patients were male (Table 1). Mean age was 67.6 years, ranging from 22 to 98 years. Seventy-three percent of the patients were diagnosed with cancer. Among the non-cancer patients, nearly one fifth had a neurological disease (19.74%).

Table 1: Sample characteristics (N = 300)					
Demographics and Diagnosis	Mean (Mdn, SD) Percentage				
Age $(n = 299)$	67.6 (69.0, 15.0)				
Sex (male; $n = 299$ )	52.2 %				
Diagnosis (cancer; $n = 287$ )	73.5 %				
Nationality (Swiss; n = 298)	77.2%				
Marital status (married; $n = 242$ )	72.3%				
Living place (urban, $n = 204$ )	68.5%				
Children (yes; n = 198)	72.7%				
Retirement (yes; $n = 300$ )	48.0%				
Religious affiliation (yes; n = 295)	91.9%				

For most patients (98.3%, 238 out of 242) a discussion about ADs, values and wishes was documented in the records. Almost two thirds of the patients (62.6%, 154 out of 246) engaged in ACP and around one quarter (26.3%, 52 out of 198) laid down their preferences in an AD (see figure 1). Eighteen percent of the patients (39 out of 215) had a Do Not Resuscitate (DNR) order.

## Factors influencing advance care planning

## Relation between ACP and AD

Almost every patient had a discussion about his or her values and wishes regarding ACP (98.3%). Moreover, sixty-three percent of the PC patients engaged in ACP (154 out of 246). Exploring whether ACP actually predicted whether or not a patient had an AD, GLMM revealed that patients who engaged in ACP more often had an AD than those patients who did not engage in ACP (Figure 1; Table 2). To illustrate, in our sample engaging in ACP resulted in higher odds of having an AD by a factor of 17.5 (p = .000)

Table 2: GLMM of advance directives (n = 172)									
	В	SE	t	P	P	P	Odds	95% CI for Odds Ratio	
	ъ	SE	ι				Ratio	Lower	Upper
Intercept	-3.004	0.938	-3.202	.002					
ACP	2.863	.055	5.670	.000	17.506	6.437	47.613		
Age	0.002	.013	0.115	.909	1.002	0.975	1.028		
Gender	.083	.406	.205	.838	1.087	.486	2.429		

*Note.* The dependent variable in this analysis is "advance directive" so that 0 = no and 1 = yes. Results were adjusted for centre clustering.

In the following, further variables that might influence either advance care planning or having an advance directive are tested.

### Diagnosis

Forty-three percent of non-cancer patients (23 out 54) and twenty-one percent of cancer patients (29 out of 144) had an AD. Results of a Chi-square test of independence indicated that the relation between diagnosis (cancer versus non-cancer) and whether or not a PC patient had an AD was significant, that is, non-cancer patients more often than cancer patients had an AD,  $X^2$  (1, N = 198) = 10.2, p = .001, V = .23 (see figure 1).

Yes **ACP** 37.4% 62.6% AD 50.0% 5.4% with AD with AD Overall 26.3% 20.1% 42.6% with AD with AD Cancer - Cancer Diagnosis 73.7% 26.3%

Figure 1. Advance care planning and advance directives

### Partner

A Chi-square test of independence revealed no significant relation between whether or not a patient had a partner, and whether or not a patient had an AD, X2 (1, N = 191) = .082, p = .863, V = .02. Similarly, there was no significant relation between whether or not a patient had a partner, and whether or not a patient engaged in ACP, X2 (1, N = 195) = .087, p = .769, V = .02.

### Children

The relation between whether or not a patient had an AD and whether or not the patient had children was tested. Chi-square test of independence revealed no significant relation,  $X^2$  (1, N = 144) = 2.5, p = .113, V = .13. Thirty-eight percent (14 out of 37) of patients without children had an AD and only twenty-four percent (26 out of 107) of patients who had at least one child, had an AD. The same pattern was observed for the relation between whether or not a patient had engaged in ACP and whether or not the patient had children. Again, Chi-square test of independence did not reveal a significant relation,  $X^2$  (1, N = 168) = 2.5, p = .110, V = .12. However, descriptively differences existed, with sixty-eight percent (32 out of 47) of

those childless patients had engaged in ACP and only fifty-five percent (66 out of 121) of those patients who had at least one child.

Place of care

Chi-Square tests for independence (Bonferroni-Holm correction was applied) neither revealed a significant relation between whether or not a patient engaged in ACP and the place of care,  $X^2$  (1, N = 239) = 3.0, p = .086, V = .11, nor between whether or not a patient had an AD and the place of care,  $X^2$  (1, X = 184) = .09, X = .768, X = .02.

# **Discussion**

In our study, we analyzed data about wishes and values of PC patients hospitalized in an acute care setting. Almost every patient had a discussion about his or her wishes and values and six out of ten engaged in ACP. In addition, we focused on patients who were seen by hospital-wide mobile palliative care teams. Hence, we do not know if the proportion of discussions among other patients – who were also diagnosed with an advanced disease but not referred to a specialized palliative care team – is similar. There are studies from other countries that compared PC units and mobile PC services within hospitals regarding economical and patient related outcomes<sup>21,22</sup>. Our results revealed that around three quarters of patients for whom hospital units contacted the PC team were diagnosed with cancer and mean age was approximately 70 years. Similar findings were reported in studies that focused on specialized PC institutions or mobile PC teams<sup>2,21,23,24</sup>.

# Discussions about advance care planning and advance directives

A study from the United States showed that the incidence rate for patients requesting ACP was 39%<sup>25</sup>. In our study, 63% of patients engaged in ACP. An intervention study showed that for patients who actively obtained ACP, wishes were more often known and respected (86%), as compared to those receiving only standard care (30%)<sup>12</sup>. For our sample, results of the

GLMM showed that those patients who engaged in ACP more often had an AD. At the same time – even though ACP entails discussions about ADs<sup>12</sup> – only every second patient who engaged in ACP had an AD.

Moreover, since we analyzed medical records, it was not always possible for us to retrospectively determine whether patients had an AD prior to ACP or if they filled out an AD after they engaged in ACP. This gives rise to two different interpretations, namely, first, that a high number of patients who engaged in ACP finally completed ADs and, second, that health-care professionals more easily engage patients in ACP when patients already had an AD. However, based on the results of the GLMM, we can say that the first interpretation is more accurate: ACP increases the number of patients with ADs. As previously shown, ADs can gain importance once patients are unable to articulate their wishes and preferences regarding medical treatments<sup>7,14</sup>. Therefore, it is important to provide every patient with the opportunity to engage in ACP and to make an informed decision on filling out an AD. Ultimately, this process promotes autonomous decision-making of patients<sup>7</sup>.

## **Advance directives**

Overall, 26.3% of the PC patients in our sample had an AD. However, a study from Switzerland from 2015 reported a lower percentage of PC patients regarding existence of ADs prior to hospital admission, namely 11%. Nevertheless, it needs to be said that this number slightly increased during hospitalization<sup>18</sup>. The higher percentage in our study represents an encouraging finding as it suggests that the number of patients with ADs increases during hospitalization and has increased over the past years. There are several explanations for this. First, the increase could be caused by a raised public awareness in the last years through the launch of the national strategy aiming to promote PC<sup>26</sup>. Another explanation for the increase in ADs could be the new Adult Protection Law (since 2013) in Switzerland. The aim of the new law is to promote patients' right for self-determination and

to enable and encourage competent adults to write ADs<sup>16</sup>. Finally, the increase could be attributed to the fact that, in our sample, a comparably high number of patients engaged in ACP (compared to the Swiss study from 2015, where only 11% of the patients had an AD prior to hospitalization<sup>18</sup>) which resulted in a higher number of patients with ADs.

## Children

Although the relation between having children and the presence of an AD or ACP was statistically not significant, descriptively childless patients more frequently engaged in ACP and had ADs. Subject to further research exploring this relation, such a finding might be explained by a greater certainty among patients with children that, in case of becoming unable to consent to medical treatments, children can and will decide on their behalf. Confidently assuming that your child(ren) will make decisions on your behalf diminishes the necessity to think about ACP and to sign an AD. However, it is important to note that these differences were not significant and that this relation, therefore, need further careful examination in order to draw robust conclusions.

## Limitations

First, medical notes are written by health-care professionals and they might have different ways of writing and reporting. Therefore, medical notes are not exhaustive, and they may not always render verbatim the exact dialog which took place between the patient and health-care professionals. Second, the analysis of medical records is necessarily limited to both, the written information, but also to the quality of this information found in the records. Although the information in the medical records might not be complete, it is this information the health-care professionals have at their disposal when they have to discuss with the family and/or make a decision.

# Conclusion

Our analysis revealed that almost every patient had discussions about values and wishes and almost two out of three patients engaged in ACP. Moreover, 26.3% of the patients in our sample had an AD. Knowing patients' wishes is needed to base the decisions on patients' values<sup>15</sup>, and to alleviate the family's burden in the decision-making process<sup>12</sup>. Therefore, PC patients should be supported to state their goals and preferences by engaging them in ACP and, eventually, filling out ADs.

# **Conflict of interest**

The authors have no conflict of interest in the authoring of this manuscript.

# References

- Bundesamt für Statistik (BFS). Sterblichkeit und deren Hauptursachen in der Schweiz.
   https://www.bfs.admin.ch/bfs/de/home/statistiken/gesundheit/gesundheitszustand/ster blichkeit-todesursachen/spezifische.assetdetail.2022361.html Published February 21, 2017. Accessed November 17, 2017.
- 2. Bundesamt für Gesundheit und Schweizerische Konferenz der kantonalen Gesundheitsdirektorinnen und -direktoren (BAG) und (GDK). Nationale Strategie Palliative Care 2013–2015. 2012; <a href="http://www.gdkcds.ch/fileadmin/docs/public/gdk/themen/strategie\_bundkantone/palliativecare/bt\_nationale\_strategie\_palliative\_care\_2013-2015\_d.pdf">http://www.gdkcds.ch/fileadmin/docs/public/gdk/themen/strategie\_bundkantone/palliativecare/bt\_nationale\_strategie\_palliative\_care\_2013-2015\_d.pdf</a> Accessed 23.06.2017.
- 3. Torke AM, Sachs GA, Helft PR, et al. Scope and outcomes of surrogate decision making among hospitalized older adults. *JAMA Internal Medicine*. Mar 2014;174(3):370-377.
- 4. Pendlebury ST, Lovett NG, Smith SC, et al. Observational, longitudinal study of delirium in consecutive unselected acute medical admissions: age-specific rates and associated factors, mortality and re-admission. *BMJ open*. Nov 16 2015;5(11):e007808.
- 5. Corrada MM, Brookmeyer R, Paganini-Hill A, Berlau D, Kawas CH. Dementia Incidence Continues to Increase with Age in the Oldest Old The 90+ Study. *Annals of Neurology*. 2010;67(1):114-121.
- 6. Bundesamt für Gesundheit (BAG) und Schweizerische Konferenz der kantonalen Gesundheitsdirektorinnen und -direktoren (GDK). Nationale Strategie Palliative Care 2013–2015. 2012; https://www.bag.admin.ch/dam/bag/de/dokumente/nat-

- gesundheitsstrategien/strategie-palliative-care/nationalestrategie.pdf.download.pdf/07 D Nationale Strategie Palliative Care 20132015.pdf Published October, 2012. Accessed July 22, 2017. Accessed 05.12.2017.
- 7. Singer PA, Robertson G, Roy DJ. Bioethics for clinicians: 6. Advance care planning.

  Canadian Medical Association Journal. December 15, 1996 1996;155(12):16891692.
- 8. World Health Organization. WHO Definition of Palliative Care. 2002; <a href="http://www.who.int/cancer/palliative/definition/en/">http://www.who.int/cancer/palliative/definition/en/</a> Published 2002. Accessed May 11, 2017.
- 9. Yamamoto S, Arao H, Masutani E, et al. Decision Making Regarding the Place of End-of-Life Cancer Care: The Burden on Bereaved Families and Related Factors.

  \*\*Journal of Pain and Symptom Management. 5// 2017;53(5):862-870.
- 10. SAMW. Patientenverfügungen. 2013; <a href="https://www.rwi.uzh.ch/dam/jcr:f7c5b696-6fab-4e8a-9004-18800a87c0e7/SAMW-RL%20Patientenverf%C3%BCgung.pdf">https://www.rwi.uzh.ch/dam/jcr:f7c5b696-6fab-4e8a-9004-18800a87c0e7/SAMW-RL%20Patientenverf%C3%BCgung.pdf</a>
  Published May 19, 2009. Updated January 1, 2013. Accessed May 08, 2017.
- 11. SAMW. Palliative Care. 2017; <a href="www.samw.ch/dam/jcr:d7ae1138-0213-481b.../richtlinien\_samw\_palliative\_care.pdf">www.samw.ch/dam/jcr:d7ae1138-0213-481b.../richtlinien\_samw\_palliative\_care.pdf</a> Published May 23, 2006. Updated January 1, 2013. Accessed May 08, 2017.
- 12. Detering KM, Hancock AD, Reade MC, Silvester W. The impact of advance care planning on end of life care in elderly patients: randomised controlled trial. *BMJ*. 2010;340.
- 13. Escher M, Perneger TV, Rudaz S, Dayer P, Perrier A. Impact of advance directives and a health care proxy on doctors' decisions: a randomized trial. *J Pain Symptom Manage*. Jan 2014;47(1):1-11.

- 14. Gillick MR. Advance Care Planning. New England Journal of Medicine. 2004;350(1):7-8.
- 15. Pautex S, Notaridis G, Déramé L, Zulian GB. Preferences of elderly cancer patients in their advance directives. *Critical Reviews in Oncology/Hematology*. 2010/04/01/2010;74(1):61-65.
- 16. Medienmitteilungen E. Neues Erwachsenenschutzrecht tritt am 1. Januar 2013 in Kraft. 2011; <a href="https://www.ejpd.admin.ch/ejpd/de/home/aktuell/news/2011/2011-01-12.html">https://www.ejpd.admin.ch/ejpd/de/home/aktuell/news/2011/2011-01-12.html</a> Published January 1, 2011. Accessed June 5, 2017.
- 17. Mental Capacity Act. Advance decisions to refuse treatment: general. 2005; <a href="http://www.legislation.gov.uk/ukpga/2005/9/section/24">http://www.legislation.gov.uk/ukpga/2005/9/section/24</a> Published 2005. Accessed August 11, 2017.
- 18. Pautex S, Gamondi C, Philippin Y, et al. Advance directives and end-of-life decisions in Switzerland: role of patients, relatives and health professionals. *BMJ Supportive & amp; Palliative Care.* 2015.
- 19. Rao JK, Anderson LA, Lin FC, Laux JP. Completion of advance directives among U.S. consumers. *American journal of preventive medicine*. Jan 2014;46(1):65-70.
- 20. Teno JM, Gruneir A, Schwartz Z, Nanda A, Wetle T. Association Between Advance Directives and Quality of End-of-Life Care: A National Study. *Journal of the American Geriatrics Society*. 2007;55(2):189-194.
- 21. Gaertner J, Frechen S, Sladek M, Ostgathe C, Voltz R. Palliative Care Consultation Service and Palliative Care Unit: Why Do We Need Both? *The Oncologist*. 02/2109/22/received11/26/accepted 2012;17(3):428-435.
- 22. Morrison R, Penrod JD, Cassel J, et al. Cost savings associated with us hospital palliative care consultation programs. *Archives of Internal Medicine*. 2008;168(16):1783-1790.

- 23. Stiel S, Heckel M, Seifert A, Frauendorf T, Hanke RM, Ostgathe C. Comparison of terminally ill cancer- vs. non-cancer patients in specialized palliative home care in Germany a single service analysis. *BMC palliative care*. Jul 25 2015;14:34.
- 24. Bennett MI, Ziegler L, Allsop M, Daniel S, Hurlow A. What determines duration of palliative care before death for patients with advanced disease? A retrospective cohort study of community and hospital palliative care provision in a large UK city. *BMJ open.* 12/0905/09/received11/09/revised11/10/accepted 2016;6(12):e012576.
- 25. Garand L, Dew MA, Lingler JH, DeKosky ST. Incidence and Predictors of Advance Care Planning Among Persons with Cognitive Impairment. *The American journal of geriatric psychiatry : official journal of the American Association for Geriatric Psychiatry*. 2011;19(8):712-720.
- 26. Judith Binder LvW. Nationale Leitlinien Palliative Care, . 2010;
  <a href="http://www.pallnetz.ch/cm\_data/Nationale\_Leitlinien\_Palliative-Care.pdf">http://www.pallnetz.ch/cm\_data/Nationale\_Leitlinien\_Palliative-Care.pdf</a> Published November, 2010. Accessed May 08, 2017.

# **Chapter 7 – General Discussion**

This PhD thesis aims to integrate empirical and normative analyses in order to promote respect for elderly patients' autonomy in various *clinical settings*. More specifically, the purpose is to explore and evaluate possible *hindering and promoting factors* of autonomous decision-making in research and Palliative Care (PC) contexts (section 1 and 2). The empirical results were obtained from i) in-depth interviews regarding clinical trial (CT) participation involving synthetic biology (SB) with 36 patients suffering either from diabetes or gout, and ii) 300 medical records of PC patients regarding quality of PC. These results are discussed from a bioethical perspective. The final sections of this chapter present the *research limitations, implications for further research*, and *conclusion* (section 4, 5, and 6).

# Hindering and promoting factors in a research context

Clinical trial participation

Chapter 2 examines patients' willingness to participate in CT's involving SB and evaluates the extent to which their views and preferences change after receiving detailed information about SB. Our results indicate that patients are anxious and lack understanding of SB when discussing possible CT participation. Our results revealed that the most prevalent concern of patients is a general anxiety that SB creates something that cannot be controlled (one interviewed patient was afraid that "somehow a superhuman gets created"). Nevertheless, after a concrete example was given on how a future medical treatment involving SB could look, patients began to see SB in a more positive light:

"(...) Mr. Müller suffers, as you, from gout, but he is not taking any drugs. He needs to go rarely to his GP to get his uric acid level controlled. Mr. Müller has a capsule implanted under his skin. This capsule contains human cells. These cells are from Mr. Müller himself. They were removed and modified in the laboratory. These modified

cells from Mr. Müller are able to produce a certain protein that removes the surplus uric acid in the blood. Additionally, a sensor was integrated that constantly measures the uric acid levels in the blood (...)." (See Chapter 2, p. 49)

Therefore, it is important to educate patients and give them realistic expectations about the synthetic biology CT's. Achievement of each of these recommendations would promote patients' autonomy in decision-making. Although providing adequate information is not novel<sup>1</sup>, we think that providing information on something relatively common (e.g. cancer and treatment) is different from providing information on SB, which is often foreign to or misconceived by patients. Most interviewed patients did not know the purpose of the new technology and viewed it as a risky procedure (one interviewed patient thought that the researchers are "playing God"). Patients appeared to have difficulty understanding the purpose of medical research given that the future applications are still hypothetical. The patients' incomplete and inaccurate picture of SB could lead to therapeutic misconception by which patients believe that enrolling into the CT will treat their disease<sup>2</sup> and therefore, might prompt them to make decisions based on mistaken beliefs. This lack of accurate information contradicts the requirements of an informed consent (IC), which requires (among other things) an understanding of the purpose of the research<sup>3</sup>.

Patients who do not understand the purpose of medical research (i.e. difference between research and treatment<sup>2</sup>) or lack information regarding the main aims (e.g. one patient noticed that "[SB] is a closed book") might be misled in their decision-making on CT participation.

To reduce this bias, the provision of information must be as accurate as possible and avoid misconceptions. If clear information about SB is not delivered, then patient autonomy is undermined, as the principle of autonomy relies on informed patient understanding of the objectives and potential consequences of the CT<sup>3</sup>. Furthermore, IC has little value if a

patient's decision whether or not to participate in a CT is mainly based on emotions (e.g. one patient feared that "something that cannot be controlled anymore" will be created) evoked by an inaccurate understanding of the CT. Because SB is not widely understood by patients, providing sufficient knowledge for IC is of particular relevance for these CT's.

To address the challenge of misunderstanding the aim of SB, the use of simple language or graphics instead of difficult medical jargon<sup>4</sup> would enhance patients' understanding. Our study found that patients viewed SB and CT participation more positively after they were given a concrete example of a hypothetical patient suffering from diabetes/gout and how SB might be able to help that patient (see example, p. 49). Given that example, most patients were able to distinguish SB from other technologies (e.g. genetic engineering) and expressed a better understanding of SB and the possible benefits to their own health.

### Research ethics committees

Another possible approach to overcome the obstacle of patient misunderstanding in SB CT's is to have research ethics committees (RECs) examine very carefully the content of ICs to ensure all information presented is detailed, accurate, and understandable, as information is one of the key elements of ICs<sup>5</sup>. Cassileth et al. stress that patients have difficulties to understand ICs (e.g. the purpose or possible complications related to medical procedures)<sup>6</sup>. To overcome these challenges discussed by Cassileth et al., IC information could be reviewed and tailored by patients who are actively involved in this process. This could make the information more easily accessible for possible CT participants. Results from our study, which aims to explore how patients perceive their own participation as active members of RECs (Chapter 3), indicate a positive attitude towards participating in RECs.

Patients in our study often mentioned benefits to their own health as a reason for their participation in RECs. More precisely, patients saw a chance for a long-term benefit on their own health condition from participating in RECs and perhaps allowing more CTs. In other words, if these CTs show a positive effect, patients could profit from an optimization for their own treatment over time<sup>7</sup>.

Even though these patients suffer from chronic diseases with a decreased life expectancy and lower quality of life<sup>8,9</sup>, they are not imminently near death<sup>7</sup>. Therefore, they have a realistic opportunity to profit from medical developments<sup>10</sup> in the near future. Thus, the risk of therapeutic misconception<sup>2</sup> does not seem to be applicable in this situation, since this is not their last hope<sup>7</sup>. Another reason mentioned by patients was that they wanted their situation to be visible (see patient's quote p. 61). Therefore, not involving patients in RECs might be seen as unjustified paternalism<sup>11</sup>.

"To be able to see the patients' situation. He/she can, only he/she can, say how hard the disability is, caused by the disease. He/she can estimate the agony caused by the disease (...)." (See Chapter 3, p. 61)

Patients mentioned in our study that they wanted to express their experiences of the disease, acquired knowledge about the disease, and their unique perspectives to be taken into consideration within RECs. Our results align with other studies, which indicate that patients' perspectives on their disease, use of language, and experiences of their diseases differ from those of health-care professionals<sup>12</sup>. Thus, health-care professionals should take patients' lifeworld into account to improve the comprehensibility of written medical information (e.g. by using graphical symbols)<sup>12</sup>. Lifeworld is understood here as patients' individual

experiences and problems in their everyday lives<sup>13,14</sup>. Moreover, Barry et al. point out possible consequences, if patients' lifeworld is not taken into account by physicians:

"Here the patients talked either exclusively or for a large amount of the consultation in the voice of the lifeworld. However, the doctors completely ignored this and conducted the whole of their communication in the voice of medicine. Most of these patients had chronic physical problems and this group had the worst outcomes of any group [...] Patients and doctors seemed to be operating at odds to each other. Patients often seemed relaxed and happy to chat away in the voice of the lifeworld about their concerns while doctors stayed rigidly inside the biomedical format." 13, p. 494-495

Miscommunication between patient and physician can result in a poor outcome if health-care professionals are not sensitive to the patient's experiences and narratives<sup>13,14</sup>. Therefore, health-care professionals should take patients' illness narratives into account during medical discussions to integrate more fully the first-person perspective of patients. Involving stable patients actively in RECs and letting them participate in decision-making processes could enhance their autonomy and ensure that ICs are more accurately and understandable for prospective CT participants.

#### Норе

As discussed in Chapter 4, we explore the importance of hope as a determining factor for patient participation in CTs for SB therapies. Our results suggest that hope can act as a 1) coping strategy in the face of either anxiety or 2) uncertainty, or as 3) self-therapy.

Health-care professionals might be hesitant to provide full information (e.g. about a patient's condition) out of fear of taking away the patient's hope 15. However, hope does not mean exaggerating benefits for future patients, nor does it entail reinforcing therapeutic misconceptions in patients and hiding information from them 15. Based on our findings, health-care professionals should be more aware of the resources of hope as it seems to be beneficial to patients' well-being. Moreover, these three kinds of hope can coexist with fully informing patients and thereby respecting their autonomy. Our results are supported by other studies. For example, Milna et al. concluded that informing patients about the side effects of their disease and helping them find strategies to manage these side effects helped patients acquire specific knowledge about their disease. This acquired knowledge relieved them to restore their hope and become more autonomous regarding their own care 16. At the same time, we have shown that hope can also imply the awareness of uncertainty regarding the chance of convalescence, and this aspect of hope does not sink us into despair.

#### Hindering and promoting factors in a Palliative Care context

Advance care planning and advance directives

Patients are encouraged to record their preferences – e.g. through advance care planning (ACP) or advance directives (ADs) – and determine their medical treatments in advance <sup>17,18</sup>. By encouraging patients to do so, the bioethical principle of respecting patients' autonomy is implemented <sup>19</sup>. Moreover, knowing patients' preferences is a prerequisite to align care with patients' values <sup>20</sup>. Chapter 6 discusses how ACP/ADs are approached in a PC setting in Switzerland using data from medical records. Our results indicate that almost every patient discussed his or her values and wishes regarding ACP with a health-care professional. Furthermore, patients who engaged in ACP more often had an AD than those patients who did not engage in ACP. Therefore, patients should be supported to state their goals and preferences by involving them in ACP<sup>21</sup>. This becomes even more important for patients with an expected decline in mental status because these patients might not be able to decide for themselves anymore<sup>22</sup>, creating a large burden on the family to make decisions in their place<sup>23</sup>.

#### Burden

Respecting patients' autonomy and including informal caregivers (e.g. family members) in the care plan are central needs of PC patients<sup>24</sup>. Nevertheless, patients fear that they might be a burden to their informal caregivers<sup>24</sup>, and conversely, informal caregivers feel burdened by providing care for patients<sup>25</sup>. In Chapter 5, we analyse what was reported about burden in PC patients' medical records. Our results reveal that patients felt often burdened by their own disease, their family being constantly around them, and the situation at home. Families felt burdened by issues related to patients' medical condition, home care, and decision-making. These common situations illustrate that the individualistic concept of autonomy is challenged

when the two parties have different goals, interests, resources, and needs. For example, some patients want to spend their last days at home, but their relatives cannot provide the desired homecare as it is too burdensome for them<sup>26</sup>. The apparent incompatibility of views might be minimised under a less individualistic framework of autonomy. Indeed, the standard understanding of autonomy as self-determination in Western modern society<sup>27</sup> might neglect important cultural, social, and practical aspects that undergird patients' sense of autonomy<sup>26,28</sup>. Walter et al. conclude that the individualistic understanding of autonomy puts an emphasis on health-care professionals as the sole information providers for patients that base their decisions only on rationality. Furthermore, emotions are seen as counter-productive to this rationalistic understanding of autonomy because they undermine the rational decision-maker<sup>28</sup>.

#### Relational autonomy

Relational autonomy can be seen as a promising alternative to the individualistic understanding of autonomy because it emphasizes the significance of the social network in our lives<sup>28,29</sup>. The claim behind relational autonomy is that human beings are embedded in a social context. Identities are significantly impacted by social and emotional relationships to other persons and are shaped by social determinants (e.g. education, gender, language). The main focus of relational autonomy is to underline the impact of these social relationships and determinants on ourselves and our decision-making and reciprocally, our influence on other members of the social network<sup>29</sup>. Within the framework of relational autonomy, asking family members to participate in the decision-making process of patients does not necessarily undermine patients' autonomy, as patients' preferences are embedded within a larger network of relationships with family members and others<sup>28</sup>. Within a PC context, patients' autonomy can be enhanced if the family supports and recognizes their decisions<sup>30</sup>. Furthermore, Lavoie

et al. explain that PC patients describe autonomy as something that not only has to do with the individual, but also involves a relationship with others<sup>31</sup>. This aspect of patient, as well as family centred care is also emphasized by the WHO definition of PC<sup>32</sup>.

Families too need to be assisted in dealing with their own burden and needs, in order to be able to support and respect patients' decisions. Thus identifying and reducing hindering factors, such as burden or overburden of patients and their families, could foster patients' autonomous decision-making, as pointed out in the literature<sup>30</sup>. Therefore, we suggest that health-care professionals should be more sensitive to a relational understanding of autonomy because this would facilitate a more holistic understanding of patients' and families' needs<sup>32</sup>.

.

### **3** Overview

### **Hindering Factors**

- Lack of understanding and knowledge
- Inadequately provided information
- Irrational fears and emotions
- Burden and overburden
- Situation of dependency
- Lack of mental capacity
- Therapeutic misconception
- Individualistic concept of autonomy

### **Promoting Factors**

- Concrete examples of research purpose
- Adequately provided information
- Tailoring informed consent to target patient population
- Taking patients' lifeworld into account
- Enabling active participation in decision-making
- Advance directives and advance care planning
- Hope as coping strategy
- Relational autonomy

Elderly patients' autonomy

### **4 Research Limitations**

Limitations that apply to each paper included in this PhD thesis are discussed in the corresponding Chapters (2-6). The thesis merged theoretical and normative concepts (autonomy) and empirical findings (promoting and hindering factors of autonomy in a clinical setting). It has been repeatedly argued that ethics focuses too heavily on normative claims rather than reaching out for the real-life experience<sup>33</sup>.

"While ethicists contribute normative judgments relying on secondary, distant information or vignettes made available to them by others, normally clinical staff are completely involved in the phenomena described and provide, thus, first hand statements in the mode of self-report and subjective, personal experience." 33, p. 27

In order to reduce the gap between prescribing what should be (normative level) and describing what is (empirical level), medical ethicists should use empirical methods to apply their normative concepts to clinical challenges<sup>33</sup>. Without committing a naturalistic fallacy<sup>33</sup>, we believe it is possible to bridge these two approaches, and moreover, that they can mutually enrich each other. The discipline of bioethics is in this regard very useful insofar as it attempts to bridge the empirical and normative approach, thereby acknowledging the importance of empirical research to mirror the reality (e.g. patients in a clinical setting and their autonomy). Bioethics also emphasizes the importance of a normative approach to identify the ethical issues at stake and find solutions based on a real-life approach<sup>33-35</sup>.

Limitations of this PhD thesis include the partially qualitative nature and thus, its lower degree of generalizability. The quantitative part of the thesis focused exclusively on Switzerland and therefore, it is unclear to what extent these results are applicable to other countries. Nevertheless, our findings provide interesting and important insights on how elderly patients' autonomy can be promoted in a clinical setting.

### 5 Implications for further research

Respect for elderly patients' autonomy remains important throughout the illness course and especially during the end of life. First, more qualitative research (e.g. interviews with patients) is needed to understand more precisely what patients understand under the term "autonomy". Second, our initial qualitative findings could be expanded through additional quantitative research to make more generalizable statements regarding patients' willingness to participate in CTs involving novel technologies or being part on RECs.

#### **6 Conclusion**

The ethical imperative of respecting patients' autonomy is not new. Nevertheless, in the light of an increase in chronic diseases, new medical developments (e.g. SB), and complex end-of-life situations (e.g. PC), it remains of utmost importance to examine what might hinder patients' autonomy in these situations and what can be done to promote it. This PhD thesis aimed to discover hindering and promoting factors of patients' autonomy throughout their course of disease.

Moreover, our results indicated that there is not a single concept of autonomy valid for everyone at any time. These findings point to the culturally determined nature of autonomy and serve as a reminder that health-care professionals should be aware that the perception of what a patient understands under autonomy might shift throughout the disease and patients' life narrative. Therefore, communication among the different parties involved in decision-making (e.g. patient, physician, family member) should be consistently and rigorously

pursued. Through frequent discussions (e.g. family conferences), a shared understanding of what autonomy means for each individual can be built. Health-care professionals should not focus solely on the individualistic definition of autonomy, but also take into account patients' network of relations and their lifeworld to best promote their autonomous decision-making. Relational autonomy and inclusion of patient's lifeworld into the clinical setting can act as promoting factors to their autonomy.

### References

- Peters E, Hart PS, Fraenkel L. Informing Patients: The Influence of Numeracy, Framing, and Format of Side Effect Information on Risk Perceptions. *Medical Decision Making*. 2011;31(3):432-436.
- 2. Appelbaum PS, Roth LH, Lidz C. The therapeutic misconception: informed consent in psychiatric research. *International Journal of Law and Psychiatry*. 1982;5(3-4):319-329.
- 3. Beauchamp TL, Childress JF. *Principles of biomedical ethics*. New York: Oxford University Press; 2001.
- 4. Fagerlin A, Zikmund-Fisher BJ, Ubel PA. Helping Patients Decide: Ten Steps to Better Risk Communication. *JNCI: Journal of the National Cancer Institute*. 2011;103(19):1436-1443.
- 5. Raich PC, Plomer KD, Coyne CA. Literacy, comprehension, and informed consent in clinical research. *Cancer Investigation*. 2001;19(4):437-445.
- 6. Cassileth BR, Zupkis RV, Sutton-Smith K, March V. Informed consent -- why are its goals imperfectly realized? *The New England Journal of Medicine*. Apr 17 1980;302(16):896-900.
- 7. Dresser R. First-in-Human Trial Participants: Not a Vulnerable Population, but Vulnerable Nonetheless. *The Journal of Law, Medicine & Ethics: a Journal of the American Society of Law, Medicine & Ethics.* 2009;37(1):38-50.
- 8. Franco OH, Steyerberg EW, Hu FB, Mackenbach J, Nusselder W. Associations of diabetes mellitus with total life expectancy and life expectancy with and without cardiovascular disease. *Archives of Internal Medicine*. Jun 11 2007;167(11):1145-1151.

- 9. Wandell PE. Quality of life of patients with diabetes mellitus. An overview of research in primary health care in the Nordic countries. *Scandinavian Journal of Primary Health Care*. Jun 2005;23(2):68-74.
- 10. Ye H, Daoud-El Baba M, Peng RW, Fussenegger M. A synthetic optogenetic transcription device enhances blood-glucose homeostasis in mice. *Science (New York)*. Jun 24 2011;332(6037):1565-1568.
- 11. Shaw D, Elger B. Putting patients on research ethics committees. *Journal of the Royal Society of Medicine*. 2014;107(8):304-307.
- 12. Davis TC, Williams MV, Marin E, Parker RM, Glass J. Health Literacy and Cancer Communication. *CA: A Cancer Journal for Clinicians*. 2002;52(3):134-149.
- 13. Barry CA, Stevenson FA, Britten N, Barber N, Bradley CP. Giving voice to the lifeworld. More humane, more effective medical care? A qualitative study of doctor–patient communication in general practice. *Social Science & Medicine*. 2001/08/01/2001;53(4):487-505.
- 14. Abma TA. Patient Participation in Health Research: Research With and for People With Spinal Cord Injuries. *Qualitative Health Research*. 2005;15(10):1310-1328.
- 15. Mack JW, Smith TJ. Reasons Why Physicians Do Not Have Discussions About Poor Prognosis, Why It Matters, and What Can Be Improved. *Journal of Clinical Oncology*. 2012;30(22):2715-2717.
- 16. Milne L, Moyle W, Cooke M. Hope: a construct central to living with chronic obstructive pulmonary disease. *International Journal of Older People Nursing*. 2009;4(4):299-306.
- 17. SAMW. Palliative Care. 2017; <a href="https://www.samw.ch/dam/jcr:d7ae1138-0213-481b-9023-6583bed2de12/richtlinien-samw-palliative-care.pdf">https://www.samw.ch/dam/jcr:d7ae1138-0213-481b-9023-6583bed2de12/richtlinien-samw-palliative-care.pdf</a>. Accessed 15 March 2018.

- 18. SAMW. Patientenverfügungen. 2013; <a href="https://www.rwi.uzh.ch/dam/jcr:f7c5b696-6fab-4e8a-9004-18800a87c0e7/SAMW-RL%20Patientenverf%C3%BCgung.pdf">https://www.rwi.uzh.ch/dam/jcr:f7c5b696-6fab-4e8a-9004-18800a87c0e7/SAMW-RL%20Patientenverf%C3%BCgung.pdf</a>
  Accessed 15 March 2018.
- 19. Singer PA, Robertson G, Roy DJ. Bioethics for clinicians: 6. Advance care planning. *CMAJ: Canadian Medical Association Journal*. 1996;155(12):1689-1692.
- 20. Pautex S, Notaridis G, Derame L, Zulian GB. Preferences of elderly cancer patients in their advance directives. *Critical reviews in oncology/hematology*. Apr 2010;74(1):61-65.
- 21. Mullick A, Martin J, Sallnow L. Advance care planning. *BMJ*. 2013;347(7930):28-32.
- 22. Torke AM, Sachs GA, Helft PR, et al. Scope and outcomes of surrogate decision making among hospitalized older adults. *JAMA internal medicine*. Mar 2014;174(3):370-377.
- 23. Detering KM, Hancock AD, Reade MC, Silvester W. The impact of advance care planning on end of life care in elderly patients: randomised controlled trial. *BMJ*. 2010;340.
- 24. Bundesamt für Gesundheit BAG SKdkGu-dGupc. Allgemeine Palliative Care. Empfehlungen und Instrumente für die Umsetzung. 2015; <a href="https://www.bag.admin.ch/bag/de/home/themen/strategien-politik/nationale-gesundheitsstrategien/strategie-palliative-care/grundlagen-zur-strategie
- 25. Garlo K, O'Leary JR, Van Ness PH, Fried TR. Burden in caregivers of older adults with advanced illness. *Journal of the American Geriatrics Society*. Dec 2010;58(12):2315-2322.
- 26. Wilson F, Ingleton C, Gott M, Gardiner C. Autonomy and choice in palliative care: time for a new model? *Journal of Advanced Nursing*. May 2014;70(5):1020-1029.

- 27. Schneewind JB. *The invention of autonomy: A history of modern moral philosophy*.

  Cambridge: Cambridge University Press; 1998.
- 28. Walter JK, Ross LF. Relational Autonomy: Moving Beyond the Limits of Isolated Individualism. *Pediatrics*. 2014;133(Supplement 1):S16-S23.
- 29. Mackenzie C, Stoljar N. Relational autonomy: Feminist perspectives on automony, agency, and the social self. Oxford University Press on Demand; 2000.
- 30. Wiesemann C. Vertrauen als moralische Praxis-Bedeutung für Medizin und Ethik. *Autonomie und Vertrauen*: Springer; 2016:69-99.
- 31. Lavoie M, Blondeau D, Picard-Morin J. The Autonomy Experience of Patients in Palliative Care. *Journal of Hospice & Palliative Nursing*. 2011;13(1):47-53.
- 32. World Health Organization (WHO). WHO definition of Palliative Care. 2002; http://www.who.int/cancer/palliative/definition/en/ Accessed 28 February 2018.
- 33. Reiter-Theil S. Does empirical research make bioethics more relevant? "The embedded researcher" as a methodological approach. *Medicine, Health Care and Philosophy*. April 01 2004;7(1):17-29.
- 34. Hedgecoe AM. Critical Bioethics: Beyond the Social Science Critique of Applied Ethics. *Bioethics*. 2004;18(2):120-143.
- 35. Borry P, Schotsmans P, Dierickx K. The birth of the empirical turn in bioethics. *Bioethics*. 2005;19(1):49-71.

# Appendix

### Questionnaire for Patients (Chapter 2, 3, and 4)

Leitfaden für Gicht- und Diabetespatienten zur Wahrnehmung von Risiken und Nutzen früher klinischer Studien im Bereich der synthetischen Biologie

(Bewertung des Risiko-Nutzen-Verhältnisses und Teilnahmebereitschaft)

Allgemeine Angaben	
Interviewnummer:	
Grunderkrankung: 🚨 Gicht	☐ Typ-2-Diabetes
Interviewer(in):	
Datum:	
Uhrzeit:	
Dauer:	
Begrüßung	

Vorstellung der eigenen Person

Danke für Teilnahme am Interview

Zeitrahmen: ca. 45 bis 90 Minuten, Pausen und Unterbrechungen möglich, Abbruch ohne nachteilige Konsequenzen

Ziel des Gespräches: persönliche Vorstellungen und Wünsche in Bezug auf innovative Behandlungsmethoden, vor allem aus dem Bereich der synthetischen Biologie ermitteln. Beantwortung dieser Frage sehr wichtig, da bisher noch nicht bekannt ist, was chronisch kranke Patienten über Forschungsstudien zu diesen neuen Therapien denken, Interviewpartner als Experte für die Sichtweise der Patienten; Herausstellen, dass IP sich frei äußern kann, es nicht um Wissensermittlung, sondern um pers. Meinung geht; wenn Frage nicht beantwortet werden kann, kein Problem, Zeit zum Überlegen; selbst: neutrale Befragung, keine Diskussion, eigene Meinung zurückstellen.

2 Teile: ihre jetzige Behandlung und Entwicklung neuer Behandlungsmethoden

Bitte um Aufzeichnung des Gesprächs (damit keine wichtigen Informationen verloren gehen) & Zusicherung Anonymität und Datenschutz

### Allgemeine Fragen: Gesundheitszustand und momentane Behandlung

- 1. Wie würden Sie ihren jetzigen Gesundheitszustand beschreiben?
  - 1.1 Wann wurde Gicht/Typ-2-Diabetes bei Ihnen diagnostiziert?
  - 1.2 Haben Sie noch weitere Krankheiten? Wenn ja, welche?
  - 1.3 Ich werde Ihnen jetzt ein paar konkretere Fragen zu ihrem Wohlbefinden stellen.

Standardform SF-36 vorlegen und gemeinsam ausfüllen (Dauer: ca. 5 Minuten)

#### 2. Wie werden Sie zurzeit behandelt?

2.1. Wie geht es Ihnen denn mit ihrer Behandlung?

Wenn Leid erwähnt oder beschrieben wird: Worunter leiden Sie persönlich am meisten? Wie gehen Sie hiermit um? Hat sich das im Laufe ihrer Krankheit verändert?

- 2.2. Haben Sie vorher bereits andere Behandlungen erhalten?
- 2.3. Was würden Sie sich für ihre Behandlung wünschen?

### Einstellung gegenüber Forschung/technischer Entwicklung

- 3. Um die Behandlungsmöglichkeiten für Krankheiten zu verbessern, wird in Laboren und Spitälern viel geforscht. Gehen wir noch einmal einen Schritt zurück. Forschung findet ja in fast allen Lebensbereichen statt. Was denken Sie denn über technische Entwicklungen und Forschung ganz allgemein? Sind Sie da eher neutral eingestellt oder würden Sie sich hierzu positionieren?
  - 3.1. Würden Sie sich selbst eher als technikkritisch oder als Technikbefürworter bezeichnen?
  - 3.2. Gibt es vielleicht Lebensbereiche, in denen Sie technische Entwicklungen eher gutheißen und andere, in denen Sie so etwas generell ablehnen?

### Einstellung gegenüber synthetischer Biologie

- 3. Ein sehr neues Forschungsgebiet ist die Synthetische Biologie. Haben Sie das Wort "Synthetische Biologie" schon mal gehört?
  - 4.1. *Falls ja:* Was stellen Sie sich denn darunter vor? Wie haben Sie denn von Synthetischer Biologie gehört? *Anschließend noch einmal kurz erklären.*
  - 4.2. Falls nein:

Ich habe Ihnen hier einmal eine Erklärung aufgeschrieben, die Sie sich in Ruhe durchlesen können.

Dem Patienten/der Patientin Karte 1 zur Veranschaulichung übergeben (Anhang 2).

Anschließend dem Patienten/der Patientin Grafik 1 zur Veranschaulichung übergeben und kurz erklären (Siehe Anhang 1).

- 4.3. Haben Sie Fragen hierzu?
- 4.4. Wenn Sie das so lesen und hören, was denken Sie denn hierüber? Was finden Sie hieran gut, was eher nicht?
- 4.5. Falls es nicht erwähnt wird: Was denken Sie denn über synthetische Biologie als Behandlungsmethode für Krankheiten? Was finden Sie hieran gut, was eher nicht?

### Fallbeispiel: synthetic circuit

4. Ich möchte mit Ihnen nun ein kurzes Gedankenexperiment machen.

### a. Fallbeispiel für Gichtpatienten

Ich werde Sie nun in die Welt von Herrn Müller im Jahr 2025 entführen. Herr Müller leidet wie Sie an Gicht, nimmt aber keine Medikamente ein. Er muss nur selten zum Arzt, um seinen Harnsäurespiegel kontrollieren zu lassen. Herr Müller hat sich eine Kapsel unter die Haut implantieren lassen. In dieser Kapsel befinden sich menschliche Zellen. Diese Zellen stammen von Herrn Müller selbst. Sie wurden ihm entnommen und im Labor verändert. Die veränderten Körperzellen von Herrn Müller sind in der Lage ein bestimmtes Eiweiss herzustellen, das die überschüssige Harnsäure im Blut abbaut. Zudem wurde noch ein Sensor in die Zellen eingebaut, der konstant den Harnsäurespiegel im Blut misst. Wenn dieser zu stark ansteigt, wird die der Zelle. Sie produziert dann mehr von dem harnsäureabbauenden Eiweiss. Hiervon merkt Herr Müller nichts. Alles geschieht automatisch und ganz von allein. Sie können sich das hier ansehen.

#### b. Fallbeispiel für Diabetespatienten

Ich werde Sie nun in die Welt von Herrn Müller im Jahr 2025 entführen. Herr Müller leidet wie Sie an Diabetes. Herr Müller hat sich eine Kapsel unter die Haut implantieren lassen. In dieser Kapsel befinden sich Zellen. In dieser Kapsel befinden sich menschliche Zellen. Diese Zellen stammen von Herrn Müller selbst. Sie wurden ihm entnommen und im Labor verändert. Die veränderten Körperzellen von Herrn Müller sind in der Lage eine Substanz herzustellen, welche die Insulinproduktion im Körper anregt und die Blutzuckerkonzentration senkt. Auf der Haut trägt Herr Müller ein Pflaster in das LED-Leuchten eingearbeitet sind. Nach dem Essen schaltet Herr Müller per Knopfdruck diese LED-Lämpchen an. Das Licht sorgt dann dafür, dass die Substanz in den Zellen produziert und ins Blut ausgeschüttet wird. Sie können sich das hier ansehen.

- 4.1. Wenn Sie das so hören, welche Gedanken gehen Ihnen hierzu durch den Kopf? Was finden Sie gut und was finden Sie nicht so gut?
- 4.2. Diese Behandlungsmethode wurde vor ein paar Jahren hier in Basel entwickelt. Bis diese neue Behandlungsmethode Patienten verschrieben werden kann, muss deren Wirksamkeit und Ungefährlichkeit nach fest vorgegebenen Regeln getestet und bewiesen werden. Das geschieht in Studien.

### Meinung zu klinischen Studien

5. Damit Sie eine bessere Vorstellung von solchen Studien haben, habe ich hier eine Erklärung aufgemalt

Dem Patienten/der Patientin Grafik 2 zur Veranschaulichung übergeben (Anhang 1).

Wie Sie sehen können, beginnt man zunächst damit den Wirkstoff in Zellen zu testen. Dann testet man ihn im lebenden Organismus aus. Zunächst an verschiedenen Tieren und dann schließlich im Menschen. Die Studien mit Menschen bezeichnet man auch als klinische Studien. Sie verlaufen in 3 Phasen. Ich habe Ihnen hier ein paar Karten mitgebracht, auf denen Sie ganz in Ruhe nachlesen können, bevor wir darüber sprechen. Sie können mir jederzeit Fragen stellen.

Dem Patienten/der Patientin Karten 2 bis 7 zum Lesen übergeben (Anhang 2).

- 5.1. Was ging Ihnen durch den Kopf, als Sie eben die Karten gelesen haben? Inwiefern war Ihnen dieses Prozedere vertraut oder ganz neu?
- 5.2. Gibt es etwas, dass Sie überraschend finden?
- 5.3. Falls sich ergibt, dass der Patient/die Patientin Vorwissen hat: Wo haben Sie denn schon einmal vor unserem Gespräch von klinischer Forschung gehört?
  - 5.3.1. Inwiefern haben Sie mit Ihrem Arzt über Studien zur Erprobung neuer Behandlungsmethoden gesprochen und falls ja, was hat er Ihnen hierüber erzählt? Falls nicht, unbedingt erklären: Das kann ich verstehen. Je nach Spital wird mehr oder weniger Forschung durchgeführt, d.h. Forschung gehört nicht zum regulären Behandlungsplan.
  - 5.3.2. *Nur, falls es sich aus dem Gespräch ergibt:* Haben Sie schon einmal an klinischen Erprobungen, sogenannten klinischen Studien, teilgenommen? <u>Wenn ja</u>, wie oft, wann und was wurde wie getestet? Können Sie es mir etwas genauer erzählen?
  - 5.3.3. Warum haben Sie sich entschlossen, mitzumachen?
  - 5.3.4. Wie sehen Sie diese Entscheidung heute, wenn Sie zurückblicken?

5.3.5. <u>Falls nein</u>: Wurde Ihnen schon einmal eine Teilnahme angeboten? Falls, ja: Weshalb haben Sie damals abgelehnt?

### Teilnahmebereitschaft und Risikowahrnehmung

- 6. Stellen Sie sich einmal vor, die Gicht-/Diabeteskapsel, von der ich Ihnen vorhin erzählt habe, soll erprobt werden. Inwiefern können Sie sich denn grundsätzlich vorstellen hieran teilzunehmen?
  - 6.1. Falls der Befragte eine Teilnahme vollständig ausschließt: Können Sie sich vorstellen an einer Medikamentenstudie teilzunehmen, wenn ein neues Gicht-/Diabetesmedikament erprobt wird? Falls ja, weiter mit 6.2., falls nein: An wem sollten neue Behandlungsmethoden ihrer Meinung nach erprobt werden? Wer wäre denn ihrer Meinung nach gut geeignet?
  - 6.2. Falls der Befragte sich eine Teilnahme grundsätzlich vorstellen kann: Können Sie sich eine Teilnahme in allen Phasen gleichermaßen vorstellen oder gibt es Unterschiede? Wenn ja, warum?
  - 6.3. Was denken Sie über den allerersten Schritt, d.h. Behandlungsmethoden, die zum ersten Mal am Menschen getestet werden?
    - Wie ich Ihnen schon erzählt habe, wurde diese Therapie hier in Basel entwickelt. Die Wirksamkeit und Sicherheit wurde bisher im Labor und in Versuchen mit Mäusen belegt. Bis zur ersten Erprobung mit Menschen kann es aber noch etwas dauern. Stellen Sie sich aber bei den nächsten Fragen einmal vor, die ersten Studien zur Erprobung dieses Implantats am Menschen würden tatsächlich jetzt durchgeführt. Es geht uns hierbei nicht darum, Sie für eine Versuchsteilnahme zu gewinnen. Wir möchten nur untersuchen, was Sie und andere Patienten hierüber denken.
  - 6.4. Was wären denn für Sie Gründe an einer solchen Studie als PatientIn teilzunehmen/nicht teilzunehmen? Welche Ziele würden Sie mit einer Teilnahme verfolgen? Wie sehen Sie eine Teilnahme für sich persönlich?
    - Gesucht werden Hinweise zur Wahrnehmung und Bedeutung von persönlichem Nutzen (aktuell oder mittelbar, direkt, indirekt), therapeutic misconception, Risiken
    - 6.4.1. Falls der Patient/die Patientin therapeutischen Nutzen nennt: Die Graphik erneut anschauen: Wie wir vorhin angeschaut haben, ist in den frühen Phasen eher nicht mit Heilung zu rechnen, sondern wird eher überprüft, welches die richtige Dosis ist und ob es Nebenwirkungen gibt. In der Regel profitieren Patienten von frühen Testphasen deshalb in der Regel nicht. Was würde Sie dennoch zu einer Teilnahme bewegen?
    - 6.4.2. Wenn Sie an einer solchen Versuchsreihe teilnehmen, helfen Sie, eine neue Behandlungsmethode auf den Weg zur Zulassung zu bringen. Inwiefern spielt es für Sie eine Rolle, später einmal vom zugelassenen Medikament zu profitieren?

6.4.3. Wie wir vorhin gesehen haben, ist es schwer, wenn eine Therapie noch nicht am Menschen getestet wurde, eventuelle Risiken oder Nebenwirkungen vorherzusagen. Sie haben sich sicher schon einmal den Beipackzettel ihrer Medikamente angesehen. Alles, was dort steht ist irgendwann einmal als Nebenwirkung ihres Medikaments in Studien beobachtet worden. Probiert man eine neue Therapie zum ersten Mal am Menschen aus, ist es besonders schwierig, eventuelle Nebenwirkungen vorherzusagen. Wenn Ihnen ihr Arzt erklären würde, dass man nicht genau vorhersagen kann, ob und welche Nebenwirkungen auftreten und wie stark sie sein werden, wie würden Sie hiermit umgehen?

Gesucht werden Hinweise zum Entscheidungsprozess, zum Umgang mit Unsicherheit, zum persönlichen Verständnis von Risiken und Nebenwirkungen

6.4.4. Wenn Sie einmal an Ihren Beipackzettel und Nebenwirkungen, die Sie kennen denken, wie sehen Sie das? Bis wohin wäre es denn für Sie in Ordnung mitzumachen?

Gesucht werden Hinweise zur Persönlichen Risikoschwelle/zu inakzeptablen Risiken und zum Umgang mit unbekannten Risiken

- 6.4.5. Wenn Ihnen ihr Arzt erklärt, dass die kurzfristigen Nebenwirkungen sehr gering sind und relativ genau bekannt sind, eventuelle späte Nebenwirkungen nach 2, 5 oder auch 10 Jahren aber gänzlich unbekannt sind, welche Rolle würde das bei Ihrer Entscheidung spielen?
- 6.5. Gerade im Bereich innovativer Behandlungsmethoden dauert es mitunter sehr lange bis eine Medizinanwendung wirklich im ärztlichen Alltag ankommt. Es ist deshalb möglich, unter ganz strengen Voraussetzungen auch sogenannte Phase-0-Studien mit besonders niedrigen Wirkstoffdosen durchzuführen, bei denen ein Teil der Vorversuche mit Tieren ausgelassen wird. Ziel ist es, schneller herauszufinden, wie verträglich und nützlich das Medikament bei Menschen ist. Was denken Sie hierüber?

#### Wer soll über Ihre Studienteilnahme entscheiden?

- 7. Stellen Sie sich einmal vor, ihr Hausarzt erzählt Ihnen von dieser Studie und schlägt Ihnen eine Teilnahme vor. Was würden Sie dann machen? Wie würden Sie mit diesem Vorschlag umgehen?
  - 7.1. Wie schätzen Sie sich selbst ein, wenn es darum geht über die Studienteilnahme zu entscheiden?
  - 7.2. Was denken Sie ist für Sie wichtig, wenn es darum geht über eine eventuelle Studienteilnahme zu entscheiden?

- 7.3. Inwiefern würden Sie dies mit jemandem besprechen? Mit wem? Oder würden Sie ihre Entscheidung ganz allein treffen wollen?
- 7.4. Inwieweit spielen hierbei die Aussagen ihres Arztes eine Rolle?
- 8. Bevor eine neue Behandlungsmethode wie zum Beispiel die Gicht/Diabeteskapsel in klinischen Studien an Patienten erprobt wird, muss ein
  Expertengremium entscheiden, ob die Studie überhaupt stattfinden darf. Dieses
  Gremium bewertet hierfür den möglichen Nutzen und die möglichen Risiken, die
  mit der Studie verbunden sind und wägt diese Positionen gegeneinander ab. In
  der Schweiz gibt es viele unterschiedliche Vorstellungen darüber, wer in einem
  solchen Gremium vertreten sein sollte. Was denken Sie denn dazu? Wer sollte
  in diesem Gremium vertreten sein und warum?
  - 8.1. Wie sehen Sie denn die Teilnahme von Patienten in solchen Gremien?

### **Abschlussfrage**

- 9. Wie sehen Sie denn den Zusammenhang zwischen ihrer Erkrankung und ihrer Bereitschaft an einer Studie teilzunehmen?
  - 9.1. Wenn Leidensdruck als Hauptfaktor erwähnt wird hinterfragen: Finden Sie das problematisch? Wie sehen Sie das?

### "Noch was loswerden?"

10. Gibt es etwas was Sie noch sagen möchten, auf das wir im Gespräch noch nicht eingegangen sind?

### **Angaben zur Person**

Geschlecht:	☐ weiblich	☐ männlich		
Haushaltsgröße:	1-Person	2-Personen	3-Personen	> 3-Personen
Geburtsjahr:				
Höchster Bildungsabschluss				

Vielen Dank für das angenehme Gespräch und dafür, dass Sie sich die Zeit genommen haben, heute hier mit mir über diese Fragen zu sprechen.

## **Medical Records Data Extraction Sheet (Chapter 5 and 6)**

### **Data Extraction Sheet Palliative Care Patients**

I Socio	demographic variables	
•	Month and year of birth:	Date of death:
•	Gender:	
•	Nationality:	
•	Country of origin:	
•	Religion:	
•	Marital status:	
•	Place of care after hospital (at home, re	etirement home, hospital):
•	Living place (urban, countryside):	
•	Number of children:	
	1-4; if more than four children plea	se indicate:
•	Profession:	
•	Mother tongue:	
II Disea	ises	
	Diagnosis:	
	Year and month of main diagnosis:	
	Begin of Palliative Care (year, month a	nd day):
	Treatment and duration:	
	- Disease modifying treatment:	

Symptom- and problem-related treatment:

• Complexity of palliative case: (Phase 1-5 PCOC)
Phase 1: Stable
Phase 2: Unstable
Phase 3: Deteriorating
Phase 4: Terminal
Phase 5: Bereaved
III Advance Directives
Is /was there
-discussion about Advance Care Planning:
no
yes; if yes, what kind of discussions?
-written statement of goals of care:
no
yes; if yes, what kind of goals?
-ls/was there an AD (Advanced Directives)
no; are/were there any indications why not?
yes; if yes, oral or written AD?:
yes; if yes, long or short form?:
yes; if yes, did the patient receive advice (doctor, nurse, lawyer)
yes; who signed the AD?

Motivation and personal values for AD:
No special motivation to fill in the AD:
-If yes, specify:
This AD is applicable to the following situation:
-All situations:
-Specific situations (specify):
<ul> <li>Was there an adaption of the AD during the hospitalization?</li> </ul>
no yes
->If yes, please specify what kind of adaptions:
Preference for or against specific treatments:
-I do not wish to give any instructions concerning specific medical measures, but I would ask the health care team to act as far as possible in accordance with my wishes:
- I wish to give specific instructions for the following situations:
Artificial hydration and nutrition
☐ yes ☐ no
Resuscitation in the event of a cardiac and/or respiratory arrest
☐ yes ☐ no
Comments:

Trusted person:  no
☐ yes: ->specify: Proxy: GP:
Organ donation:  yes  no
<ul><li>-&gt;If yes, specify which organ(s):</li></ul>
Did/does the patient call for pastoral/spiritual care?     yes  no
->If yes, specify how often did/does the patient receive pastoral/spiritual care?
<ul> <li>When did discussions take place regarding PC treatment, patient's wishes, values preferences and Ads with family members, the patient's GP, Spitex nurses and othe relevant care givers?</li> <li>at admission during hospitalization after hospitalization</li> </ul>
after patient's death never
<ul> <li>Were there conflicts within the health care team or among family members and HCPs about patient's values and preferences?</li> </ul>
☐ yes ☐ no
>If yes, specify:
General wishes of patients during hospitalization:
Other information:
Contact person: